

RETURN TO 396-SS Department of Energy Washington, DC 20545

February 26, 1988



Hr. Charles E. MacDonald, Chief Transportation Branch Division of Safeguards and Transportation. NMSS U.S. Nuclear Regulatory Commission Washington, D.C. 20555

Dear Mr. MacDonald:

The Department of Energy (DOE) requests renewal of the NRC Certificate of Compliance (CoC) Number 9152 having an expiration date of March 31, 1983. DOE programs have a continuing need for the packaging. The operating history and a brief status of the package are given below:

ņ	Package Identification Number	USA/9152/D( )
ç	Model Humber	CNS 1-BC II
0	Number of Shipments in Last Five Years	23 Shipments
0	Operating Incidents	None

 Maintenance of packaging has been carried out in accordance with Section 8.2 of the SARP (EGG-TM1-8008)

There have been no changes in the packaging, design, hardware or contents during the current period of certification. The Safety Analysis Report for Packaging (SARP) (EGG-TMI-8008) for the CNS 1-13C II has been consolidated to include all amendments made to the SARP when previously held by Chem Nuclear Systems, Inc. The CNS-1-13 II packaging was purchased by DOE in April 1982.

Should you require any further information on this package or enclosures, please call up (353-5394), or Erich Opperman (353-3954) of my staff.

Sincerely. Marke 1. Mauck Charles J. Mauck Chief of Packaging Certification Office of Security Evaluations Defense Programs 6 Enclosures 24085 8803030081 880226 PDR ADUCK 07102152



1

.

~ -

8803030090 880226 PDR ADDCK 07109152 C PDR

EG&G IDAHO, INC. 1955 FREMONT AVE. IDAHO FALLS, IDAHO 83415

Submitted By:

DECEMBER 1987

10 CFR 71 TYPE "B" FACKAGING REQUIREMENTS

SAFETY ANALYSIS REPORT

٠

FOR

MODEL 1-13C II PACKAGING

TO



RETURN TO 395-55

BUDWATORY CLERATOR. FILE COLD

EGG-TMI-8008



SAFETY ANALYSIS REPORT

FOR

MODEL 1-13C II PACKAGING

то

10 CFR 71 TYPE "B" PACKAGING REQUIREMENTS



DECEMBER 1987

Submitted By:

EG&G IDAHO, INC. 1955 FREMONT AVE. • IDAHO FALLS, IDAHO 83415



19085



## TABLE OF CONTENTS

																				Page
1.	GENERA	L INFORM	IATION .		• •	• •	••	•	•	•	•	•		•	•	•	•		•	1-1
	1.1 1.2	Introduc Package	tion Descript	 ion	•••	•••	 	•	•	•	•	•	•••	•	•	•	•	•	•	1 - 1 1 - 3
		1.2.1 1.2.2 1.2.3	Packagin Operatio Contents	g nal F of P	 eatu acka	 res ging	•••	•	• •	•	• •	•	•••	•	• •		• •	• •	•	1-3 1-8 1-8
	1.3	Appendix	<		•••			•	•	•	•	•		•	•	•	•	•	•	1-10
2.	STRUC	TURAL EVA	LUATION		•••	•••		•	•	•		•	•••	•	•	•	•	•	•	2-1
	2.1	Structur	ral Desig	n.	••	•••		•	•	•	•	•		•	•		•	•	•	2-1
		2.1.1 2.1.2	Discussi Design C	on . riter	 ia	•••	•••	•	•	•	•	•	•••	•	•	•	•	•	•	2-1 2-4
	2.2 2.3 2.4	Weights Mechanic General	and Cent cal Prope Standard	ers o rties s for	f Gr of all	avit Mate Pac	y ria kag	ls es		• •	• • •	• •	• • • •	• •	• • •	• •	• •	• •	•	2-6 2-7 2-11
	ć	2.4.1 2.4.2 2.4.3 2.4.4	Minimum Tamperpr Positive Chemical	Packa oof F Clos and	ge S eatu ure Galv	ize re anic	Re	act	tio		•	• • •	• • • •	• • •	• • •		• • •	•	• • •	2-11 2-11 2-11 2-11
	2.5	Lifting	and Tied	own S	tand	ards	fo	r /	411	P	ac	ka	ges	•	•	•	•	•	•	2-11
		2.5.1 2.5.2	Lifting Tiedown	Devic Devic	es es	•••	 	•	•	•	•	•		•	•	•	•	•	•	2-11 2-51
	2.6	Normal (	Condition	s of	Tran	spor	t.	•	•	•	•	•	•••	•	•	•	•	•	•	2-66
		2.6.1 2.6.2 2.6.3 2.6.4 2.6.5 2.6.6 2.6.7 2.6.8 2.6.9	Heat Cold Reduced Increase Vibratio Water Sp Free Dro Corner D Compress	Exter d Ext n . ray p . rop ion	 nal erna  	 Pres 1 Pr  	sur ess	e ure	• • • • •	• • • •	• • • • • • • •	• • • •	· · · · · · · · · · · · · · · · · · ·	• • • • •	• • • • •	• • • • •	• • • • •	• • • •	• • • • •	2-66 2-77 2-79 2-81 2-81 2-81 2-81 2-91 2-91
		2.6.10 2.6.11	Penetrat Conclusi	ion ons	•••	•••	•••	•	•	•	•	•	•••	•	•	•	•	•	•	2-91 2-91



•

## TABLE OF CONTENTS (CONT'D)

····· ·

2.7	Hypothet	tical Accident Conditions	2-92
	2.7.1 2.7.2 2.7.3 2.7.4 2.7.5 2.7.6	Free Drop	2-92 2-110 2-112 2-116 2-116 2-116
2.8 2.9 2.10	Special Fuel Roo Structu	Form	2-117 2-117 2-117
	2.10.1 2.10.2 2.10.3	Analytic Methods	2-118 2-152 2-154
2.11	Appendi Package	x - Engineering Drop Test for 1-13C II Shipping	2-169
	2.11.1 2.11.2 2.11.3 2.11.4 2.11.5	Scope	2-169 2-169 2-171 2-189 2-204
2.12	Attlach	ment	2-213
THERM	AL EIVAL	UATION	3-1
3.1 3.2 3.3 3.4	Discuss Summary Technics Thermal	ion	3-1 3-5 3-8 3-8
·	3.4.1 3.4.2 3.4.3 3.4.4 3.4.5 3.4.6	Thermal Model	3-8 3-20 3-31 3-31 3-36 3-36
3.5	Hypothe	tical Accident Thermal Evaluation	3-37
	3.5.1 3.5.2 3.5.3 3.5.4 3.5.5 3.5.6	Thermal Model	3-37 3-37 3-38 3-56 3-56 3-66





3.

•

## TABLE OF CONTENTS (CONT'D)

	3.6	Appendix
4.	CONTA	INMENT
	4.1	Containment Boundary
		4.1.1       Containment Vessel       4-1         4.1.2       Containment Penetration       4-1         4.1.3       Welds and Seals       4-2         4.1.4       Closure       4-2
•	4.2	Requirements for Normal Conditions of Transport 4-3
		4.2.1Containment of Radioactive Material4-34.2.2Pressurization of Containment Vessel4-34.2.3Containment Criterion4-3
	4.3	Containment Requirements for Hypothetical Accident Conditions
		4.3.1Fission Gas Products4-54.3.2Containment of Radioactive Material4-54.3.3Containment Criterion4-8
	4.4 4.5	Appendix A - Normal and Accident Radioactive Material Limits for the 1-13C II Cask
5.	SHIEL	DING EVALUATION
	5.1 5.2 5.3	Discussion and Results
		5.3.1       Damage Predictions - Corner       5.3         5.3.2       Damage Prediction - Side       5.6         5.3.3       Damage Prediction - Ends       5.10
	5.4 5.5	Shielding Evaluation
6.	CRITIC	CALITY EVALUATION
	6.1 6.2 6.3 6.4 6.5 6.6	Discussion and Results



•





## TABLE OF CONTENTS (CONT'D)

7.	OPERAT	ING PROCEDURES
	7.1 7.2 7.3 7.4	Procedures for Loading Package
8.	ACCEP	ANCE TESTS AND MAINTENANCE PROGRAM
	8.1	Acceptance Tests
		8.1.1Visual Inspection8-18.1.2Structural and Pressure Tests8-18.1.3Leak Tests8-18.1.4Component Tests8-28.1.5Tests for Shielding Integrity8-28.1.6Thermal Acceptance Tests8-2
	8.2	Maintenance Program
		8.2.1Structural and Pressure Tests8-28.2.2Leak Tests8-28.2.3Subsystem Maintenance8-28.2.4Valves and Gaskets on containment Vessel8-38.2.5Shielding8-38.2.6Thermal8-38.2.7Appendix - Leak Test Procedures8-3
	8.3	Appendix

ι.







#### 1.0 GENERAL INFORMATION

#### 1.1 Introduction

This Safety Analysis Report describes a reusable insulated and shock absorbing shipping package designed to protect radioactive material from both normal conditions of transport and hypothetical accident conditions. The package is designated as the Model 1-13C II Package. It is intended to deliver Fissile Class III material, irradiated reactor hardware, and other solid materials including process solids, in excess of Type A quantities.

The package is an improved lineal descendant of the CNS 1-13C, Package Identification Number USA/9081/B ( )F, and the General Electric Model 1600 Shipping Cask, Package Identification Number USA/9044/B ( )F.

The improvements contained in the Model 10-3C II Package include the following:

- A pair of circular shock or impact absorbing limiters, placed peripherally around both top and bottom of the circular cask body. They provide capability to protect the package from damage under hypothetical accident conditions.
- A fire shield surrounding the cylindrical cask body. The fire shield is designed to protect the package from damage during the hypothetical fire events.
- Numerous improvements to closure and sealing details providing greater containment integrity and improved decontamination features.

1-1







- Shipment of fissile material within the cask is accomplished through the use of various container configurations. The fissile material is radioactive fuel bearing debris samples with a maximum enrichment of 3% in U-235. Dependent upon the container utilized, size, geometry or mass of the sample will be controlled to preclude criticality.
  - A. General Debris Container Configuration

Shipment of up to 15.4 kg UO<sub>2</sub> (400 grams of U-235) can be placed in closed debris containers with a defined geometry. The debris containers are placed in DOT approved 2R containers. A shoring cage within the cavity of the cask segregates each of the three 2R containers and ensures a close fit within the cask cavity. The 2R containers provide the supplementary containment of the debris samples. The debris containers define the size and geometry configuration of the debris within the cask cavity which, in turn, limits the mass of fissile material.





Revision O

#### 1.2 Package Description

#### 1.2.1 Packaging

A.

Cask and Impact Limiters (See CNS Drawing No. E-1-436-111)

The packaging system consists of a pair of circular impact limiters (or overpacks) placed peripherally around each end of a cylindrical cask. The cylindrical cask is 39.12 inches in diameter and 68.06 inches high. The cylindrical cask body is comprised of a 1/2 inch stainless steel shell, Type 304, overlaid with 1/4" stainless steel fire shield. Between these two materials is a .065 (16 gauge) stainless steel wire wrap, thus providing an air gap for additional thermal protection. The inner cylindrical shell of the cask, or cavity, is fabricated of 1/2 inch stainless steel, Type 304. The annular space between these inner and outer shells is filled with lead, having a thickness of approximately 5 inches. The base of the cask consists of inner and outer circular plates, fabricated of 1/2 inch stainless steel, Type 304, separated with a lead shield of approximately 6 inches thickness.

The cask lid is comprised of a conical sector attached top and bottom to inner and outer circular flat plates. The conical segment of lid is fitted within a step recess approximately 6 inches depth in the cask body. Conical surfaces and inner surfaces of the lid are fabricated of 1/2 inch stainless steel, Type 304. The outer circular flat plate segment of lid is fabricated from a 1 inch plate, machined to approximately 0.8 inches thickness. Α load bearing step of approximately 0.2 inches thickness, surrounds the periphery of this outer lid plate, reacting containment bolt preload forces and confining the flat gasket. A solid silicone O-ring exists near the edge of bottom of the lid. The space between inner and outer flat lid plates is filled with lead, having a thickness of approximately 5.78 inches.

The removable lifting lugs are attached to lifting pads integral with the cylindrical cask body. Tiedown lugs are integral to the top energy absorbing overpack assembly.



1-3



#### Revision O



Each limiter has an external shell, fabricated from ductile low carbon steel, which allows it to undergo large deformations without fracturing. Each is secured to the cask body by six (6) one inch ratchet binders, connecting top and bottom overpacks. The volume between the inner and outer shell of the overpack is filled with a shock and thermal insulating material consisting of rigid polyurethane foam, having a density of approximately 16.5 lbs/ft<sup>3</sup>. The insulating material is poured into the cavity between the two shells and allowed to expand, completely filling the void. It bonds to the shells, creating a unitized construction of the packaging. Properties of these materials are further described in Section 2.3.

#### <u>Containment Vessel</u>

The containment boundary of the package is defined as the inner stainless steel shell of the cask body together with closure features comprised of: the lower surface of the cask lid, 12 equally spaced 1-1/4 inch closure bolts, a solid silicone flat gasket, and a silicone O-ring.

Refer to Section 4.0 for a complete description of the containment boundary for the 1-13C II Package.

#### <u>Neutron Absorbers</u>

There are no materials used as neutron absorbers or moderators in the Model 1-13C II Package.

#### Package Height

Gross weight for the package is approximately 27,000 pounds. This includes a payload weight of 3,000 pounds.

#### <u>Receptacles</u>

There are no internal or external structures supporting or protecting receptacles.

### <u>Crain/Vent Ports</u>

The cask is provided with two 1/2" O.D. lines. These are used for the draining, venting and removal of entrapped liquids. Refer to Section 4.1.2 for drain/vent system descriptions.

#### <u>Tiedowns</u>

Tiedowns are a structural part of the package. From the General Arrangement Drawings shown in Appendix 1.3, it can be seen that four tiedown lugs are provided as an integral part of the top energy absorbing overpack assembly. Refer to Section 2.5 for a detailed analysis of their structural integrity.

#### Lifting Devices

Lifting devices are a structural part of the package. From the General Arrangement Drawing shown in Appendix 1.3, it can be seen that two removable lifting lugs attached to the cylindrical cask body are provided. A single lifting lug is also provided for removal and handling of the lid. Refer to Section 2.5 for a detailed analysis of their structural integrity.

#### Pressure Relief System

The drain port is provided with a plug, permitting venting of internal pressure within the containment cavity generated by decay heat, prior to lid removal. The top vent port is provided with the same plug for venting any pressure between the O-ring and flat lid gasket prior to lid removal.

Refer to Section 4.1.2 for a description of this pressure relief feature.





Revision O

### Feat Dissipation

There are no special devices used for the transfer or dissipation of heat. The package maximum design capacity is 800 watts. Refer to Section 3.0 for the thermal evaluation of the 1-13C II Package.

### • <u>Coolants</u>

There are no coolants involved.

<u>Protrusions</u>

There are no outer or inner protrusions. Tiedown lugs are removed prior to transport.

### <u>Shielding</u>

Cask walls provide a shield thickness of 5 inches lead and 1-1/4 inch steel. Cask ends provide a minimum of 6 inches lead and 1 inch steel.

The contents will be limited such that the shielding provided will assure compliance with dose rate limits as specified in 10CFR71.

### 3. Fissile Material Containers

### 1. <u>General Debris Container</u>

The general debris container configuration as described in Part A of Section 1.1 consists of a general debris container (loaded with either a fuel pin sample cell or debris container) loaded into a 2R container in a shoring cage. A shipping configuration is shown in EG & G drawing TMI 1113 in Section 1.3.

The following is a description of these components.

#### <u>General Debris Container</u>

This component is a handling insert which consists of a 300 series stainless steel vessel, with welded concentric shells and encapsuiated lead shielding. The external dimensions are nominally 8.0 in diameter by 12.75 in height. The cavity size is 2.75 in diameter by 7.79 in height. The nominal cavity volume is 2300-2600 cubic centimeters. The cask exterior walls are smooth. A steel lid closes the upper portion of the cavity. The lid is held in place by a jam-nut arrangement. A 1/4 in drain line located in the lower portion of the container permits dewatering of the contents. A fine steel mesh screen covers the drain tube orifice.

The container permits debris loading into either of two small baskets; (1) The fuel pin sample cell, or (2) the debris basket (see Drawing 2E-3200-1040 in Section 1.3).

Eye bolts on the shielded container body are used for in-plant handling and are removed prior to shipping. The lead shielding in the container is solely required for in-plant handling, (it is not required to augment the cask shielding).

#### <u>2R Container</u>

The 2R container meets the DOT requirements of 49CFR 178.34. It is fabricated from steel pipe and has external dimensions of 10 3/4 inches O.D by 14 1/4 inches high. The bottom end is welded to the container walls. The container cover is one inch thick and is threaded to the container wall to provide closure. Per DOT requirements, a non-hardening thread lubricant is used and the lid closure is adequately torqued. The debris container is placed within the 2R container or it may remain empty. Radial movement of the debris container within the container is accommodated by pine strip shoring (1"x2" x 10").



#### Cask Cavity Shoring Cage

The wooden (pine) shoring is constructed in a cylindrical shape, and includes inner and outer  $2 \times 1/8$  in. thick rings of carbon steel as strapping.

#### 1.2.2 Operational Features

Refer to the General Arrangement Drawing of the packaging in Appendix 1.3. There are no complex operational requirements associated with the Model 1-13C II package, and none have any transport significance.

#### 1.2.3 Contents of Packaging

- (1) Type and form of material
  - (1) Greater than Type A quantity of nonfissile ridioactive
     material as solidified or dewatered process solids (resins)
     within a sealed secondary container: or
  - (11) Greater than Type A quantity of irradiated solid reactor components within a sealed secondary container.
  - (111) Greater than Type A quantity of irradiated fuel debris (dewatered) within secondary containers.
- (2) Maximum quantity of material per package

For the contents described in (1)(i), (ii), and (iii):

Not to exceed a decay heat generation of 800 watts and 3,000 pounds including weight of the contents and secondary container; and

For the contents described in (1)(1):







For the contents described in (1)(111):

The maximum U-235 enrichment of the uranium oxide fuel material must not exceed 3 weight %. The average burnup of the fuel material must not exceed 3,165 MWD/MTU and must be cooled for a least six years.

General Debris Container Configuration
 Fissile contents not to exceed 15.4 kg U0<sub>2</sub> (400 grams U-235 prior to irradiation).





### 1.3 APPENDIX



- 1.3.1 Drawings
  - 1.3.1.1 CNS Drawing No. E-1436 111, Rev. D., "CNS 1-13C II Shipping Cask"
  - 1.3.1.2 GPU Nuclear Drawing No. 2E-3200-1040, Rev. 0, "General Debris Shielded Container - 1 Assy & Fuel Pin Sample Shielded Container - 2 Assy"

### 1.3.2 Sketches

1.3.2.1 Model 1-13C II Package







.





· · · · · · · · · ·

• •

#### 2.0 STRUCTURAL EVALUATION

This Chapter identifies and describes the principal structural engineering cesign of the packaging, components, and systems, important to safety and to compliance with performance requirements of 10 CFR 71.

#### 2.1 <u>Structural Design</u>

#### 2.1.1 Discussion

The purpose of the 1-13C II Package is to provide a safe means of transport for radioactive materials in excess of Type A quantities. The package has been designed to provide a shielded containment vessel that can withstand the Normal Conditions of Transport as wellas those associated with the Hypothetical Accident Condition.

Principal structural elements of the system consist of:

- A. Containment Vessel
- Biological Shield
- C. Fire Shield
- D. Impact Limiters

Components A-D design and function in meeting the requirements of 10 CFR 71 are discussed below.

The 1-i3C II assembly is designed to protect the payload from normal transport and hypothetical accident conditions. These include: transport environment, 30 foct drop test, 40 inch puncture test, 1475<sup>0</sup>F thermal exposure and transfer or dissipation of any internally generated heat.







#### 2.1.1.1 <u>Containment Vessel</u>

The cask is comprised of inner and outer stainless steel shells Type 304, one-half inches in thickness, enveloping the lead biological shield. The inner shell serves as the package containment boundary. A removable tapered lid is attached to the cask body with twelve (12) equally spaced 1-1/4 - 7 UNC bolts. All welds are subjected to non destructive examination.

The lid to cask body interface employs a high temperature, solid silicone O-ring and a flat, high temperature, silicone seal. The flat silicone seal is captured in a machined circumferential lid step. This step is designed to preclude overcompression of the flat gasket and O-ring materials under bolt preloads and inpact forces. all transport environments as well as accident conditions, i.e., 30 foot drop, 40 inch puncture test requirements etc., are met by the external shell with impact limiters installed as discussed in Section 2.1.1.4 below. All thermal loading and dissipation requirements are met as discussed in Sections 2.1.1.3 and 2.1.1.4 below.

#### 2.1.1.2 Biological Shield

The area between the two shells discussed in Section 2.1.1.1 is filled with lead. The lead fill is subjected to Gamma Scan inspection to assure lead integrity. The designed thickness assures that no biological hazard is presented by the 1-13C II and all shielding requirements of 10 CFR 71 are met. Lead bonding is not assumed to take place between the inner and outer shells.



2-2

#### 2.1.1.3 Eise Shield

The maximum thermal loading of 800 watts is dissipated via condition through the cask structure and normal convection to the atmosphere.

The fire shield consisting of .065 diameter stainless wire spiral wound on the outside surface of the cask and covered with 1/4" stainless steel cladding provides thermal protection during the  $1475^{\circ}F$  thermal exposure test. The impact limiters provide thermal protectiom for the ends of the package. During hypothetical fire accidents, seal temperature is maintained below  $360^{\circ}F$  and lead temperature is maintained below  $509^{\circ}F$ . These temperatures are well below the design limit for the seal material and well below lead melt temperature.

#### 2.1.1.' Impact Limiters

The impact limiters (overpacks) are designed to protect the package from deformation during the 30 foot drop and provide thermal protection as discussed in Section 2.1.1.3. Their construction consists of full welded steel shells filled with foamed in place rigid structural polyurethane foam of a 16.5  $1b/ft^3$  density. The foam provides controlled deformation rates during impact. The foam degrades at 400°F and forms an air space within the impact limiter shell which provides an effective thermal barrier during the 1475°F thermal exposure test.

#### 2.1.1.5 Summery

Detailed discussions of all components and material utilized in the CNS 1-13C II Package including stress, thermal, and pressure calculations are contained in the applicable sections of the SAR.



#### 2.1.2 Jesiga Criteria

The 1-13C II has been designed to be a simple strong package that will provide takimum flexibility for multiple usage as well as minimize potential exposure to operating personnel. Its size and shielding capacity will allow a variety of existing and future payloads to be safely transported. The containment vessel is fully sealed and is <u>fully isolated</u> from the cask exterior by lead and a heavy steel jacket. Since the package ends represent the most critical portion of the cask, they are further protected by impact limiters. Impact limiters of this type have been successfully used on many Type "B" packages. They not only reduce impact loads but also provide efficient thermal

The dominant loads imposed upon any Type "B" package are due to the hypothetical accident drop events. Successive full scale drop tests have demonstrated the capabilities of the 1-13C II Package to survive these dominant load without incurring stresses or strains measurably in excess of yield values.

Where appropriate, the evaluations of package structural integrity are based upon demonstrated test results. Where not appropriate, analytic methods are used to augment test data. In all cases, the package has been designed to provide well defined load paths which lend themselves to simple, highly reliable structural analysis methods. No new state-of-the-art approaches have been used for energy absorption, thermal insulation or analytic evaluation. All analytic techniques used throughout the SAR are proven methods that have been used in past submittals. Details of these methods have been used in past submittals. Details of these methods appear in Section 2.10.



#### 2.1.2.2 <u>Stress Categories</u>

Regulatory Guide 7.6 "Design Criteria for the Structural Analysis of Shipping Cask Containment Vessels" was used in conjunction with Regulatory Guide 7.8 "Load Combinations for the Structural Analysis of Shipping Casks" to evaluate the 1-13C II Package. Material properties used in the analysis can be found on Table 2.3-1.

#### 2.1.2.2.1 <u>Containment Vessel</u>

The containment vessel is interpreted to be the inside stainless steel shell and its closures. Regulatory Guide 7.6 was used for the evaluation of the containment vessel for both the Normal Conditions of Transport and the Hypothetical Accident Conditions. Material properties used in the evaluation correspond to the design stress values  $S_m$ ,  $S_y & S_u$  given in the ASME Code, Section III. Class I, 1977 Edition as amended.

#### 2.1.2.2.2 Cask and Overpack

Structural evaluation of non-containment vessel items such as the external skins, closures, overpacks, lifting and tiedown fitting were evaluated against yield and ultimate material properties as presented in the ASME Code, Section III, Class I. For Normal Conditions of Transport, yield strength was used as a maximum stress for the cask. The overpack is allowed to exceed yield stress for nomal conditions; hence, ultimate evaluating Accident Conditions, ultimate stress or ultimate strains were used as the acceptance criteria.

2.1.2.3 Other Structural Failure Modes



2-5
Revision O

#### 2.1.2.3.1 Eristle Fracture

The primary material used in the cask is 304 stainless steel. This material does not experience a ductile to brittle transition in the temperature ranges of interest (down to  $-40^{\circ}$ F); hence is safe from brittle fracture.

#### 2.1.2.3.2 <u>Buckline</u>

Buckling per Regulatory Guide 7.6 is an unacceptable failure mode for the containment vessel. The intent of this provision is to preclude large deformations which would compromise the validity of linear analysis assumptions and quasi-linear stress allowable as given in Paragraph C.6. of NRC Regulatory Guide 7.6.

Successive drop tests at the most critical orientations of the package failed to induce any measurable distortion of the containment vessel, or exterior shell. Under service conditions, internal pressures would induce membrane biaxial tensile stress components in the containment vessel. These tensile stresses would tend to reduce compressive stresses due to hypothetical accident impact induced internal forces. Thus, under these conditions, the package would be less susceptable to buckling failures than under the conditions tested. Since no incipient buckling was experienced during the test, it may be safely concluded that buckling is not a probable failure mechanism for the 1-13C II Package. No further consideration of buckling is therefore provided in this report.

2.2 <u>Weights and Center of Gravity</u>

The center of gravity of the 1-13C II package is located at the geometric center of gravity and is shown in the sketch found as Figure 2.5.2-1 on page 2-54.









Weight breakdown is as follows:

Cask Body	17,955
Lid	2,115
Top Overpack	1,550
Bottom Overpack	1,255
Overpack Binders	480
Contingency (2.8%)	
Net Package	24,000 lbs.
Payload	3.000
TOTAL GROSS PACKAGE WEIGHT:	27.000 lbs.

#### 2.3 <u>Mechanical Properties of Materials</u>

The Model 1-13C II package is fabricated from stainless steel, low carbon steel, and structural foam. Drawing  $\Sigma$ -I-436-111; Section 1.3, defines the specific material used for each item of the package. Table 2.3.1 presents material properties used through the analysis and references the sources of these data.

The energy absorbing overpacks are constructed of rigid polyurethane foam with a density of 16.5  $1b/ft^3$ , foamed in place, self-extinguishing. Figure 2.3-1 represents the atress-strain curve for the foam used for this package. The curve provides both minimum and maximum compressive properties and was derived from eight samples of varying density and grain direction. A 95% probability factor was applied to the standard deviation to establish the spread shown. CNSI Foam Specification No. 1-436-112 defines the detail foaming testing procedure. It specifies that foam samples will be taken during the actual foaming process and tested to verify that they are within  $\pm$  10% of the mean curve at 10%, 30% and 60% strains.



2-7







7

# m

TABLE 2.3-1 Mechanical Properties of Materials

	Grade		Strength (ksi)			Elastic	Thermal Expansion	
Haterial	or Type	Tempegature F	Yield (1)	Ultimate (2)	Allowable(Sm) (3)	Modulua [4] (x10 <sup>°</sup> psi)	Coefficient (5) (x10-6 in/in.	
ASTH-A240 Plate . (Shells)	304	70 • 100 200 300 400 • 500	30.0 25.0 22.5 20.7	75.0 71.0 66.0 64.4 63.5	- 20.0 20.0 20.0 10.7	28.3 - 27.7 27.1 26.6 26.1	9.11 9.16 9.34 9.47 9.59 9.70	
λSTM-λ479 Bar, Round (Bosses & Lid Lug)	304	70 100 200 300 400	30.0 25.0 22.5 20.7	75.0 71.0 66.0 64.4	- 20.0 20.0 20.0 18.7	28.3 - 27.7 27.1 26.6	9.11 9.21 9.50 9.73 9.96	
ASTM-A514 Plate (Tiedown Frame)	B	70	100 <sup>(8)</sup>	110 <sup>(8)</sup>	-	29 <b>.</b> 9	6.07	
ASTM-A516 Plate (Overpack Inner Shell)	70	70 100 200 300 400	38.0 34.6 33.7 32.6	70.0 70.0 70'.0 70.0	- 23.3 23.1 22.5 21.7	27.9 - 27.7 27.4 27.0	6.07 6.20 6.67 7.10 7.54	
ASTM-A414 Sheet (Overpack Ext. Shell) or ASTM-A 285 Plate	c c	70 100 200 300 400	30.0 (8)	55.0 (8)	- 13.7 (min)	29.9	6.07	
ASTH-A354 Bolts (Closure Bolts) [SAE J429]	DD [A]	70	125 (9)	150 (9)	- ·	29.9	6.07	

1.2







Tabl -1 (Continued) Nech. .al Properties of Materials

Haterial	Grade or Type	Tempgrature F	Yield (l)	Strength Ultimate (2)	(kai) Allowable(Sm) (J)	Sitev[3 Formpow (4) (4) (1)	Thermal Expansion Coefficient (5) (x10-6 in/in.
ASTH-A449 Bolts (Rachet Binder Bolts) [SAE J429]	λ <b>ι</b> Β [5]	70	92 (8)	120 (8)	-	29.3	6.07
QQ-L-171e or ASTH B-29-55 (Lead Shielding)	λ or C Chem- Ical	: 70	5.0 Dynamic Compres- sion Hin 	10.0	2.3 (yield) (6)	2.0	16.4 (50 <sup>0</sup> -212 <sup>0</sup> r)
ASTM-A193 Bolts	Class 1, Grade B8 c Class 1A,Grade	r 70 or 100 B8A	30 @ 100° (8)	75 8 100 (8)		28.3 8 70	6.07 8 70°

2-9

- Reference:

- (1) ASHE Code, Section III, Appendices, Table 1-2.1 for ferritic material; Table 1-2.2 for austentic materials.
- (2) ASHE Code, Section III, Appendices, Table I-3.1 for ferritic material; Table 1-3.2 for austentic materials.
- (3) ASHE Code, Section III, Appendices, Table I-1.1 for ferritic material; Table 1-1.2 for austentic materials.
- (4) ASHE Code, Section III, Appendices, Table 1-6.0, Page 69.
- (5) ASHE Code, Section III, Appendices, Table I-5.0, Page 68, coefficient B.
- (6) Cask Designers Guide, ORNL-NSIC-68.
- (7) Sandia lab report No. SAND 77-1872 (NUREG/CR-0481)
- (B) ASTH Axxx"Standard Specification for . . . " (xxx = Alloy designation).
- (9) ASHE Code, Section III, Table I-1.3, Page 98.

Revision O



2-10

.

Revision O

# 2.4 General Standards for All Packages

# 2.4.1 MINITUM Package Size

The cylindrical cask is 39.12 inches in diameter and 68.06 inches high.

#### 2.4.2 Tamperproof Feature

Seal wires are attached to the designated ratchet binder top and bottom pins.

## 2.4.3 Fositive Closure

The positive closure system has been previously described in Section 1.2.1.

#### 2.4.4 Chemical and Galvanic Reputions

The materials from which the packaging is fabricated (steel, lead, and polyurethane foam) along with the contents of the package will not cause significant chemical, galvanic, or other reaction in air. nitrogen, or water atmospheres.

#### 2.5 Lifting and Tiedown Standards for All Packages

#### 2.5.1 Lifting Devices

The I-I3C II Cask is provided with two lifting lugs attached to the side of the cask by which the cask and load can be lifted. The lid is provided with a lifting ring by which the lid may be removed from the cask. Neither the cask lifting lug nor the lid lift ring will be used for tiedown and each will be provided with a cap or locking device to prevent such use.

The load requirements for lifting devices are defined in 10 CFR 71, Subpart E. Para 71.45 as being capable of supporting three times the weight without generating stresses in excess of its (the containers) yield strength.







Revision O

#### 2.5.1.1 Lifting Lugs & Bolts

The load condition imposed on the lift lugs is as follows:



۸.

The lugs can be used only with overpacks removed. Therefore, the total lifted weight is: W = 27000 - 3280 = 23720 lbs.

The lug load is: P = (23720 lb)(3g)/2 lugs = 35,580 lb/lug

Shear in Bolts - each lifting lug is attached to the container with 4 1-8 UNC-2A x 2-3/4, ASTM-A449 Hex Head cap screws.

For each bolt: Tensile Stress Area =  $0.606 \text{ in}^2$ 

Nominal Shear Stress =  $\frac{(35,580)1b}{4(0.606)in^2}$  = 14678 psi

The maximum Shear Stress is found at the center of the bolt cross section and is  $4/3 \times$  the nominal stress. Therefore

σ<sub>5</sub> = 4/3(14678)= 19570 psi.

The effects of this stress are evaluated when combined with the Tensile Bolt Stress computed in the next section.







plate. This loading will consist of tension in the lower bolts and a "bearing pressure" between the plate and the lifting ear.



(1) Summation of Forces

1/2 qbl = F

(2) Summation of Moments

M = F(L - 1/3)

L = 5 inch

b = 5.5 inch

 $A_{bolt} = 0.606 in^2$ 

 $E_{Bolt} = 29.9 \times 10^6$  psi

 $E_{\text{stainless steel}} = 28.3 \times 10^6 \text{ psi}$ 

(3) Condition of Compatibility - Deflection Analysis  $\frac{\delta \text{ bolt}}{\delta \text{ plate}} = \frac{\frac{\sigma \text{ bolt}}{E \text{ bolt}}}{\frac{\sigma}{E} \text{ s.s.}}$   $\frac{\delta \text{ bolt}}{\delta \text{ plate}} = \frac{\sigma \text{ bolt}}{1.06 \text{ q}} = \frac{F}{\frac{2A \text{ bolt}}{1.06 \text{ q}}}$ Also:  $\frac{\delta \text{ bolt}}{\delta \text{ plate}} = \frac{L - 2}{1}$ 

$$\frac{L-L}{L} = \frac{\frac{F}{2A}}{\frac{1.06 \text{ q}}{2A}}$$

Fi = 2Aq(L-i)(1.06)The above three equations contain three unknowns:

F, q, and 1. Substituting the appropriate values produces the following set of equations.

By substituting equation (1) into equation (2), we obtain

$$M = qb1/2 (L-1/3)$$

$$q = 2M/b1(L-1/3)$$
(4)

Similarly, substituting equation (1) in equation (3) gives:  $\frac{1}{2} qbt^{2} = 2A q(L-t)(1.06)$   $bt^{2} + 4.24At - 4.24 AL = 0$   $t = -4.24A \pm \sqrt{(4.24A)^{2} + (4)(b)(4.24AL)}$   $t = -2.12A \pm \frac{1}{2b} \left[ (4.24A)^{2} + (16.96ALb) \right]^{\frac{1}{2}} (5)$ 



t = 1.39 inches

or:

2-14





Now, using equiation (4), the contact pressure is found:

$$q = \frac{(2)(231270)}{(5.5)(1.39)(5.5-1.39/3)} = 12012 \ lbs/in^2$$

Using equation (1), the bolt load is found as:

$$F = 1/2 \text{ ab1} = (12012)(5.5)(1.39) = 45915 \text{ lbs}$$

Bolt tensile stress if found as:

Maximum Normal Stress Theory will be used for evaluating the effects of the combined stresses (Tension and Shear) on the bolts. From Mohr's Circle the Maximum Principle Tensile Stress is computed as:

= 46178 rsi

Therefore the factor of safety for the bolts is

F.S. = 
$$\frac{92000}{46178}$$
 = +1.99

Bearing safety factor versus yield is found as:

F.S. = 
$$30000$$
 = +2.50  
12012



• .

Revision O



c. Lug Stresses

Figure 2.5.1-1 depicts the lug configuration. The outstanding lug plate is attached to the flush plate with a vertical double vee weld. Net section is equal to the 1" plate thickness of the outstanding lug plate. Shear stress is:

> fv = <u>35580</u> = 2737 psi (13) (1)

Factor of Safety is:

F.S. = 
$$\frac{30000}{\sqrt{3}} = +6.33$$
  
2737

Using a conventional  $40^{\circ}$  tearout relation, the capability of the lug at yield stresses is estimated as:

 $P_A = 2 F_{syt} (E_m - d/2 \cos 42^\circ) = 127,689$  lbs.

Where:

 $F_{sy=\frac{30000}{\sqrt{3}}} = 17,320 \text{ psi; ASTM-A240, Type 304}$ t = 1"  $E_m = 4-1/2$ " d = 2-1/8"

The factor of safety versus tearout is therefore:

F.S. = <u>127689</u> = +3.59 35580





# FIGURE WITHHELD UNDER 10 CFR 2.390



Bending stress in the lug is found from: fb = Mc/IM = (?) 1 $I = 5h^3/12$ For the lug and weld: b = 1 inch h = 13 inches 1 = 5 - 1/2 inch c = h/2= (6) (35580) (5.5) = 6948 psi  $f_{\rm h} = {\rm Plh}/2$ = <u>6P1</u> bh<sup>3</sup>/12 bh<sup>2</sup> (1)  $(13)^2$ The associated factor of safety is: F.S. = 30000 = +4.326948 2.5.1.2 Reinforced Lug Plate

The removable lug discussed in the foregoing section bolts to a flat reinforced lug plate of 2 inch thickness. This lug plate is attached to the cask outer shell and integral ring with a 1" circumferential bevel weld.





1. Shear Stress - fs = P/A A = (Perimeter) (Effective Weld Thickness) = ((2) (8) + 2(5.5))(1)  $= 27 \text{ in}^2$   $f_s = (35580) \text{ lbs} = 1318 \text{ psi}$   $27 \text{ in}^2$ Safety Factor =  $30000/\sqrt{3}$  = +13.1 1318 11. Bending Stress -  $f_b = Mc/I$ 

M = (Px1)







Revision 0 I = I outer - I inner =  $\begin{pmatrix} 3 \\ \frac{bh}{12} \end{pmatrix}_0 - \begin{pmatrix} 3 \\ \frac{bh}{12} \end{pmatrix}_i$ b<sub>1</sub>= 5.50 in. b = 7.50 in. 2 h = 10.00 in. h<sub>i</sub>= 3.0 in. 1 = 7.0 in. c = hc/2 = 5.0 in.  $I = 1/12 \left( (7.50) (10)^3 - (5.5) (8)^3 \right) = 390 \text{ in}^4$ f = (35580)1bs (7.00) in (5.00) in. 190 in. f = 3193 psi Safety Factor = 30000/3190 = 9.42.5.1.3 Outer Shell and Integral Ring

The 1-13C II cask is designed with a reinforcing ring 8 inches high, 1 inch thick, located directly opposite the lifting lugs on the inside surface of the outer shell.



P = Radial Force

- M<sub>L</sub> = Moment in Longitudual Direction to Shell
- M<sub>C</sub> = Moment in Circumferential Direction to Shell



It will be necessary to determine the stress generated in the shell due to the loading transmitted by the lifting lug. The presence of the inner reinforcing ring will be accounted for using STARDYNE, see Section 2.10.2.5, finite element idealization of the shell and ring as shown in Figure 2.5.1-2. The model represents a single quadrant of the outer shell, ring and reinforcing plate, or pad. The contributions of inner shell and lead shield are conservatively ignored.

Symmetric boundary conditions are imposed upon the planes of symmetry which slice through the model at the center line of the lifting pad and  $30^{\circ}$  away from this location.

Loads are introduced by a set of downward load applied at the lower edge of the outer shell. These loads are reacted by vertical restraints at the center line of the lifting eye.

Figures 2.5.1-3, -4. -5, and -6 define the model geometry completely. Figure 2.5.1-3 presents the developed surface of the cask outer shell quadrant shown in Figure 2.5.1-2. Figure 2.4.3-4 presents a detail of the reinforced pad installed in the cask outer shell. Nodes 66-69 represent the points where the lift lug forces are transmitted to the outer shell-ring mode. Figure 2.5.1-5 presents the idealized (symmetric half) lift lug, with node 74 corresponding to the center of the lug eye. Figure 2.5.1-6 presents the developed surface of the reinforcing ring. Figures 2.5.1-7, -8, -9, -10, -11, and -12 present computer generated plots of the model visually verifying the example.

Figures 2.5.1-13, -14, -15, and -16 summarize states of stress of all finite elements surrounding the reinforcing plate or pad. Computer stress output for the entire model can be found in Section 2.10.4.





Figures 2.5.1-13, -14, -15, and -16 illustrate a predicted state of stress consistent with expected outer shell and ring behavior. The offset lift lug load induces torsion in the ring. This torsion causes the top edge of the ring to deflect inward with a corresponding outward deformation of the lower edge of the ring. These deformations induce a concave deformation of shell elements above the ring and a convex deformation below the ring. As expected, the concave deformation pattern induces compressive bending stresses on the shell outer surface above the ring. Similarly, convex deformations induce tensile bending stresses on the shell outer surface below the ring. The specific states of stress in the shell elements immediately above and below the lug pad are found from Figures 2.5.1-13 and -14 as follows:

	hove (F1 68)	
Membrane <sup>(1)</sup>	ABUVE(E1: 68)	Below(E1. 69)
S <sub>x</sub> (Hoop)	-2137	+ 903
S <sub>y</sub> (Meridian)	-2018	+7103
s <sub>xy</sub>	+ 536	-1313
Bending <sup>(2)</sup>		
s <sub>x</sub>	-8374	+5805
s <sub>y</sub> .	-3970	-2040
s <sub>xy</sub> .	- 958	- 802
	_ 1 1	

HOTES: (1) 
$$S_{M} = (S_{o}+S_{i})/2$$
  
(2)  $S_{B} = (S_{o}-S_{i})/2$   
(3)  $S_{o} = Outside Stress$   
 $S_{i} = Inside Stress$ 



## FIGURE 2.5.1-2

# RING STIFFENER OUTER SHELL







# FIGURE 2.5.1-3









FIGURE 2.5.1-4

# DETAIL 'A' BOLT PAD R = 19"







FIGURE 2.5.1-5

# DETAIL "B" LIFT LUG (IDEALIZED)



NODE 74 IS THE CENTER OF LUG EYE



FIGURE 2.5.1-6

DEVELOPED SURFACE - REINFORCING RING

(R = 18.25")









.

.



FIGURE 2.5.1-7



COMPUTER GENERATED MODEL PLOT







FIGURE 2.5.1-8





COMPUTER GENERATED MODEL PLOT



2-29

Revision 0



¥. -9.37

COMPUTER GENERATED MODEL PLOT

10.87

7.12

1.17





Revision 0

22.12

18.37

14.42

26.17



.

FIGURE 2.5.4-10

٠



COMPUTER GETTERATED MODEL PLOT









•

FIGURE 2.5.4-11

•



COMPUTER GENERATED MODEL PLOT

•





.

•

FIGURE 2.5.4-12



COMPUTER GENERATED MODEL PLOT









The predicted stresses all assume a package weight of 23,720 lbs. This weight does not include 3280 lbs. of energy absorbing overpacks. The package is never lifted with overpacks installed; this is geometrically impossible. Thus, the maximum package weight, when lifted, cannot exceed 23,720 lbs.

The maximum stress in the cask cuter shell in the region of the reinforcing ring is:

f<sub>max</sub> = 14,930 psi (Quad 15)

The maximum stress in the reinforcing ring is:

f<sub>max</sub> = 18,670 psi (Quad 63)

The maximum stress in the cask away from the reinforcing ring is:

 $f_{max} = 21,870 \text{ psi}$  (Quad 43)

Quadrilateral element 43 is at the base of the model where point loads representing the total cask weight is applied. The model neglects cask endplate reinforcement existing at this location. All stresses in this region of the model are considered to be fictitious, although acceptable, and are not utilized in subsequent margin of safety calculations.

Under normal conditions of transport, the maximum temperature of the outer shell is  $188.7^{\circ}F$ . Therefore, the interpolated yield allowable in 304 avrilable (ASTM A-240) is:

fallowable = 25,565 psi







The resultant minimum factor of safety versus yield in the outer shell and integral ring is:

.

 $F.S._y = 25.565 = +1.37$ 13,670





٠

















Revision O

FIGURE 2.5.1-15

STATE OF STRESS - REINFORCING RING OUTER FACE ( + 2)





FIGURE 2.5.1-16

STATE OF STRESS - REINFORCING RING INNER FACE ( - 2)





1

Section 2.5.1.2 provided a nominal evaluation of lug plate attachment weld stresses. The finite element analysis discussed in this section provides a basis for detail evaluation of ring attachment weld stresses. Referring to Figure 2.5.1-6, a 1/2" vee-groove weld joins the ring to shell along node lines 76-82 and 83-89. A 1" vee-groove weld attaching the ring to boss extends from nodes 76 to 83. Corner forces applied to the ring at the node points by the welds are as follows:

•	•	Corner 1	Force (Globa	al Coord)
Joint Node	9 (See Fig. 2.5.1-6)	FX1 (Lbs)	FX2 (Lbs)	FX3 (Lbs)
76,	8.29280	-8075	-7437	-14449
έ8	8.29280	11033	-11516	- 9969
77	15 <sup>0</sup>	-3670	18618	- 5320
84	15 <sup>0</sup>	4277	-12832	- 1995
78	30 <sup>0</sup>	-8509	-17526	- 2602
85	30 <sup>0</sup>	6318	- 9818	· 1028
	·			

Running weld forces may be computed by transforming the global coordinate values to local coordinate values using the polar angle 9. The transformation is as follows:

$$\left\{ \begin{array}{c} F \\ F \end{array} \right\}_{L} \qquad \left[ \begin{array}{ccc} \cos\theta & \sin\theta & 0 \\ -\sin\theta & \cos\theta & 0 \\ 0 & 0 & 1 \end{array} \right] \quad \left\{ \begin{array}{c} F_{\chi} \\ F_{\chi} \end{array} \right\}_{G}$$

Shear and tension load per unit weld length are found by considering total in-plane and out-of-plane forces:

$$F_{S} = (F_{L_{2}}^{2} + F_{L_{3}}^{2}) / t , lbs/in.$$

$$F_{T} = F_{L_{1}} / t , 2-40$$





Node		1(in)	
76/83	4"+18.25 $(15^{\circ}-8.2928^{\circ})(\frac{\pi}{180})(\frac{\pi}{180})$	5.068	
77/84	$18.25 (\frac{\pi}{180})$ (22.5°-11.646°)	3.457	
78-85	$18.25 \left(\frac{\pi}{130}\right)$ (15)	4.778	

•

Where: 1 = effective weld length per node as listed below.



The corresponding weld forces are found as:

•

	WELD FORCES (1bs/in)					
Joint	Tension	Shear	Effective			
(Node)	(F <sub>T</sub> )	(F <sub>S</sub> )	Shear (F <sub>V</sub> )*			
76	-1788.	+3102	3228			
83	1826	3230	3357			
77	368	5689	5692			
84	234	3948	3950			
78	292	4103	4106			
85	118	2450	2451			
*F., = [	$*F = \left[F^{2} + (F_{m}/2)^{2}\right]^{\frac{1}{2}}$					
*F <sub>v</sub> =	$F_{g}^{2} + (F_{T}/2)$	2 5				



í.

2-41

.

••




$$F_{VA} = \frac{25565}{3}$$
 (1/2) = 7330 lbs/in

The factor of safety in the weld is:

$$F.S. = 7380 = + 1.30$$
  
5692

Computer generated corner forces and plate stresses is provided in Section 2.10.3.







2-43

.

1

.





.

#### 

.

.

JG NASE HIDE	L ዓ አ ስ			
	9 H A D F I I A I	ны стыссе.		
Ronts	ELENENT	tifneut	C 2 AND ADACT2 FUR ONLAND ALCINE	1
	FINCES	6167EN1	581HC1586 71817712	
		21862262	MAX MIN MAX SHEAK	ANGLE
12. 1	Ft + 1.2561102 1161		• • • • • • • • • • • • • • • • •	• • • •
		LF 31 + -1,174E+04		
. / 61110 / >		3Y = -1.613E+01 + 11+	ACE -1.30A(+0) -1.4010+04 6.311(+0)	- #2 . 244
		-517+ -1+676[+0]	VM 1.337[+04	
	n# * **********************************	CE SX + 1.5x9E+Dx		
	FT	SY + 1.14=1+03 -1F	ACE 1.3606+04 1.0358+03 7.2828+03	3.013
1,	· FIY6.149E.01	SAT+ 1.275E+03	······································	
		•		
17. 2	- F4 + 21824EF05 +5ET	CL SI + -7.015E+03		
• ••	FY - 5.10/[+00	57 + -K. 124F102 125	IFE 1. WILLON -4 GASCAGE & PARCAGE	-1.1. 111
. / QIND / 1	1171.6161+02	\$174 -6.1676401		- 0 ) 1 / 1 /
. / 2/	HI1.118(+02 -151		*** *****************	
./ / 10	MT2.7665401			
9,11	HIY a al ADICION	31 = • 6, 1416102 • -24	A6E - 9+419E+D} -7+275E+02++5+0731+03+	-21.818 -
	1-11.1+37302	2X1+ 7*2018+03	¥7. \$. \$03E +03	
12. )				
		LE SE + 7+305E+02	•	
	• • • • • • • • • • • • • • • • • • • •	5Y - P.048E+02 +2F.	ACE 5.602E+03 -4.061E+01 4.#33E+03	-13.233
	FIT1,041E102	\$111.11[+0]	YA 8,404E+03	
• • • •	MX 3+141E+01 -2FA	CE S# = -#+4826+02		<b></b> · ·
./ / 11	FX + 2+65HE+01	SY4.5166+02 -2F	ACE 3.723E101 -5.074(103 4.199(10)	46.455
[D)]	7771.922E+02	SXY+ 4, 1946+01	VA 7.450F101	
12. L	FT6.612E+03 +2P-1	1. 51 . 5.6535103		
,,	FY + 6. 1821 + 01			
. /0010 / 5	FTT +		ACC 1.0136401 -0.0406402 4.2416403	- 10.840
. 1 . 41			······································	•• ••• •
		3T1.2811403 - 272	ACE 9.661E+02 -8.133E+03 4.360E+03	60.234
	H11 + -1.14/E102	211- 3.930[10] .	• • • • • • • • • • • • • • • • • • •	•
•• •	<b>**</b> • • • • • • • • •			
• • • •	11 • 1.347[100 +/FA	(f 7) + 1*005(+0)		
	10+162.54	• 5 <b>7 • 1.0</b> 8668403 +287	ACE - #+689E+03 3,144E+03 3,745E+03	-17.625
• / 0 U I D / •	FIY + 7.1018+01	521+ -2.1A1E+03	¥M 8.136E+03	
. / /	#3 + 3.333E+02 -2F4	([ \$1 + -7.9976+0]		
./ / //	PT + A.0711+CL	ST + -1.486E+03 -2FF	LCE -1.1181.001 -0.2666.001 1.4741.001	70.474
12>1	MIY = -9.5971+01	517+ 2.5556103		
32. 6	FE - 4.8472+03 0/FA	E SE - 1.4291103		
	FT + -1. 7005+C1			
10040 / 7	FTT a 2.4332 etc)		er #'AAAfen] 1'nonfen] 1'einfen]	-3.801
		2410 -D.9781401	AN BUIGLIANS	
		- 318.7131103		
	M4 • 4•1001101	ST2.2411+03 -2F	LCE -2.144[+0] -0.824[+0] 3.322[+0]	83,077
[],/	FIV + -).110(+01	517= 7.951[+02	VH 7.9651103	
	<b>.</b>	•	•• • •	•
¥2. A	TT + -+.++21+02 +7EA	1 St + -1.17#2+04	•	
• • • • • • •	FT4.C+>1+D2	577.340E+03 +1F1	ICE -6.6351+03 -1.2181+04 2.9231+03	-64.670
• 10110 1 9	144 + -5"2(/)+(5	ショナー ート・タリンド・リト	VN 1.0026+34	
. / 1/	- MM - + - ++: 1/J +6221 A	4 SI + 9,938E+03		
./ / \/	MT + -2.5448+02	<pre>\1 - 4.421/103 -2/2</pre>	ITE 1.023E+04 5.697E+01 2.76#E+03	11.742
13	#14K.ZFAL+01	SIT- 1.103[10]	YA 6.6671+03	
37. 9	17 + -1.057F+02 +1F40	1 St + -1, P/PE+03		
<b>, ,</b> .	17 • 3, 2391 (13)	ST + +2.1451103 +273	WE I YPE IN - II JUSTICE & JUSTICE	
10000 1 12	• • • • • • • • • • •		····	



.

۱

2-44



1611	MTT1.616E+02						
	· · · · · · ·		1.1.c.(10)		YM	5.1521.03	
15. 10	F# + -5.074F+UZ	+7 f A CL	516.1191.02				
• ••	17 . 5.0331102		ST = 1.200(x0)				
. /OLAN / 11	117 8.9111+01			111-11 3.108		<b>N</b> 1 ( N ) ( N )	430.074
. / 4/	HI - 2.257[+0]	-1110)			ý A	0.2271.01	
-/ / !-	PT = 1.1065.00		ST = 1,1,1,1,0				
17	HTT + -1.8721+02				(+1) = (+1)(-1)(+1)	4.5201.03	51.659
	•				V/I	7.4316+01	
¥7. 11	FT + -5+141F+C2	178461	51 - 1.430(103)				
	FT + 1.6316+02		SY = 9.4636403				
. /01145 / 12	FTY + 5.7301+01		SIX = S + SO(20)	·//···································	$(0) = 2 \cdot 2 \cdot 2 \cdot 0 \cdot 1 \cdot 0 \cdot 1$	3.5511.01	- (1,12)
. / 10/	M4 1+25 16+02				Y n	5.5721103	
./ / 19	PT + 2.5636+01						
18	HIT + -1. 360F+02			-17401 2.178	E+U1 -6.50/(+0)	4.3181+03	55.055
			311- 310/42/01		۷n	7.793[+03	
72. 17	FF + -7.1571102		11			•	
	IX + +A. 0761101		3				
· /9//JO / 13	FIT + A. CIALADI		51 - 8.465[102	117ACE 4.389	E+0] -5.750L+02	2.5421+03	-31.644
. / 11/		-1115	STI / JOBE ( 0 )	• ••• •• ••••• • •	¥N	4+4056+03	••
./ / 20		-//	21 • -8.034[10]				
19	HTY9.450(10)		511.1721+03	-11216 -1+374	[+02 -1.063[+03	3.463[+0]	67.268
••••			SP14 24468EFUS		· • • • •	F*d421+D)	
12. 11			•• • • • • •				
		111110	51 + 3.764[+0]				
		••••	· 54 •·· 1*155F+03	+ 1 F X C E · · · 3 . 98 6	£+03 0.444402	1.5436401	-15.564
	FIT - 4.044[+00		5141.9801102		V //	3-4511+03	
		-1116	517.1661+03				
	PT = 3+444E+01		ST = -1.515E+03	- LFACE -1. 3761	L+03 -7.2451+03	.2.7456+01	61.012
10.0001	H117-745fr01		214- 4-305(+05		7 M	6.4476+03	
12. 16			• • • • • • • • • •				
-21 10	FF - 11634[+02	+TFACE	584+7298+03	· · · · · · · · · · ·	•	• •	•
	FT = 1+1112+03		SY = 7.3EUE+02	+2FACE 1.151	E+04 -1.530L+04	1.1501+05	-50.834
. / 01/11 / 1/	111 - 1.266[+0]		SIT1.322E+04		٧٨	2.3476+04	
	PX = -1+B20E+03	-7F XC F	SI - 4.977E+03		•		
.1 1 2.	AT = -1. (941+CO		57 + 7.44#F+02	-2FACE 1.792	+04 -1.220L+04	1.3061404	10.461
/]·;	HIY\$.275[+03		2X7+ 1.491E+04		414	2.6246+04	
	• • • • • • • •	•• • ••			• •		
32. 17	FY - 2.240[10]	+25461	SX1.684(+03				
• • • • • • • •	FY • 1.7171+01		STB.704E+01	+2FACE 1.141	E+04 -1.319E+04	1.2306+34	- 16.061
//UFD / 14	FIY = 7.650F+02	•	3171.2276+04		۲M	7.1321.04	
. / 15/	PT = -2.3711+02	-)fxCE	51 - 1.714[+03				
./ / >;	MA3*6231+01		SA = 1.0A0E+05	-LFACE 1.424	L+04 -1.2411404	1.1321.04	\$3.215
24,111	PIY + -4+7+51+03	••••	5×1+- 1+330E+04	• • • •	. VH	2.3101+04	
						-	
17. J#	FX1.544E+02	+ 1F AC C	SX - 1.500E+01				
	FA = -5*350F+05		SY = -1.871E+02	+7FACE 1.076	E+01 -9.4172+03	1,0108+04	-42.005
. /0HAN / 14	FXY = 3.251F-01		SXT+ -1.0071105		YM .	1.7516+04	
. / 16/	HX + 6.0215102	-///	SX1.712(+0)				
1 1 24	MY1.1961+01		541*5336+05	-2FACE. 9.142	E+01 -1.102E+04	1,010[+04	41.250
25	PIY + -7,7761+C3		5×Y- 1.007E+04		44	1.7521.04	
77. 10	F1 = -3.2101+U2	175265	51 - 3,4771+03				
	50+1565.1- + TI		<y3.4aue+01< td=""><td>+21 ACE 4.0021</td><td>E+01 -5.0591+01</td><td>7.7101+0)</td><td>-36.112</td></y3.4aue+01<>	+21 ACE 4.0021	E+01 -5.0591+01	7.7101+0)	-36.112
. / (11/1 / )()	1772.3751.07		517F.73HF+N1		P-V	1.2351+05	
• / 1//	FT + 1.5771+03	-7+2(+	37 4.4067+03				
	FT + 7,1241101		571. 11 C2	-78866 4.4531	1+03 -4+0141+CJ	6.7501+03	51.201
78,*1	*** * *2.5721101		VAL: 1. Martica		VH	1.1+41+44	

Revision O

:





:

/ 187	71 - 2.091E+03 - JFACE	515,1521.		
•/ / 2A	HY + 4.571F+01	ST = -6.927F101	-1FACE 6.2906+02 -6.6006+01	3.6346403 71.249
27,	HXY + -++6111+02	SIY. 2.211F+01		1.9441401
12. 22	13 + A.2045+02 +75+CC	SX	<b>-</b> · ·	
10010 1 23		ST = 4.452101	+()ALE 1.125[+04 6.311[+03	2. (77(+03 -)7.11)
		SXT2.34[E+0]-	e man a menera a correcte a VN-	9.7711403
	NX + 2+699E+02 -7FACE	51 + -4,4372+03		
•/ / <u>)</u> 0	PT + 1.217E+02	SY - 3.606[+03	-1FACE 3.613E+03 -1.614L+03	4.2296+03 84.337
29	MXY 3 . 473F+01	528++ 2.4575+02	. V#	1.1501.03
•				
xz. 7)	FX + 9.300F102 +1FACE	51 + 2.769[+03		
, ,	· FY • - 1.4016403	SY 3.2401403	1/FACE -0.1606401 -2.1396401	5.151(103 -44.)62
. /otian / 24	FTY = -9, 1995102	STY5.1476+03		0 121(10)
1 207	NX A 3.7841401 -TEACE		14	4. 41 ) [ * 0 /
	HY = 1010(101 - 17x)(100)			
10		31 - 2.3.31.03	-11112 3.2712103 2.2361101	1.624(40) 37.686
Ju,,,	HIT = -1+37HE+02	2114 144686403	YN YN	3.2601.03
22. 24	TI5.144E+02+2FACE-	3x5.043E+02	······································	•• • •
• • • • • • • • • •	Y = -2, AD1E+02	SY1.510E+01	+1FACE 4.138E+03 -6.173E+03	3.1632+03 -42.206
• /QUAN / 25	FXY4.934E102	5x7+ -3.141E+03	Yn	9.003[+0]
• / 21/ •	MI + -K.384E+01 -IFACE	SX + 2.562E+03	· .	
./ / 12	MY + -3,954F+01	SY = 3.898E+02	-ZEACE A. 824E403 -1.872E403	1.1501.03 32.535
32	811 + -1.7111+07	SXYn 3.1675103	VA	5.9855 101
12. 24				
		37		
		3T1.5JBEIUJ	+114CF 5-413F+03 -2-3/46+03	4.131(003 -52.793
. /0010 / /6	FIL • -1.404E1C2	SXT= -4+1191+03	···· • • • • • • • • • • • • • • • • •	7.2366+03
. / 22/	NY + -7.236E101 -/FACE	57 - 2.5736+03		•
./ / 33	PY + −2.005€+01	SY = -5.735E+02	-2FACE 4.706E103 -2.706E103	3.7066+03 32.439
)2X]	#XY -+ 1,557E+02	-SIY 3. 356[+03	·· ···································	6.4976403
37. 21	1X - 3.376F102 N7FACE	SX = -5+1386+02		
	5Y = -8.917E+01	SY	N78108 2.4078401 -2.4148401	2.71.55403 -58.113
101.10 / 17				1 7105403
			***	41110(10)
• • • • • •		33 • 2.1511.03		
•1 1 35	· NT · • · [+057E+01 · · ·	51	C + ALE - 4, 02 IE (03 - 1 + 70 4E + 0)	210032403 201101
33,,31	MIY = -1.059E+02	XXY+ 2,3651+03	NY .	5.095[+0]
¥2. 27	FX = -6,93#E+02 + 17FACE	5x = -2,390E+02	• • •	
	FY + 3.150E+02	ST = 1.305E+03	+2FACE 1.753[+0] -6.8728+02	1.2206+03 -64.624
. /0440 / 20	FXX = -2.256E+01	SXY= -9.449E+02	VN	2.1806+03
. 1 751	NX + -6.7771+017FACF	·SX-= · 3.014E+03		
/ / 35		SY = -1.1925+01	-15165 1.2175+01 -2.6476102	1.7518+03 14.610
		57 = -100000000000000000000000000000000000		1.1776403
34,,,	HIL3-1441-01	371- 013472402	111	3())))
PS. 29	1x + -4.3276102 +75306	5x = 1,0111+0x	•••••	
	E0+3420+2 + X1	57 + 7.1636+03	+7FALE 1.074E+04 6.532[+03	7. (4)(10) -77. 192
• 101-40 1 30	20+1004+2- + 111	2114 -1+205(+03	· Vn	4.3121.03
. / ?*/	NY - 4.577[+02 -2FACF	$2x = -1 \cdot 15 \cdot 1 \cdot 04$		
.1 / 27	PA + 1*541F105	SA + 8"31nC+05	-214CE \$.520E+02 -1.1058+04	6.4021+03 88.040
36,,*1	PIT + -4,014[+01	384- 4.5646105	V/I	1.2381+04
-				
17. 30	FX + -2.440F102 +2F4CF	51 + 5,1426+03		
	FY + 1.340E+03	SY + 4.2001+03	+2FACE 0.7565+03 5.9661+02	4,0011+01 -41,713
10000 / 31	ETT + -6.4761102	511+ -5.0546103	VN	0.4741+03
1 10110 1 1	NY	ST	•••	
	NY A ALIMATIC TEPPES	$\nabla Y = 1 + 1 + 3 + 1 + 1 + 1 + 1 + 1 + 1 + 1 +$	-TEACE I LINEADI LUITADI	1.9186401 19.014
	the second states and second s		······································	· · · · · · · · · · · · · · · · · · ·
				· •

2-45

:

.

t

# Revision O



N 7 õ

> 3 levisi

O

1 37. 41 11 - 4-1341462 . - - - - - . . . /0140 / 47 +++ + -2.2311+61 V4 1.1251+UN 317- -1.1371+01 . / 307 MT - -1.6646162 -/ FAGE SF - 1.2306104 .1 / 14 PT - -1-3+71+02 7.357 57 - 4.1578+03 -7/268 1.2448+04 4.6148+63 4.2108+03 <s.-----MIT - ----317- 1.06+t+03 VA 1.1001+06 12. 43 11 - - 3. 0101+62 1/FACL SX + 1.4742+04 +\* + N.L3L1+02 . - - - - . SY . #.4371+03 +/fact 1.405t+04 #.327t+03 5.764t+03 -1.10# /CUAN / KK + +1 + -4.7A71+61 • 547+ -1.1211+03 Vn 1.7276+05 . / 37/ PF + 3.4781+02 -7FACE SX - -2.0451+04 .1 / 51 PT - 2.7461+62 St + -4.4#28+63 -2/1(1 -4.4446+03 -2.6446+64 4.0146+03 07.384 :0.----. FIT - -3.#36[+01 - VE 1.4001+04 STT- 7.2466+02 . 32. 44 FF - -1.440F+02 +2FACE 5X - 1.371E+04 -• • • • . - - - - . FT - P.1361+02 ST - 6.313[+03 +2+4CE 1.446[+04 5.164]+03 4.8462+03 -20.147 . /01/40 / 45 FIT - -2.5441+02 317- -3.1346+03 VA 1.306[+04 . / 3#/ ## + 3.#728+62 -284C8 58 + -8.44#8+04 -./ / :7 PT - 1-9521+62 57 - -3.0561+03 -11+1C1 -2.6862+03 -1.4451+04 6.0842+03 74.473 51.-----F#1 + -1.0YOL+02 SIT- 2.0466+03 YH 1.1/11+05 12. 45 . .----. ST - 2.804[10] +///// /.054[10] -1.055(10) 4.457[140] -41.144 11 - 7.048E+02 . /01140 / NL Far - - 3- 2361+02 · SIT- -4.4171+03 VR #.436E+01 . . / 14/ P1 + 1.4421+02 -2FACL SX - -4.122(+03 1 / 13 PT - 3.7471161 ST - 3.5388+61 -2+4CE 1.7088+03 -5.744+03 3.7518+03 21.827 52, ----- , ---- 11 MIT - -1.5711+02 VA 6.4111+03 ---SIT- 3.1221+03 ---32. 50 11 - 1.6461+02 +2FACE SI - -6.346E+03 . - - - - . 17 - 5.984E+02 . JOUAD / NT 111 - -Z.7371+02 517+ -4,5258+03 VN 9.6631103 . / ../ MR - -2.411[+02 -1FACE SI + 6.445[+0] ./ 1 54 PT - - 0.720[+0] 5Y -- J.292[+0] - - 2FACE 6.672[+0] 1.104[+0] 3.704[+0] - 32.522 53.----. 11 FIT - -1.658(+02 YA 0.1766+03 517- 3.4316+03 47 12. SY - -3.689E+03 +1FACE -2.676E+03 -1.476E+04 6.041E+03 -13.164 .----, 17 . 5.4651.02 • . /01130 / 4A FTT - -1.591F+02 YM 1.362E+04 SIT- -3,349E+03 . . / . . . / ST . 5.#751+03 -2FJCL 1.5282+04 5.0431+03 5.0431+03 10.041 ./ / >> NT - -1.9431+02 MIT - -1.2636102 VA 1.3472+04 SXT- 2.713E+03 . >2. 4.6 FT - 2.1101+02 +2FACE SX - -1.7781+04 ST - -5.0798+03 +2FACE -4.9622+03 -1.7892+04 6.4661+03 -84.532 . - - - - . 11 - 5.5071+02 . /01/0 / 49 FIT - - 4.734E+01 -511+ -1.227[+03 -- vn 1.600E+05 . / 52/ ## + -7.5#3(+02 -1+4CE \1 + 1.#62E+04 ST . 7.2821103 -111CE 1.8721+05 7.1881+03 5.7641+03 ./ 1 36 PT - -2.5711+02 5,185 VA 1.635E+04 -M#T - -4.7171+01 SXY+ -- 1.0371+03 - --F1 - -2.4941107 IFACE S1 - 2.322E104 17. 50 . .----. - SY -- 5,668E+03 +7FACE 2.328E+04 5.608E+03 8.838E+03 IT - N.8H3E+02 ~1.151 . /0110 / 51 VM 2.1051+04 117 - ->.07u1+00 511+ -1.031(+03 . / \3/ MI - N. BBAE+D2 - JEACL SI - -2.4228+04 ./ / >\* PT - 1.74×1+02 57 - -2.4156+03 -1+1CE -2.86/6+03 -2.42/6+04 1.0/06+04 87.204 57.----11 F1Y - -4.2558+01 517- 1.0111+03 YA 2.2471+64 32. 51 FT - - 1. ALDEND2 +/FACE St + 1. A62E+04 - - -. .----. 57 - 4.452E+03 +2FICE 1.728E+04 3.744E+U3 6.738E+03 -12.718 FT + 6.936[102 . /00/0 / 52 VK 1.572L+05 FFT + -1.+73[+01 5=7= -2.8446+03 · / · · · / rt + 7.0776+07 -714CE ST + -1.7351+04

71 evision

Ч į, - 1

0



• • •							
./ / 39	NT - 1+2771+02	57 * -1+4 .+01	-15465	-1.1666.01	-1.7851-005	8.1271.03	80.105
<b>30</b> ,	MIT1.140L+02	Stra 2 Billion				1 1141.0.	
					10	1. 1641 101	
12. 57	FT == - 24BLADL	11.10. 1					
		2FALL 34 + 343471103					
101110 1 13	6.9751102	57 + 2.349[+0]	174 AC E	8.3916+03	-5.551102	5. N18F103	-31.565
1 /01/01/ 31		527+ -4.1201+03			٧n	0.1.221+03	
• / • • > /		1FACE 5x5.744E+03					
•/ / ٨0	MT - 4.2241+01	57 - 3.7115+02	-75465	2.1275+01	-7.7551401	5.0511.01	1.1.01.1
391!	MIT + -1.6411+02	517- 1.9476+01	• • • • •		V	4 1616401	
	•				•.,	*****	
12. 53	13 - 7,1551+01 +	11166 58					
	JT + F.#C51402						
. /2148 / 55		31 • 1.346.102	• / I ACE	5.5341.03	-0.733[+01	2.24411403	-45.057
. /		·· SXT+ -4.316[+0]		• •	A 14	1.3041+04	
		//ice si + /					
	nt • ->•043[+0]	ST + 2.5HJE+O3	-1+xce	9.428[+0]	7.4111+01	4.6771+03	31.173
ru	20+J(37+I- + YIM	527- 4,1448+03			Yn.	4.3421+03	
					•••		
72. 55	11 + 1.6101+02 +	78468 58 4 -1 4414404					
	17 · ···· 6.6721.602						
. /00110 / 55			. ALLACE	-1+0125+03	-11/101101	8.143[+03	-78.167
1 17/		3473,270(+03			V N	1.4#28+04	
• • • • • •	F11.0/4(+02 -	IFACE 33 + 1.734E+04					
+/ / 52	MY1+264E+02 -	ST - 4,3A4E+03	-IFACE	1.8058+04	3.6511+03	7.2031+03	12.901
(171	PIT1.3341+02	SIT+ 3.135E+03			¥ A	1.6551105	
					••••		
2. 55	FT = - 2,5011+02 -+	7 FACE ST	<b>.</b> . <b></b>				
	ET + A.AOACAO2						
· /01140 / 56			• <i>1</i> • <i>A</i> • C •	-2.6276.03	-2.2201104	9.7851.03	-06.422
. /		2114 -1-5146403			V/1	2.1016.01	
		LYACE 34 - 7.3C01+04	• • •				
., , , , ,	-1.0//1.02	5T = 5.345E103	-25466	2.3166+04	5.264[+0]	.8.9431.03	3.755
D7,11	PIT + -4.975FID1	SXY- 1.1648+03			¥ R	2-10/1-05	
	• • • • • • • • •	• • • • • • • • • • • • •	· •• •	·•• · •	• •		
7. 15	FT + -3.301E+03 +	EFACE ST + -7.065F105					
	FT + -7-130F+01	ST	175105	-1 1445103	-7 2761 101	0 4876403	- 10 235
. /01/20 / 64	FIT		· · · · · · · ·	-313066103	-1.110(101		-101733
/ 10/					**	2.12/11/04	
• • • • • •		(FACE ST + 1.735,+04					
., , , ,	HT + -1.7A1(+03	ST + -1.653č+03	-lface	1.1316+01	-1.107[10]	4.6576+03	5.126
fw	- FIT7,564E+C3 · ·	517+ 1,7146+03	<b></b>	• • • •	• • • •	1,0101104	
77 <b>.</b> EM	fx + 3.319[+03 +	7FACE 51 + -1.431E+03					
	FT + 3.7:AE+03	ST + 8-11/F+01	*****	9.1505+01	-1.1641401	6 2326101	- 11 107
A / CLUAD / AA	F 17 + 3.070F+02	517 - 1 2246403			Val	1 1216+04	
. /	PT = =2.1041463 =				• •		
	FT - 413261403	ST 4.86CF+03	-1116	010041403	-0.5136+03	P**PAF+03	14.304
[91]	FIT = -7.389[+0]	511- 4.0366103			vn	1.1211+05	
	·						
72. ++	FX + 1.7861+01 +	LEACE ST + 1.112E+04					
	11 - 2.1271+03	ST + -8.298F103	+75365	1.1521.05	-8.5911+01	1.0001.004	6.960
. /011PD / AL						1 1 1 4 1 4 4 4	
	1371,8368+03	<b>\   1 • 7. \ </b> DAF FO •			V M	1 . / 1	
. / •1/	1171.836F+03 21 - 6.419F+03 -				¥1	1.7340.00	
• / • 1/	FIY1. F3AF+03 FI - 6. 419F+03	7FACE SI4.336F+03	_ 11	1 1405.4.	-1 6161-004	1.7.171.408	- 10
· / 1/ ·/ / /	1171, M36F+03 M1 - 6, 197F+03 -, M76, 307F103	$\frac{1}{1} + \frac{1}{1} + \frac{1}$	-lface	1.1446+04	-1.0201+04	1.0641104	-70.4P6
· / 11/ ·/ / +1 /711	1171.7366+03 P1 - 6.9198+03 - P76.3076+03 P77 - 7.7166+03	XII     7.0000000       YI     -4.316000       XI     -1.00000       XI     -4.747100	-1/468	1.144[+04	-1.0201+U4 Ym	1.0441404	-70.4P6
· / · 1/ ·/ / · · /7	1171. P36F+03 P1 - 6. 19F+03 P76. 307F+03 P77 - 7. 216F+03	7+ ACE ST4.31AF+03 ST - 1.0627+03 ST4.2471+03	-11466	1.1446+04	-1.0201+U4 VM	1.0441404	-70.4P6
· / · · / · · · · · · · · · · · · · · ·	1171. #366+03 PT - 6. 4196+03 - PT6. 3076+03 PT - 7. 2166+03 () - 2. 474(+02)	7+ ACE 574.31AF+03 57 - 1.0627+03 5374.7471+03 7+ ACE 574.3471+03	-21ACE	1.1.46.04	-1.0201+U4 YM	1.0441404	-70.XPG
· / · · · · · · · · · · · · · · · · · ·	$\begin{array}{rrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrr$	(1)     (1) <td>-2145</td> <td>1.1447+04</td> <td>-1.0201+04 YM -1.3641+34</td> <td>1.0441.04</td> <td>- 70. KP6</td>	-2145	1.1447+04	-1.0201+04 YM -1.3641+34	1.0441.04	- 70. KP6
· / · !/ · / · / · · · / · · · · · · · · · · ·	$ \begin{array}{rcrcr} 117 & - & -1 & +3 & +6 & +6 \\ +1 & - & 6 & +1 & 9 & +6 & 9 & -1 \\ +7 & - & -8 & -3 & 0 & 7 & +6 & 9 & -1 \\ +7 & - & -8 & -3 & 0 & 7 & +6 & 1 \\ +7 & - & 7 & -8 & -4 & +6 & 1 \\ +7 & - & -7 & -8 & -4 & +6 & 12 \\ \end{array} $	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	-111CE	].]44[+U4 ].]961+U4	-1.0201 +U4 YM -1.1441 +74	1.0441+04 1.47741+04 1.47741+04	- 70, 4P6 - 57, 684
· / · · / · · · · · · · · · · · · · · ·	$\begin{array}{rrrrr} 11Y &=& -1 \cdot F_3 \wedge F_1 \wedge 0_3 \\ F_1 &=& 6 \cdot f_1 (F_1 \wedge 0_3) \\ F_2 &=& -A \cdot 3 \circ 3 \circ F_1 \wedge 0_3 \\ F_1 &=& -A \cdot 3 \circ 3 \circ F_1 \wedge 0_3 \\ F_1 &=& -A \cdot 3 \circ 5 \circ F_1 \wedge 0_3 \\ F_2 &=& -A \cdot 3 \circ F_1 \wedge 0_3 \\ F_1 &=& -A \cdot 3 \circ F_1 \wedge 0_3 \\ F_2 &=& -A \cdot 3 \circ F_1 \wedge 0_3 \\ F_3 &=& -A \cdot 3 \circ F_1 \wedge 0_3 \\ F_4 &=& -A \circ 5 \circ F_1 \wedge 0_3 \\ F_4 &=& -A \circ 5 \circ F_1 \wedge 0_3 \\ F_4 &=& -A \circ F_1 \wedge 0_3 \\ F_5 &=& -A \circ F_1 \wedge 0_3 \\ F_6 &=& -A \circ F_1 \wedge 0$	7FACE ST4.316F+03 ST - 1.0627+03 ST - 1.0627+03 ST4.2471+03 VFACE SF6.3454+03 ST - 3.764(+03) ST - 1.1267+04 (FFC SF - 6.7291+03)	-1++CF	1.144[+U4 1.1966+U4	-1.0201 +UK YM -1.3451 +JK YM	1.0A41+04 1.0A41+04 1.479L+04	- 70, \P 6 - 57, 64 \
· / 11/ · / · / · · · / · · · · · · · · · · ·	$\begin{array}{rrrrrr} 117 & - & -1 & -1 & -1 & -1 & -1 & -1 &$	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	-2140	1.144F+04 1.106E+04	-1.0201 +04 -1.1641 +94 -1.261 +94	1.0441+04 1.479L+04 1.479L+04 2.1470L+04 2.1470+04	-70.4P6
/ 11/ / / 4 // / 4 // / 4 // / 4 // 11/ // 4 // 4	117     -1. M3AF+03       P1     6. 419F+03       P7     -A. 307F+03       P77     -2. 216F+03       17     -2. 474[+07]       17     -7. 474[+07]       17     -7. 494[+07]       17     -7. 494[+07]       17     -7. 494[+07]       17     -7. 494[+07]       17     -7. 494[+07]       17     -7. 494[+07]	7FACE ST4, 336F+03ST4, 316F+03ST4, 7471+03ST4, 7471+03ST1, 741+03ST1, 1761+04ST1, 183F+63ST2, 163F+63	-21 ACE	1.1497+04 1.1961+04 1.1321+04	-1.0201.04 YM -1.1441.74 YM -N.7571.01	1.0541.00 1.0541.00 1.4791.00 1.4791.00 1.4791.00 1.1051.00	-70.4P6 -57.044 11.175

•

•

-

.

Ļ





.

.

CUADRILATERAL STRESSES AND FORCES FOR OUTPUT VECTOR cont'd

•

•

Revision O

Y				
. /aujn / h7	FXX = 1*33AE+D3	5175.4028103	Ан	1.7851.01
. / 11/	MT + 8.793[103 -2F	ACE SI = -1.143E+04		2 2224 202 51 214
./ / +:	NT - 7.671163	SY = ~8.305E+03	-11166 -2-2666403 -1-1146404	
221	MIY4.314[+0] .	- 514- 7.1418+03		1.0001.004
12. 72	JT5-1576+03 12F	ACE ST5.0701+03		
	FT + -1.019F+01		+1FACE - 2,315E+03 -1+256E+04	- 7.4376+03 -44.601
. /01140 / 70	FTT + -4.1861+C3	517+ -7.436F+03	VN.	1.3866+05
. /	PT + 1.559[+01 -75	ACE 52 + -2.2445103		
	PY + -1.4595+02	ST + -2.951[10]	-1FACE -2.612C+03 -5.376E+03	1.4#2[+0] -70.453
73	MIY3.4181+02	SIT9.347E+02	VA VA	4.4325+03
12 24	FT	AFE SX = =1-6776=08		
	EV + 1.2656+01	54 - 3.2851+03	+15165 6.4335+03 -3.1446+03	4.2401+03 -35.026
10010 / 72	FIT + -5-300F+03	SIT= -4.300f+01	AN AN	8.45#[+0]
- / - 55/	PT + D/F	ACE SI + -1.5771-08		
	ry = 0.	ST + 3, 2031+03	-IFACE 6.4336+03 -3.1481+03	4.7401+01 -55.026
7	FTT + 0.	SIT+ -4.30L(+0]	AA A	8.4581+03
	••••	•		

2-49

•

. ~

2.5.1.4 Lid Lifting Eye

> $W_L = 2115 \text{ lbs.}$   $P = 3 W_L$  P = (3) (2115) lbs.= 6345 lbs.

1. Tension in the lifting eye.



2. Welding Around Lifting Eye

$$A_{eft} = \pi Dt_{eff} = \pi (1.0) \left(\frac{3}{8}\right) \left(\frac{\sqrt{2}}{2}\right) = 0.833 \text{ in}^2$$
  
$$\sigma_s = \frac{P}{2A_{eff}} = \frac{6345}{2(.833)} = 3810 \text{ psi}$$

Safety Factor = <u>20200</u> = 5.30 3810





2-50

#### 2.5.2 <u>Tie-Down Devices</u>

The tie-down system for transporting the package is designed to load conditions defined in 10 CRF 71, Paragraph 71-45 (b)(1). This load condition is defined as follows ". . . The system shall be capable of withstanding, without generating stress in any material of the package in excess of its yield strength, a static force applied to the center of gravity of the package having vertical component of two times the weight of the package and its contents, a horizontal component along the direction in which the vehicle travels of 10 times the weight of the package with its contents and a horizontal component in the transverse direction of 5 times the weight of the package with its contents."

The 13C II Cask has been provided with a tie-down frame which is integral with the top mounted overpack assembly. The overpacks surround the cask and are attached together with six (6) 1" diameter ratchet binders. These ratchet binders primarily carry inertial loads only since both end impact loads and tie-down loads are reacted by direct compression of the overpack and the integral tie-down frame upon the cask body.

The tie-down frame is designed for the loading conditions previously described. For purposes of this analysis, the package is assumed to weigh 27,000 lbs. Since a minimum factor of safety of 1.16 is calculated, the tie-down system is considered to satisfy the requirements of 10 CFR 71 for package weights up to 31,300 lbs. The analysis is subdivided into individual sections dealing with: loads, lug stresses, stresses included in the cask and tie-down frame stresses.

#### 2.5.2.1 <u>Tie-Down Forces</u>

Stress in the frame is determined for the forces transmitted by the tiedown cables, in resisting the "G" loading, to the frame. Since the cask will be blocked at the base on the





transporter to prevent sliding, the cable resisting forces will be those resisting tipping.

The individual cable loads are determined from principals of static equilibrium as outlined below. The direction cosines for the cables are found as: (Refer to the Tie-Down Geometry Figure 2.5.2-1 on page 2-55 for notation.)

 $\frac{B_{x}}{2} = \left[a - \frac{d_{c}}{2}\sin \alpha\right]/i$   $\frac{B_{y}}{2} = \left[\frac{w}{2} - \frac{d_{c}}{2}\cos \alpha\right]/i$   $B_{z} = h/i$ 

Where:

 $i = \left[ (a - \frac{d_c}{2} \sin \alpha)^2 + (\frac{w}{2} - \frac{d_c}{2} \cos \alpha)^2 + h^2 \right]^{\frac{1}{2}}$  $a = \tan^{-1} (2a/w)$ 

The log longitudinal load is resisted by two cables, in tension. The individual cable force is obtained by taking moments about point "O":

$$-10Wc + 2F_{cx}h + F_{c2}(d_{9}+d_{c}) = 0$$

But:  $F_{cx} = {}^{\beta}x \cdot F_{c_1}$  $F_{cz} = {}^{\beta}z \cdot F_{c_1}$ 

Thus, the cable force due to a 10 g longitudinal load is:

$$F_{c_1} = \frac{10 \text{ wc}}{2 \text{ s}_x \text{h} + \hat{\varepsilon}_z (d_0 + d_c)}$$



...

•

0

The 5g lateral load is resisted by two cables, in tension. The individual cable force is found from the equilibrium expression:

But:

$$F_{cy} = \beta_{y} \cdot F_{c_{2}}^{\dagger}$$

$$F_{cz} = \beta_{z} \cdot F_{c_{2}}^{\dagger}$$

Thus, the cable force due to a 5 g lateral load is:

 $F_{c_2} = \frac{5Wc}{2\beta_y h + \beta_z (d_0 + d_c)}$ 





Revision O



The 2g vertical load is resisted by all four cables. in tension. The individual cable force is found from the equilibrium expression:

 $-2H + 4F_{cz} = 0$ But:  $F_{cz} = B_z \cdot f_{c3}$ 

Thus, the cable force due to a 2 g vertical load is:

$$F_{c3} = 2W/4B_z$$

Assuming the three loadings occur simultaneously, the most severely loaded cable experiences a load of:

$$F_{c} = H \left[ \frac{10c}{2 B_{x}h + B_{z}(d_{0} + d_{c})} + \frac{5c}{2B_{y}h + B_{z}(d_{0} + d_{c})} \right] + \frac{2}{4B_{z}}$$

This expression is evaluation with the following 1-13C II properties and geometry:

H = 27000  1bs.	d = 40.00 inches
a = 76 inches	h = 99-5/8 - 15 = 84.625 in.
c = 58.75/2 + 15-7/8 = 50.25 in.	w = 96 inches

The direction cosines of the cable are:

 $B_{\chi} = .53829$  $B_{\chi} = .34004$  $B_{\chi} = .77105$ 

Cable length, 1 - 109.75 inches





The cable loads are:

Load Case	<u>Magnitude (155)</u>
10 g longitudinal	80650
5 g lateral	• 50378
2 g vertical	17509_
Combined	148536

.

For stress evaluation of components, the corresponding load set based upon simultaneous application of the lOg, 5g, and 2g load cases to the circular tie-down frame imposes consistent cable forces (out of the plane of the sketch) as follows:





. .

These forces may be resolved into components parallel and perpendicular to the plane of the frame with the following expressions:

$$F_{p} = (\hat{s}_{x}^{2} + \hat{s}_{y}^{2})^{\frac{1}{2}} \cdot F_{c} \qquad (Parallel)$$

$$F_{n} = \hat{s}_{z} \cdot F_{c} \qquad (Normal)$$

e

The results are:





•

# 2.5.2.2 <u>Stresses in Tie-Down Lug</u>

The tie-down lug is oriented precisely in line with the cable force vectors evaluated in the preceding section. As a consequence, the tie-down lug experiences inplane loads only; no bending loads are induced in this element.



(a) <u>Tearout Stress</u>

Using a conventional 40° tearout relation, the capability of the lug at yield stresses is estimated ap:

 $P_{A} = 2 F_{sy} t (E_{m} - d/2 \cos 40^{\circ}) = 245060 lbs.$ 





Where:  $F_{sy} = \frac{100000}{\sqrt{3}} = 57735 \text{ psi; } \text{ASTH-A514, Gr. 3}$  t = 7/8 in.  $E_m = 3 \text{ in.}$ d = 1.5 in.

Thus, the factor of safety for lug tearout is

F.S. = <u>245060</u> = 1.65 148536

(b) <u>Net Section Tension Stress</u>

The net section tension yield load capacity is:

 $P_A = F_{ty}^A$ = (100000) (4-1.5) (.875) = 218750 15s.

Thus, the factor of safety for lug tension is:

F.S. - <u>218750</u> - 1.47 148536

(c) <u>Welds to Frame</u>

The tensile capacity of weld (1) is:

 $P_{u1} = (100000)$  (4) (.875) = 350000 lbs.

The shear capacity of weld (2) is:

 $P_{y2} = (100000) (1/\sqrt{-33}) (.707) (.875) = 35,716$  lbs.



•

Collectively, the weld capacity of the lug attachment is 385,722 lbs. Thus, the factor of safety is:

> F.S. = 385.722 = 2.60148,536

Therefore, should tie-down forces exceed the criteria specified in 10 CFR 71, the lug will fail at 240,625 lbs. (ultimate) in the net tensile section. This failure will not impair the ability of the cask or its contents to meet other provisions of 10 CFR 71.

## 2.5.2.3 <u>Tie-Down Frame Stresses</u>

The reaction forces between the frame and cask are found from the forces of Section 2.5.2.1. The horizontal planar forces applied to the circular tie-down plate may be viewed as the superposition of two force sets as shown below.

The vectors external to the circles are applied by the tie-down cables; whereas the vectors within the circles are applied by cask reactions.









. \_\_\_\_\_

These planar horizontal reaction forces are applied to the cask as a lateral bearing load. The corresponding force acts on the tie-down frame as a lateral load applied to the 12 inch long cask restraint flanges located at each tie-down lug. Detail geometry and applied loads are shown in Figure 2.5.2-2.

Bending stresses are induced in the cask restraint flange by this cask to frame load. The moment area for this load is conservatively estimated assuming the center of pressure at mid depth of the cask body top plate. The previous sketch indicates two critical sections where flexural stresses will be checked.





FIGURE 2.5.2-2

TIEDOWN LUG DETAILS





# Section 1

The section properties are:



+ 
$$(10.5)(.65625-.4375)^2 = 2.9030$$
 in<sup>4</sup>

The applied moment involves both the cask reaction force plus the cable force. Moment arms are scaled from the preceding sketch.

$$M_1 = (83436) (1.25) - (17509) (0.35) = 98167 in-1b$$
  
 $f_{b_1} = Mc/I = (98167) (1.09375) = 36986 p.s.i.2.9030$ 

Section 2

The moment at Section 2 involves the cask reaction force plus the cable force. Applied moment is:

 $M_2 = (83436) (1.688) - (17509) (0.52) = 131, 735 in-1b.$ 







The section is a single thickness of 7/8" plate with length b found as:

Thus the bending stress:

•

$$f_{b_2} = 6M/bt^2 = (6)(131735) = 86319. psi (11.96)(.875)^2$$

The minimum factor of safety in the tie-down frame is therefore:

$$F.S. = \frac{100000}{86319} = 1.16$$

2.5.2.4 <u>Cask Tie-Down Stresses</u>

The contact bearing stress due to lateral loads, calculated in paragraph 2.5.2.3, is:

$$f_{b_r} = \underline{83436} = 2,781 \text{ psi}$$
  
(12) (2.5)

•







.



..

13500 1bs. 75685 1bs. 114529 1bs. 52344 1bs.

These loads exert a total downward force on the cask external shell of 256,058 lbs.

The external shell is constructed of 1/2" thick 304 stainless steel. Resultant compressive stress is:

> $f_{t} = P/A = \underline{256058}_{\Pi} = 4347 \text{ psi}_{\Pi}$ (37.5)(.5)

Since the lead provides adequate restraint to the outer shell, crippling need not be considered. The resultant cask factor of safety is:

 $F.S. = \frac{30000}{4347} = \frac{6.90}{1000}$ 



----



Revision O

# 2.5 Normal Conditions of Transport

The Model 1-13C II package has been designed and constructed, and the contents are so limited (as described in Section 1.2.3) that the performance requirements specified in 10 CFR 71 will be met when the package is subjected to the normal conditions of transport specified in 10 CFR 71. The ability of the Model 1-13C II Package to satisfactorily withstand the normal conditions of transport has been assessed as described below:

# 2.5.1 <u>Heat</u>

A detailed thermal analysis for normal conditions of transport can be found in Section 3.4. Internal heat loads up to 800 watts were considered.

The maximum cavity temperature was found to be 214.4<sup>o</sup>F. External package temperature was less than 156.3<sup>o</sup>F, assuming 100<sup>o</sup>F ambient air and full solar insolation. These temperatures will have no detrimental effects on the package.

# 2.6.6.1 <u>Summary of Pressures and Temperatures</u>

From Sections 3.4.2 and 3.4.4, it was found that the maximum temperatures and pressure are:

Pressure:

#### Pmax = 19 psig

Temperature:

Fire Shield	182.1 <sup>0</sup> F
Lift Lug	209.1 <sup>0</sup> F
Outer Shell	212.6 <sup>0</sup> F
Outer Ends	214.3 <sup>0</sup> F
Containment Cavity	214.4 <sup>0</sup> F



2-66

Revision O

## 2.6.1.2 Differential Thermal Expansion

From the summary temperatures shown in Section 2.6.1.1 and Section 3.1 it can be seen that the temperature variations between the external shell and the inner containment vessel are small, i.e.,  $212.6^{\circ}F$  vs.  $214.4^{\circ}F$ , respectively.

Since the lead bonding is not present, fabrication stresses are minimized because of the short term creep properties of lead. Therefore, a stress free temperature of 70°F was assumed.

The analysis of package stresses due to differential thermal expansion has been performed using a pair of axisymmetric finite element models, Figures 2.6.1-1 and 2.6.1-2, representing the cask and lid, respectively. These models are used for cask body analytic stress predictions throughout Section 2.6 and 2.7 of this report. The E3SAP program, described in Appendix 2.10.2.4, was used for all these analyses.

The referenced figures completed define model geometry. All material properties of the model are taken directly from Table 2.3-1 for stainless steel, Type 304, and lead. To represent the unbonded lead, the lead shear modulus has been reduced by a factor 10. This assures that only secondary shear forces are transmitted between steel shells and lead; yet permits proper treatment of direct stresses induced in steel shells due to differential thermal expansion effects.







FIGURE 2.5.1-1 AXISYMMETRIC FINITE ELEMENT MODEL OF CASK 1-13 C II







のないのないというないのである



Comprehensive stress summaries for these two models are provided in Appendix 2.10.3. These summaries directly provide membrane (center) and surface stresses in terms of stress intensities as set forth in NRC Regulatory Guide 7.6. For differential expansion loads, stress intensity results are reported in Case 5\* (= Run 2, Case 1).

In the mid body region of the cask, stresses are modest, with little variation from inner to outer surfaces. Values at a typical section through elements 23/44, where inner shell stresses maximize, see Figure 2.6.1-1, are as follows:

Location		STRESSES (psi)				
	Element	Arial	Hoop	Radial	-1	
	Number	(° <sub>2</sub> )	(° <sub>9</sub> )	(° <sub>r</sub> )	(5.1.)	
Inner Shell	23	+11520	+ 4406.	-210	11830.	
- I.D.		+11890	+ 4578	-294	12184.	
- 0.D.		+11350	+ 4225	-121	11471.	
<u>Outer Shell</u>	44	12270	+18380	-265	18645.	
- I.D.		12160	18590	-508	19098.	
- O.D.		12390	18170	- 16	18186.	

\*The seven reported cases were performed in two runs; Run 1 contains 4 cases; Run 2 contains 3 cases. For clarity and brevity, they are identified and discussed as 7 sequential cases in this report.





#### Revision J

		STRESS INTENSITY (psi)		
LOCATION	ELEMENT	CENTER	SURFACE	
Top - Inner Shell	30	13671	16342	
- Outer Shell	46	19503	19737	
Base - Inner Shell	15	8005	9366	
- Outer Shell	31	18464	19942	

The axial variation of temperature induced stress intensity is illustrated by the following comparisons at top and bottom of cask.

Stresses elsewhere in the cask body are less than these values excepting for localized stresses at the outer shell to base juncture, see elements 5 and 8. The membrane stresses here are:

ELEMENT	MEMERANE S.I.		
	(251)		
5 (Inner Shell)	19565.		
8 (Outer Shell)	28261.		

The stresses observed at elements 5 and 8 are highly localized; adjoining element stresses show extremely rapid decay of the obvious classical damped shell bending behavior. These high stresses, slightly in excess of yield, pose no threat to cask safety or integrity since they are self limiting secondary thermal stresses which disappear upon initial yielding of the outer shell. The only consequence of these stresses will be a slight outward ripple of cask outer surface subsequent to initial loading.







Stresses in the lid, Figure 2.6.1-2, are approximately of the same magnitude as in the cask body. The most highly stressed location occurs at element 8. The state of stress of element 8 is as follows:

Location	Axial	Ноор	Radial	Intensity	
	( <sub>CZ</sub> )	( <sub>су</sub> )	(g <sub>r</sub> )	(S.I.)	
Center (Membrane)	10550	22450	+3174	19819	
O.D.	32440	29580	6124	26331	
I.D.	-4735	18140	4735	23405	



As in the cask body, the dominant stress contributor is a bending stress arising from the connection of the conical shell sector to the flat inner end disk.

# 2.6.1.3 <u>Stress Calculations</u>

Combination of thermal stresses discussed in Section 2.6.1.2 with internal pressure (19 psig) stresses slightly increases the total state of stress.

Stress intensities for pressure and temperature phenomena at those locations discussed in Section 2.6.1.2 are obtained directly from case 7 of Appendix 2.10.2, as follows:



Revision O



LOCATION	ELEMENT NU74BER	STRESS INTENSITY S.I. (psi)
CASK BODY		
Inner Shell - Center	28	11864.
- I.D.	•	12224.
- O.D.		11510.
Outer Shell - Center	44	18788.
- I.D.		19252.
- O.D.		18326.
Top - Inner Shell, Center	30	13696.
Surface	30	16331.
- Outer Shell, Center	46	19645.
Surface	46	19871.
Base - Inner Shell, Center	15	8255.
Surface	15	9592.
- Outer Shell, Center	31	18480
Surface	31	19957
Juncture, Outer Side	8	28265
Outer BAse	5	19590
LID		
Center	8	19664
0.D.	8	26169
I.D.	8	23809
		1



.



.



Maximum stress intensities and factors of safety are as follows:

1.	Conta	Containment Vessel (Inner Shell)	
	(NRC	Reg. Guide 7.6, Para. C.2)	
	a.	S <sub>m</sub> = 20 ksi @ 214.4 <sup>0</sup> F (Table 2.3-1)	
		Allowable Membrane Stress = $S_m = 20$ ksi Allowable bending + membrane = 1.5 $S_m = 30$ ksi	
	b.	Maximum Membrane Stress @ El. 44	
		f <sub>m</sub> = 18.788 ksi F.S. = 20/18.788 = <u>+1.06</u>	
	c.	Maximum Membrane + Bending @ E1. 44	
		f <sub>b</sub> = 19.252 ksi F.S. = 30/19.252 = <u>+1.56</u>	
2.	<u>C101</u>	<u>ure (Lid).</u> E1. 8	
	<b>a</b> .	Maximum Membrane Stress	
		f <sub>m</sub> = 19.664 k±i F.S. = 20/19.664 = <u>+1.02</u>	
	۵.	Maximum Membrane + Bending	
		f. = 26.169 ksi	







3. Duter Shell

المعادة المتعسم والداري المراج الم

a. Allowable Stress:

 $S_{y} = 24.64$  ksi  $\ell = 214.4^{\circ}F$  (Table 2.3-1)

5. Maximum Stress at El. 31, excepting juncture:

 $f_5 = 19.957$  ksi F.S. = 24.64/19.957 =  $\pm 1.23$ 

c. Maximum Stress at Base Juncture, El. 8

 $f_{\rm m} = 28.265$  ksi F.S. = 24.64/28.265 = 0.87

The slight (13%) overstress at Element 8 is highly localized as shown by the factors of safety for adjoining elements.

Element	2 Location	f	F.S.
Number_	(inches)	<u>(ksi)</u>	
7	. 1.75	7.610	3.24
8	5.00	28.265	0.87
9	7.38	19.843	1.24

This highly localized self-limiting thermal stress yielding will produce a slightly outward ripple on the exterior cask shell approximately 5 inches above the base, well beyond the heat affected zone of shell to base plate weld.

This initial yielding cannot lead to progressive collapse or shakedown rupture of the shell since subsequent load and unload cycles will always induce stresses within the elastic regime.





This fact is assured by the "Baushinger Effect"\* so long as the total stress range is less than twice yield stress. The ASME Boiler and Pressure Vessels Code. Section III, explicitly recognizes this fact by the allowable factor of 3 S<sub>m</sub> on combined primary and secondary stresses (S<sub>m</sub> - 2/3 S<sub>y</sub>). In this instance, the zero to peak stress range is -1.147 S<sub>y</sub>. Based on the "Bauschinger Effect" the factor of safety is recomputed as:

$$F.S. = 2 S_y/1.147S_y = +1.74$$

This condition does not impair the effectiveness of the packaging to satisfy the requirements for Normal Conditions of Transport as set forth in 10 CFR 71, Part 71.71.

The pressure (19 psig) associated with normal transport produces insignificant loads upon the 12 1-1/4"-7 UNC closure bolts (ASTH SA 354, Gr. B). The loads are assessed as follows:

F.B. = 
$$(\pi/(4) (12)) (33^2) (19) = 1354$$
 lbs.

The effective section area of the bolts is  $0.968 \text{ in}^2$ , therefore, the stress is:

The allowable tensile stress for the bolting material, Table 2.3-1, is 125 ksi. Therefore, the factor safety is:

Therefore, the Model 1-13C II packaging can safely satisfy the requirements of thermal and pressure loading of the Normal Conditions of Transport.

\* Timoshenko, S. P., <u>Strength of Materials</u>, Part II, "Advanced Theory and Problems" Third Edition, D. Van Nostrand Company, Inc., 1956, pp. 413.



# 2.6.2 <u>Cold</u>

The materials of construction for the packaging, including the stainless steel, carbon steel, overpack and the seals themselves are not significantly affected by an ambient temperature of  $-40^{\circ}$ F. Brittle fracture has been discussed in Section 2.1.2.3.1 and is not a physically plausable mechanism for the materials of construction.

There are two cold conditions corresponding to the  $-40^{\circ}F$  ambient air event minimum decay heat and maximum decay heat and maximum decay heat. In the first instance, the differential thermal expansion coefficients between steel and lead cause the lead and steel to separate (gap) at temperatures below the  $70^{\circ}F$  stress-free temperature. Both lead and steel then experience an unconstrained shrinkage; hence, approach a stress free condition.

In the second instance with maximum decay heat, the predicted temperatures as summarized in Table 3.4.3-1 are as follows:

LOCATION	NODE_NO.	TEMPERATURE_(°E)	
Fire Shield	9	37.4	
Lift Lug	2	74.8	
Side Outer Shell	10	79.6	
Top Outer Shell	3	81.4	
Cavity	22	81.5	

Except for the fire shield, all temperatures are very slightly above the  $70^{\circ}$ F stress-free/temperatures; the maximum temperature differential above this stress-free temperature reference occurs at the cavity and is  $11.5^{\circ}$ F. The temperature gradient between the cavity and the exposed side outer shell is:

$$\Delta T = T_{22} - T_{10} = 1.9^{\circ}F$$


This is essentially identical to the maximum gradient reported for "hot" conditions in Table 3.4.2-1.

A conservative estimate of thermal expansion stresses in the steel shell induced by differential coefficients of thermal expansion between steel and lead is:

$$f = E_s \Delta T \quad (\alpha_{Pb} = \alpha_s)$$
  
= (28.3x10<sup>6</sup>)(11.5) (16.4-9.11)x10<sup>-6</sup> = 2373 psi

The stresses induced by the gradient through the wall can be approximated by assuming a thick cylinder of steel. The stress is given in Roark, 5th Edition, Case 16, Section 15.6, as:

 $F = (\Delta T = 2g^{2} + c^{2} - b^{2}) \cdot \ln (c/b) = -393 \text{ psi}$ where: b = inner radius = 13.25 in.c = outer radius = 19.25 in.

g = b, for outer surface stress

= c, for inner surface stress

Assuming these two stresses are additive, the factor of safety is:

F.S. = 30,000/2373 + 393 = 10.8







## 2.6.3 <u>Reduced External Pressure</u>

10 CFR 71.71 (c)(3) requires that the package should be able to withstand a reduced external pressure of 3.5 psia. Conversely, the package should be able to withstand a 14.7 - 3.5 = 11.2 psi internal pressure. Assume the inner shell is supported by the lead in resisting the internal pressure.

For the lead and steel shells in contact, the yield of the steel plate will not be exceeded as entire pressure can be taken by the lead without yield.

#### 2.6.4 Increased External Pressure

The requirement for external pressure is that the cask must be able to withstand an external pressure of 25 psig without the loss of contents. Assume the outer shell is supported by the lead in resisting the external pressure.

2-79







For the lead and steel shells in contact, the yield of the steel plate will not be exceeded as entire pressure can be taken by the lead without yield.

$${}^{P}c = \frac{2t}{D} \left( \frac{\sigma_{V}}{1 + \frac{\sigma_{V}}{E} \left( \frac{D}{t} \right)^{2}} \right)$$

E = Modulus of Elasticity = 28 x 10<sup>6</sup> psi  $\sigma_y$  = Yield Strength = 30,000 psi D = 37.5 in. t = 0.50 in. Pc =  $\frac{(2)(.50)}{(37.5)} \left( \frac{30,000}{1 + \frac{30,000}{.5}(\frac{37.5}{.5})^2} \right)$ 

= 114 psi

Safety Factor: SF =  $\frac{114}{25}$  = 4.6

\*Roark, R. J., Formulas for Stress and Strain, Fourth Edition. McGraw-Hill Book Co., New York, 1965 Chapter 12, page 298, Case 1.





## 2.5.5 <u>Vibration</u>

Section 1.1 observes that the Model 1-13C II is an improved lineal descendent of a proven package with well over ten years of operational use in a transcort environment. This experience conclusively demonstrates that vibrations normally incident to transport will have no effect upon the Model 1-13C II package.

#### 2.5.5 Hater Spray

Since the package exterior is constructed of steel, this test is not required.

#### 2.5.7 Free Drop

The package weight of 27000 pounds indicates that the Model 1-13C II package must survive a two foot free fall without substantially reducing its effectiveness in reacting subsequent accident conditions. The package has been designed and successfully tested to withstand 30 foot drops, see Section 2.7.1 and 2.11. Energies generated during the 30 foot drop are 9.72x10<sup>5</sup> in-1bs. Energies associated with the two foot drop are:

KE = (27000 lbs) (24 in) = 648,000 in-lbs.

For impact energies of this magnitude, the computer generated data shown in Section 2.7.1 and described in Appendix, Section 2.10.1.1 provides maximum impact accelerations. These loads are summarized below:

Condition	<u>2 Foot Drop (g's)</u>
End	62.2 (EYDROP)
Corner	14,2 (CYDROP)
Side	28.3 (SYDROP)



These energy balance load predictions were verified by performing package dynamic analyses for a full range of impact orientations from near vertical to near horizontal, as described in Appendix Section 2.10.2.2. Load results from the dynamic analyses compare with the energy balance results, as follows:

	Energy	Dynamic
Crientation	Ealance	<u>Analysis</u>
End	52.2g	25.7g (02.5 <sup>0</sup> )
Corner (29.7 <sup>0</sup>	14.2g	14.2g
wrt. Vert.	)	
Side	28.3g	N/A

The corner orientation values agree precisely. For an end impact orientation, the difference appears to exceed a factor of two. The plotted dynamic response results, Figures 2.6.7-1 and 2.6.7-2, show why. As the orientation angle approaches 0°, the dynamic response force predictions increase very steeply while the predicted deformation of the energy absorbing overpack decreases very rapidly. For assessment of stresses, the most conservative of the alternative load predictions are employed.

## 2.6.7.1 End\_Impact

End impacts produce the largest forces on the Hodel 1-13C II Package. At other orientations, the forces rapidly decrease. The end impact acceleration of 62.2g induces stresses in the cask and lid as reported in Case 3. Appendix Section 2.10.3. Case 3 stresses are developed for 30 foot drop accident conditions with an acceleration of 100.184g's. Case 3 stresses must therefore be multiplied by the ratio of (62.2/100.184)=.6209 to correspond with the 2 foct drop provisions for normal conditions of transport. Maximum stresses are as follows:



Revision O

FIGURE 2.6.7-1

IMPACT FORCES (LBS) -2 FT DROP







.

## FIGURE 2.6.7-2

OVERPACK DEFLECTION AND RESIDUAL CLEARANCE -2 FT DRCP







		STRESS INTE	NSITY (psi)
LOCATION	El. Nr.	SURFACE	CENTER
		l I	
C <u>*:k_Body</u>			
o Inner Shell	14	7150.	4326.
o Outer Shell*	8	12304.	9574.
riq	16	18082	10365.
	17	22493	4791.



\*Stresses at element 7 are neglected due to the unrealistic concentrated support of the cask at the lower end of this element.

For the containment elements of the package, the minimum factors of safety in membrane and combined stress states of stress are:

o Membrane (Cask Element 14)

$$F.S. = 20 = \pm 4.62$$
  
4.326

o Combined Membrane + Bending (Cask Element 14)

$$F.S. = 30 = \pm 4.20$$
  
7.150



.

S.I.

22494



For the outer shell and lid, the minimum factor of safety is at lid element 17:

$$F.S. = 24.64 = \pm 1.10$$
  
22.493

$$(Sy = 24.64 \text{ ksi } 214.4^{\circ}\text{F} \text{ per Table } 2.3-1)$$

When these maximum stresses are combined with normal temperature and pressure stresses, Case 7, the minimum factors of safety become:

o Cask Element 14 - Membrane:

F. S. = 
$$20$$
 =  $\pm 1.92$   
(4.326+6.095)

o Cask Element 14 - Combined:

F. S. = 
$$30$$
 =  $\pm 2.20$   
(7.150+6.507)

oad	σ	σ	τ	σ
	r	z	rz	•
	<b>^</b>	-		

o Lid Element 17 (Top Surface)

	$\Gamma.S. =$	$54 = \pm 1.22$	2		
Combined	.+327	-19321	-2162	-4213	20118
Press + Temp	-849	+ 1510	166	5566	6427

Is is concluded that the Model 1-13C II Package, including overpack, safely withstand the end drop requirements for normal transport conditions.

1176 -20831 -2328 -9779



prop

## 2.6.7.2 <u>Side Impact</u>

Predicted side impact acceleration under normal conditions is 28.3g's, per Section 2.6.7. This lateral acceleration induces bending stresses in the package as it lands upon the overpack protected ends. The magnitude of the bending moment at mid length is:

$$M = \frac{1}{8}; \qquad W = (27000)(28.3) \text{ lbs.}$$

$$1 = 68 \text{ inches}$$

$$= 6.495 \times 10^6 \text{ in-lb.}$$

The moment of inertia of the steel segments of the cask are:

$$I = \frac{\pi}{4} \left( \frac{R_0^4 - R_1^4}{R_0^2 - R_1^4} \right) = \frac{\pi}{4} \left[ (13.74^4 - 13.25^4) + (14.25^4 - 18.75^4) \right]$$
  
= 14,642 in<sup>4</sup>

The bending stresses in the outer shell and inner shell are:

$$f_{bi} = (6.495 \times 10^6)(13.5) = 5988 \text{ psi}$$

These bending stresses must be added to the normal pressure stresses summarized in Case 7, Appendix Section 2.10.2. Elements 23 and 29 correspond to the mid-length where the above bending stresses are calculated.





				STRESSES	(psi)
LOCATION	a r	o z	T FZ	σ θ	S.I.
223 er_Shell_(El.19)					
- Bending	٥	± 8428	0	0	-
- Case 7 (Pres&Ther)	-285	13530	-11	19140	19425
- Combined	-	+21958	-11	19140	22243
Inner_Shell_E1.23)					
- Bending	0	± 5988	0	o	-
- Case 7 (Pres&Ther)	53	11300	-11	4704	11247
- Combined	53	17288	-11	4704	17235

## Stresses are summarized as follows:

The minimum factors of safety under side impact and pressure and temperature conditions are:

o The Inner Shell (Containment Vessel):

F.S. = 
$$20$$
 =  $\pm 1.16$   
17.235

o The Outer Shell:

.

$$F.S. = 24.64 = \pm 1.11$$
  
22.243

It is concluded that the Model 1-13C II Package, including overpack, can safely withstand the side drop requirements for normal transport conditions.





Revision O

## 2.5.7.3 Corner Impact

Corner Impact accelerations are a rather modest 14.2g's. As outlined in Appendix 2.10.1.2, moments are introduced into the cask body by these oblique impacts. Figure 2.6.7-3 depicts the moments produced in the cask body as a function of orientation angle. The maximum moment is found as:

 $M = 2.1594 \times 10^{6}$  in-1b 9 51.3° wrt horizontal

This is approximately a factor of three less than the moments examined for side impact, Section 2.6.7.2, and consequently will result in stresses less than those examined in that section.

It is concluded that the Model 1-13C II Package, including overpack, safely withstand the corner and oblique force drop requirements for normal transport conditions.

••••••••••••••••••••••••••••••••••••••	External	Shell	Containment Vessel			
Drop Grientation	Stress	Factor of Safety	Stress	Factor of Safety		
End	20118	+1.22	10421	+1.92		
Corner	-	-	-	-		
Side	22243	+1.11	17235	+1.16		

## 2.6.7.4 <u>Summary of Results</u>

From the above it can be concluded that the package experience no inelastic deformation and all stresses are well below yield. Therefore, the Model 1-13C II Packaging can safely satisfy the loading due to the drop condition associated with Normal Conditions of Transport.





FIGURE 2.6.7-3

## CASK BENDING MOMENT (IN-LB) -2 FT DROP





2-90

### 2.6.8 Gorner From

This requirement is not applicable since the Model 1-13C II Packaging is fabricated of steel and weighs more than 110 lbs.

2.6.9 <u>Compression</u>

Not applicable since package weighs more than 10,000 lbs.

#### 2.6.10 Penetration

Impact energies resulting from a 13 pound rod dropping from a height of 4C inches will have no significant effect on the exterior of the cask. The overpack fully protects both ends of the package leaving only the central cask body which is manufactured from 1/2 inch thick steel and backed with over 5 inches of lead. No values, value covers or fragile protrusions exist.

#### .2.6.11 Conclusions

As the result of the above assessment, it is concluded that under normal conditions of transport the package complies with criteria in 10 CFR 71, as follows:

- There will be no release of radioactive material from the containment vessel;
- The effectiveness of the packaging will not be substantially reduced;
- 3) There will be no mixture of gases or vapors in the package which could, through any credible increase in pressure or an explosion, significantly reduce the effectiveness of the package.

#### 2.7 <u>Hypothetical Accident Conditions</u>

The Model 1-13C II package has been designed and its contents are so limited that the performance requirements specified in 10 GFR will be met if the package is subjected to the hypothetical accident conditions specified in 71.73 of 10 GFR 71.

To demonstrate the structural integrity of the package and its ability to withstand accident conditions, a comprehensive full scale drop test program has been conducted. These tests employed an existing 1-13C II package modified to bring the assembly into conformance with the configuration detailed in Section 1.3.1.1. Analytic predictions are used to augment test data where topics have not been directly measured in tests.

#### 2.7.1 Free Drop

Section 71.73 of 10 GFR 71 requires that the package survive a 30 foot drop onto a flat essentially unyielding surface. The test and analysis methods used to demonstrate this capability closely parellel the techniques used for past Type B packages. Analytic techniques are completely described in Appendix Section 2.10.1. Test methods, plans, data, results, conclusions and interpretations are consolidated in a special drop test Appendix, Section 2.11.

As described in Section 1.2, the Model 1-13C II package features circular energy absorbing overpacks surrounding each end of the cask body. These overpacks are designed to minimize damages to the cask body during 30 foot drops at any orientation upon unyielding surfaces. Test results, described in Section 2.11, demonstrate that these overpacks function as designed; the cask body experience no damage and incurs no measurable strains in excess of yield. This behavior under tested 30 foot drop conditions assures the complete effectiveness of the cask closure features essential for preservation of package containment integrity.



Overpacks loads and performance analyses preceded and followed the drop tests. Analysis prior to test were conducted to refine the geometry and properties of the overpacks and to select the most critical drop orientations for these tests. Analyses subsequent to test were conducted using actual physical properties of the test articles. These post-test analyses were used to derive essential parameters not directly measured by test. An important result of these tests was the validation of analytic overpack performance prediction methods described in Section 2.10.2.

Prior to conduct of the two drop tests described in Section 2.11, a full scale engineering prototype drop test was conducted in July, 1980, using a simulated cask and prototype overpack. This engineering development test, like the test described in Section 2.11, was completely successful and the simulated cask suffered no damage. Notably, the simulated cask was fabricated of materials with significantly less robust structural properties than the actual 1-13C II cask. This engineering development test resulted in slight modifications of crushable foam properties and overpack shell thicknesses to improve overpack performance. This development test data also resulted in slight modifications of predictive methods for oblique impacts, Section 2.10.2.2, to improve the accuracy of pitching moment calculations. That is, the location of the effective impact force was changed to allow variation as a function of crushed overpack footprint geometry.

Using the methods of Section 2.10.1.1, three drop conditions for the package have been "valuated, i.e., end, corner, and side. For each, relevant test results < : Malytic values are then correlated with these data and combined with appropriate analytic temperature and pressure data. These combined results are then compared with applicable criteria to demonstrate compliance of the Model 1-13C II package with requirements for hypothetical accident conditions.





### 2.7.1.1 Free Drop Impact, End Drop

The first of two drop tests, discussed in Section 2.11 was an end impact upon the package base. Of all the potential orientation angles, end drop produces one of the largest package deceleration forces.

These deceleration forces induce large edge bending forces on the edge supported and sealed lid. Test results indicated no degradation of containment integrity, and no measurable change to the closure geometry.

These deceleration forces together with edge prying forces potentially induce severe loadings in the closure bolts. Test results, including instrumented bolt force measurements, indicate surprisingly modest forces in closure bolts; at or below yield.

These deceleration forces also provide large forces which could induce slump of the unbonded lead shield. Gamma scan test results indicated no lead slump; in fact, there was no measurable change to the lead shield configuration of any fashion.

For a thirty foot end impact drop, measured dynamic deformation of the overpack amounted to 4.46 inches. Post-test analytic predictions were adjusted to agree precisely with this value, as outlined in Section 2.11.5.2. These analytic predictions employed the computer program EYDROP, described in Section 2.10.1, and the energy absorbing foam properties of Figure 2.3-1. EYDROP output results are shown in Table 2.7.1-1. Peak deceleration of 95.56g was calculated. A slightly higher deceleration of 100.18g was forecast by the oblique dynamic analyses. For conservatism, the higher value has been used for stress calculation purposes.









• •				 . •	• • • • • • • • • • • • • • • • • • • •	* • • • • • •	• •	
				•••				
10401 16401	•	CHS1 1-	130 0463510	K END DRCP	inalisis 16.	5 PCF F013		· · · ·
····	•• • • ••	· · •	FICKIST NO	1611	*	• • • • • • •	- ···-·	
			PICKIGE DI	4r=1(2	17.36 []H]			
			CVEPPICE D	ЕРТН -	15.00 (18)			
-			-CX08-HETCH	······································	36260. [111		•••••••••••••••••••••••••••••••••••••••	• • • • •
·- •	•	• •	•	• • •				
CEUSH		· · · · · 17.PL	.1	* * * * * *	EHEPGY ++++	• • •		
DCP1H · ···	534#1H ··	·· · · · · · · · · · · · · · · · · · ·	+0000					
(16)		(155)	(()	118-161	IIN-LEI	(31/41)		
• • •	· •· •	· ·	•		•	•••		
. 50	.033	1317501			•••••••••••••••••••••••••••••••••••••••	•••		•
1.((	.067	2251306.	44.7	9767226	324392.	• 634	-21 Baup: 5	637
	• 100		90.9	9761100:	2411409		مر	11 1 70 000 1/1
2.00	• 1 2 2	751127E.	93.0	\$774666.	36:6951.	. 176	/-	• 1,017,792,10
2.50	.157	2205914.	97.4	5787566.	4931924.	. : . ) ?	6	· 67.2 g
2.00	• 3()	2470724.	. 21.0	"YBOICCC.	"6119260.	* + Z E	/	ş
5.66	• 6 2 2	2:00156.	72.6	Y314560.	7464669,	.754		
	• * 0 / • · · • = - 7 c A		97.4	9525JGD.	6662799.	.(+1		to a set
5.07	. 331	7116673.	45.7.	5841200		1:010	20 3200 1.0	· · · · · · · · · · · · · · · · · · ·
5.10	.357	2767351.	102.5	98530CC.	11254230.	1,142		= 2,550,037 16
6.00.		2856146.	102.5	YBE4360.	12612314.	1.278		
5.10 ·	. 532	26.27745	112.5	1122000	14027323,	1.426	Ğ	* 15.56g.
7.(C	. 4 4, 7	1552521.	117.7	5935263.	17078469.	1.71	·	' /
7. 50				4922:00				
· • •.co	• 2 3 3 •	3796256.	140.7	,933566.	20500573.	2.671		
· · · · · · · · · · · · · · · · · · ·	- 1.7	4171675.	154.1	\$985500.	2257311L.	2.264		
9.10			172.3 .	···· \$963000.	2-776927.	2.487	• • • • •	•
10.00	. 1 67	x252647	196.7	5972560.	27269227.	2.733		
10750			، بين 179 مالين	-1603160	36135902. ••• 5310000	3.019		• • • • • • • •
11.00	.732	0147052.	. 336.6	1(0)76(0.	77782444	1 777		
11.10	.767	11510300.	526.3	1(03:500.	42947016.	5.712		
	····.tcc·-·	14927746	*** 5: 5 . 1**		*** \$\$\$71556.	4.935		
12.10	. ( 23	19723695.	730.5	10017100.	18249455.	5.742		
13.10	.267	27145739.	1005.4	10071660.	69966163.	6.547		
		-37727760		10084:007-		8.346		
14.10			1141.0	·10694666.	.100301322.	10.733		
15.00			2 5 0 7 . 4 1 1 1 0 5 ··-		13/805094. 	] ] . C Z b	••••	·
			3110.3	1()//////	112126114.	17.224		
			•					
•				•				
			TABLE	2.7.1-1				
			• • • • • • • •			•••••••	•••••	· •·· · •

.

Detailed cask stress calculations were conducted using the cask and lid finite element models discussed in Section 2.6.1, see Figures 2.6.1-1 and 2.6.1-2. The stresses associated with an end impact deceleration of 100.18g's must be combined with maximum normal temperature and pressure stresses, as outlined in NRC Regulatory Guide 7.8. Combined stresses for this set of conditions are found in Case 6, Appendix Section 2.10.3.

Maximum stresses are summarized on the following page.





	Stress Intens	sity (psi)
Center Location	(Membrane)	Surface
Cask Body (Fig. 2.6.1-1)		
Base - Inner Shell	9554 (12)*	14497 (12)
- Outer Shell	27257 (5)	60493 (5)
Side - Lover Quarter		,
- Inner Shell	10935 (15)	13110 (14)
- Outer Shell	44009 (9)	65270 (7)
Side - Mid Height		
- Inner Shell	10372 (25)	10478 (25)
- Outer Shell	18703 (40)	19197 (40)
Side - Top Quarter		
- Inner Shell	13522 (30)	16223 (30)
- Outer Shell	18703 (41)	19197 (41)
Lid (Fig. 2.6.1-2)		
- Base Plate	14181 (2)	27806 (4)
- Side Cone	21238 (9)	34815 (9)
- Top Plate	11472 (16)	38091 (16)

\*( ) Denotes element number.

:

Factors of safety are calculated using allowable stresses at a temperature of  $214.4^{\circ}$ F, based on material properties from Table 2.3-1, as follows:

.



2-97

.







# o <u>Containment Vessel (Inner Shell)</u> (NRC Reg. Guide 7.6) $S_{i} = 20 \text{ ksi}$ S. = 70.28 ksi - Membrane 2.4 $S_m = (2.4) (20) = 48$ ksi $0.7 S_{11} = (.7) (70.28) = 49.20$ ksi - Membrane + Bending 3.6 $S_m = (3.6) (20) = 72 \text{ ksi}$ $S_{11} = 70.23$ ksi o <u>Outer Shell</u> S. = 70.28 ksi For the containment vessel the minimum factors of safety are: o Membrane (Lid Cone): F.S. = <u> 48</u> = <u>+2.26</u> 21.238 • Membrane + Bending (Lid Cone) F.S. = <u>\_\_\_\_\_\_\_ = +2.02</u> 34.815 For the exterior shell, the minimum factor of safety (at lower cask side, Element 7) is: F.S. = <u>\_\_\_\_\_\_ = +1.08</u> 65.27





#### 2.7.1.2 Free Drop Impact, Corner Drop

The second of the two drop tests, discussed in Section 2.11 was a corner drop, lid end down, with package oriented at  $40^{\circ}$  with respect to a horizontal plane. This orientation angle was chosen for two reasons:

- o To minimize overpack clearance margins, see Figure 2.7.1-2.
- To maximize secondary "slap-down" loads on the lid closure and thereby maximize closure bolt loads.

Of note, the earlier engineering development test conducted in July, 1980, was conducted at an orientation angle of about  $60^{\circ}$  with respect to horizontal plane. This corresponded to the traditional c.g. over struck corner prientation.

The test results indicated no change to the cask and no measurable residual atrains excepting for a single closure bolt that could have experienced a strain of as much as 0.44%; about twice the value of yield strain. See Section 2.11.5.1.3 for details on closure bolt experimental strains.

Analytic prediction: of package performance, used to determine cask loads, employed two computer programs described in Appendix Section 2.10.1 - CYDROP and OBLIQUE. CYDROP uses an energy balance technique to determine loads and deformations of the overpack. Since CYDROP assumes all drop energy is absorbed in deformation of the overpack, it provides valid results at the c.g. over struck corner orientation (no rotational motions). CYDROP output for this orientation is provided in Table 2.7.1-2.







At other orientation angles, the force-deflection values generated by CYDROP are used in OBLIQUE, a dynamic analyses model that properly treats rotational motion effects. Figures 2.7.1-1 to 1.7.1-3 summarize predicted overpack responses for all orientation angles from near vertical to near horizontal. Figure 2.7.1-1 clearly shows that near vertical orientation produce maximum forces on the cask. Figure 2.7.1-2 shows that the tested 40° orientation corresponds to minimum clearance between the impact plane and cask "hard points." Figure 2.7.1-3 maps bending moments induced in the cask body.

At the tested orientation, predicted dynamic overpack deformation amounts to 13.9 inches. Post-test measurements of the deformed overpack, Figure 2.11.4-6, shows about 9.1 inches residual deformation following recovery. The foam utilized in these overpacks recovers about 30% following removal of load. This gives a test based dynamic deformation estimated at (9.2/.7=) 13.1 inches; about 6% lower than the analytic predictions.

For cask stress evaluation purpose, moment and thrust forces are a maximum at  $53.75^{\circ}$  with respect to a horizontal plane. At this orientation, cask loads are:

Moment =  $12.634 \times 10^6$  in-1b. Thrust =  $1.399 \times 10^6$  lbs (51.8g)



1.	******				· • • • • • • • • • • • • • •	*	and the second		· · · · · · · · · · · · · · · · · · ·					
	i n	C+C+1(C)	·#[#3 *			· · · · · · · · · · · · · · · · · · ·	•	•	•	10.55.51	. • 1/	01/21	**	41 22
- ( ·	• •		• ••• •• •	· · · - ·	· CHS 1-11	C PACALOR	13343 0461836					•		
••														
•	··· •	••••		PLANE 11"	107			SHERGT		• 015741		01. 2111		05-37
		C7U34							• • • •		LACENT	DF CONI	161 110	4
	•	1121	61N21	11411	10100	seen.	±1+(11C	218734	3110	16.10	61.70	61.10	61.94	61.45
		•••	•••••			103	118-161	(14-(3)	[34741]		11.10	11.30	11.73	
		• : :	12.0	3.	207.	.0	47)):Le.	32.	. 000	100.50	0.00	0.03	0.23	0,00
	1	1.00	- 33.3					12651.		104.00	0.00	0.00	0.77	.1.02
	•	1.10	51.1	11.	24464.	.*	4766109.	10223.	.001	166.00	0.00	0.00	0.03	0.00
	•				4423P,	1.6	+ > > 4 < 00.	26766.	.(L)	100.00	0.30	0.00	0.22	0.00
		1.00	17.5.4	117.	77 4 31.	2.5	9787563. :	LOEVS.	. 334	. 103.33	C. 2C	0.00	0.33	2.00
		3.10	235.4	361	111571.	- <u>22</u>	4661660.	101204.	.011	100.00	0.00	0.00	0.03	1.07
	•••	1.66	1 233.4	117.	1413100	3.3	7814515.	373247.	.019	100.90	0,00		10.01	0.00
	•	4.10	259.9		715645.	1.0	##\$1500.	22/052.	.0/*	100.00	0.00	0.60	0.00	0.03
		5.00	347.9	717.	214777.	10.5	9111000.	494423.	. 6* 6	161.00	0.00	0.00	0.00	0.00
	- 211		- 397.3	.203.	334016.				.015	100.00	0.00	C.00	0.33	3.01
:		10.4	447.*	1115.	240354.	2000 11.5	4.2" ***2603.	#33227.	. 964	130.35	0.0	0.00	0.33	<b></b>
•		1.10	444.4	1221.	LL:035.	16.5	7845:LE.	1043025.	.105	100.00	0.33	0.00	0,04	0.00 **
		7.00	552.1	. 1994" .	.2058237	. 36 * 1	. 0332002	** 1217307. *	154	. 100.00	0.00	0.00	0, 90	0.00 F.
1		7.70	492.7	1934.	SE3265.	20.9	9922303.	1544972.	.111	130.66	0.00	0.00	0.33	1.03 4
;		A.CO	639.7	2223.	EZFASA.	23.2	<b>1936CCO</b> .	1441611.	. 14 5	100.00	0.00	0.00	0.03	0.00
•			714.2	21444	4-21 (2	22.2	9549:00.	2171326.	.215	160.30	0.00	0.93	0.03	.0.07
		9.10	127.7		1:1373.	76.1	4463000.	2334623.	-234	100.00	0.00	0.00	0.33	2.00
		10.00 -		···· · · · · · · · · · · · · · · · · ·	**************************************			2431073.		120.20	. 0.00	6.00	10.60	3.00
		10.3C	+33.2	4211.	923585.	14.5	10003300.	3440744	. 195	100.00	0.00	6.00	0.02	0.00
	:	11.00	**7.5	\$591.	1064353.	39.5	10017000.	4352421.		100.00	0.00	0.00	0,0)	2.22
	•	11.32	1041.4	:19#.	11:3472.	42.7	· 16630300,		. 489	100.00	0.00	.0.00	C. 03	5,05
		12.00	1644.7	5732.	122234.	**.*	10011000.	5500730.		100.00	0.00	0.00	0,53	3.00
		17.30	1147.3	5743.	13:4172.	20.3	10637560.	4161558.	.613	100.00	0.00	0.00	0.07.	0.00
	•	11.00	1146.0		- 1474103	37.3.	10071000	- 1671126.			<u> </u>	0.00	0.00	0.07
		11.00	1700.1	1112.		37.4	10004560.	7644544.			1.13	0.00	0.01	0.00
	1115	34.50	11117.6	* 7 * * *	1///////	• • • • • •	10048000.	44434234		44.37	1.61		0.01	0.00
	<u> </u>	15.00	3888.5	9.59	223004171	6167 13 20	1.37 10111 1011	********	1 (1)	33 03	6.28	1. 10	0.01	0.00
		15.50	1934.2	10555.	2411302-	49.1	30131500.	11434304.	1 145	90.43	1.67	2.35	3.03	2.63 6
	** ** ** *	16.00		11 525.			· - · 1c132000.* *	-12474149.	1.258	. 49.59	1.46		. 0.03.	1 0.65 F
		16-20	+ 2033.4	12510.	2756886.	• 103.6	10145500.	1 \ 20 97 3 .	1. 196	17.10	8.13	3.31	1.26	0.00 .:
	1.	17.00	2011.7	11567.	24213782	169.2	- 10179000.	15639305.	1.536	\$6.17	5.60	5.25	1.47	2.00
	•••••	17.30 -			. '7154977'		10142300	17151908.		. 11'16	6.93	3.64		1.55
•								ľ						
	. <del></del>	<b></b> .		· · · -·- ·			····					-	••	
								1						
	•					-								
	•				. 71	BLE 2.7.1-2		•		•				

• •

**. .** . .

+

• • • •

1

2-101

÷.,

FIGURE 2.7.1-1

## IMPACT FORCES (LBS) -30 FT DROP









OVERPACK DEFLECTION AND RESIDUAL CLEARANCE -30 FT DROP









\$

FIGURE 2.7,1-3

## CASK BENDING MOMENT (IN-LB) -30 FT DROP











The moment load produces a maximum bending stresses at the lower 1/3 point of the cask height (63/3 = 22.7"); corresponding to cask model elements 19 and 35, see Figure 2.6.1-1. At this location, the bending stresses are:

Where: M = 12.634 x 10<sup>6</sup> in-1b. c = 13.50 in.- inner shell 19.00 in.- outer shell

 $I = \overline{\gamma} 4 (k_0^4 - R_1^4) = \overline{\gamma} 4 [19.25^4 - 18.75^4 + 13.75^4 - 13.25^4] = 14642 \text{ in.}^4$ 

Stresses for thrust load effects are obtained by multiplying results for Case 3, Section 2.10.3, by the factor 51.8/100.184 = .5170. Stresses for maximum normal pressure and temperature are obtained from Case 7. Stress predictions are shown below.

			Str	esses (p	si)	
Locations		σ	σ	τ	σ 0	
(LIEment numbe		г 	Z	rz		S.1.
<u>Inner Shell</u> (19)	Bending	-	±11649	-	-	-
(Contain- ment)	Press & Temp (7)	47.	10550	-16	4566	10503
	Thrust (3)	0	-405	18	627	1033
	Combined	+47.	21794 -1504	+ 2	5193	21747. 6697
Outer Shell (35)	Bending	-	±16394	-	-	-
	Press &			_		
	Temp (7)	-289	13750	0	19180	19469
	Thrust (3)		-6178	45	-719	6192
	Combined	-276	23966	45	18461	24242
			-8822			27283





The resultant factors of safety using allowables defined in Section 2.7.1.1 are:

o Containment: F.S. = \_\_\_\_\_48\_\_\_\_ = ±2.21 .
21.747
o Outer Shell: F.S. = \_\_\_\_70.28\_\_\_ = ±2.58
27.283

#### 2.7.1.3 Free Drop Impact, Side Drop

Side drop impacts were not tested in the drop test program; however, the rather shallow angle oblique impact  $(40^{\circ})$  induced significant impact upon the second overpack. This test approximately simulated a side drop event. As noted previously, the cask experienced no damage in these tests.

The most critical phenomena investigated in a side drop evaluation are the bending stresses induced in the cask body by the impacts upon the end mounted overpacks. Behavior of the overpacks has been evaluated using the program, SYDROP, described in Section 2.10.1. Results are shown in Table 2.7.1-3. Peak acceleration of 137.4g was calculated. This produces a midsection bending moment of:

 $M = W1/8 = (27000)(137.4g)(68)/8 = 31.5 \times 10^6$  in-1b.







Using the cask cross section properties gives bending stresses of:

Location

Bending Stress (psi)

Inner Shell Outer Shell <u>+</u> 29,075 <u>+</u> 40,918

These bending stresses must be superimposed upon the maximum normal pressure and temperature stresses found in Case 7, Section 2.10.2. For midsection elements 23 and 39, the resultant stresses are:





	[: 1 E P O P 1 : 1 D	[)	••	• • •					222124	01/	03/22	· • • • •	rt s
- 1 D	}							• • •					
	·			(H) 1-110	710100, 1	SILG DYLAFICA						•••••	,
	<b>,</b> <sup>,</sup>												
	·	44 (FNIH )	*1+*: ++	++++ 1#F+C	1	*****	EMENET +++		011111	MCIIN	64°6041	10 0111	·····
	сторизя			108/1			11541u		+1-70	L1.7C	61.16	61.+2	61.15
	1 (1N3	(157)	11521	11353	161	11N-141	[[#-[#]	12477()		11.10	11.50	12.55	
	······· · · ·		•		•• ••	••••			• •	6.30	0.03	0.00	0.00
		125.5		202202	1.1	4778755.	11305.		160.00	0.36	1.10	0.07	2.22
									-136.00"	···· ( . ( ( `		0.20.	- 0.03
	1.16	AL].]	417.	523911.	12.4		234244.	. 074	160.16	0.95	P. 03	0.02	0.00
	. 1.7:	1,11		127152.	23.1	47537 <u>:</u> C.	377+27.	. 234	130.06	(.((	0.00	0,51	2.00
	-1.7# 1.**	· **?.3		76.230.	1. 15 24.2	7139765363.	542731.	.014	160.36	6.70	6.16	0,00	D. DO
		e>•.•	)(1+,	777574.	23.2	93132:0.	72571e.	. 615	190.00	0.00	6.0)	3.73	3.00 8
	2. ((		1121.	/ 4] 4 3 6 .	31.2	\$71\$(55.	• 31 555 .		101.36	- P. JL	b. Cb		F
			1252.	······································	- 11.(	\$1307:0.	- ]](7)/4.	111	100.73	0.30	0.00	C. 00	0.00 ,
	1 2.14	1120.1	2656-	3236636	11	97552*4.	1613411.	.114	110.00	0.05	( ل. ي	0.1)	0.00
	······································		. 2377.	1171414		.4736266	1172769.	.153	160.00	6.66	0.03	2.07	5. 65
	3.2:	\$727.2	721	1120275.	\$7.7	3E077:C.	7115	.221	133.30	C.00	L.[J	0.07	2.07
	2. : 3	115	2235.	1219452.	44.5	9114553.	2152257.	. 253	165.00	0.00	0.03	0.77	
*				*** 1265537.**				,2+2-	160.96	0,00	8.01	0,00	0.00
Į.		1117.7	3143-	1322715.	\$4.5	VESTCLE.	3045323.	- 315	100.00	0.00	6.00	0.03	0.00
ī	t 4.21	1*****	3404.	3464212	\$2.0	5234756,	3437469.	.156	100.00	- 3 36	A 00	0.03	2.02
ļ	**:50		···· · · · · · ·	. 1413311*	54.5	211116.	3736476.	- 18L	100.00	26.0	r. (A	6.77	0.00
í	4.71	3458.0	4442.	1276348,	19.7	\$146750.	4186167.		100.00	0.00	L. 65	0.00	0.07
1	2.00	1467.5		illess,		V&::((+0.			-100.00'		0.63	- 0. 22	0.00
	5.75			14431.3	20.1	*******	5471347		110.00	0.00	0.00	0.00	0.00
	1 1 1	1.1		2031646	25.3	927.210.	5997687.		100.00	c.00	0.00	0.07	1.22
			··· ++ 72.	- 2356633					- 100.LO		. 0.(0	. 0.33	- 0.00
	1.25	3759.5	76.22	2152359.	38.0	12227:6.	7666643.	.715	100.30	0.00	C.[D	6.30	3.00
	1.10	1171.2	7447.	21 47101.	52.1	414:100.	7647453.	. 771	100.00	C.00	01.0	0.03	n. D5
									12.11	-17.13	- 0.10	0,00	0.50
	1.00	+1712.5	Flug.	3336ct*.	155.8	5564CCP.	. 9174910.	.9?4	16.00	24.00	0.00	0,00	0.03
-	L								-11.59	-21:42	0.00	0.05	
•	1.()	494004											
:	7.75	1710.0	(736.	1752245.	112.3	44137:0.	12068524.	1.015	70.67				
•	······	» : • ¿ ( <u></u> -			165.0				66.07	11.67	11.60	6.00	0.03
	1.15	1111.1	4624.	SAE 2234	- 210.3	4424255,	1236990	· ),2(5	20.67	15.33	21.00	0,00	7.05
		1:2:	10012.	· · · · · · · · · · · · · · · · · · ·				1.529	30:04	"12 22	* 21.17		- A.DO -
•				* 7 7 5 7 7 6.		wial*/6	14136671.	1.4)*	32.33	12.67	21.33	10.17	9.60
		1642.0		111110	115.1	4456750.	17247821.	1,717	\$2.33	11.33	16.67	38.47	0.10
								······;, 177'	32.00	-10.00	- 1 - CC		
•	4.25	1410.0	12452.	1220010	- 117.7	¥9193:5.	156378+3.	1.910	:	10.00	12.00	4.33	14.47
	L									••• •	••••		
:													

. .

U



		Stresses (psi)					
Locations (Element Number) Load		a r	a 2	T TZ	Ф	S.I.	
Inner Shell (23)	Bending	-	±29075	-	-	-	
(containment)	Press & Temp		11240	-12	4440	11172	
	Combined	68.	40315 -17835	-12	4440	40247 22275	
<u>Outer Shell</u> (35)	Bending	-	±40918	-	-	-	
	Press & Temp	-283	13560	-12	19010	19293	
	Combined	-283	54478 -27358	-12	19010	54761 46368	

.

.



The resultant factors of safety, using allowables defined in Section 2.7.1.1 are:

o Containment: F.S. = <u>48</u> = <u>+1.19</u> 40.247

o Outer Shell: F.S. = <u>70.28</u> = +1.28 54.761

.



#### 2.7.2 Puncture

A 40 inch drop onto a 6 inch diameter pin can occur only on the cylindrical body or side wall between the overpacks.

The Nelms puncture relation is given as:

 $t = (\frac{1}{2}/5)^{0.71}$ 

Where: t = shell thickness = 1/2 (outer shell) and 1/4 (fireshield)

w = cask weight - 1bs.

S = ultimate tensile strength of outer shell

= 70,280 psi @ 214.4°F

The package weight causing puncture is:

 $W = St^{1.4}$ 

Upon impact, the fireshield is deformed and brought into direct contact with the outer shell. At this point in the impact scenario, the fireshield is effectively "backed" by the outer shell and lead. The package weight to cause puncture of this 1/4" fireshield is:

 $W_f = (70.280)(.25)^{1.4} = 10,091$  lbs.

The corresponding weight to cause puncture of the 1/2" outer shell is:

 $W_{g} = (70,280)(.5)^{1.4} = 26,631$  lbs.

\*Shappert, L.B., "Cask Designers Guide", ORNL-NSIC-68, Page 18.



Thus, the total weight at which puncture of both fire shield and outer shell occurs is:

$$W = W_{e} + W_{e} = 36,722$$
 lbs.

The actual package weight is 27,000 lbs; therefore, the factor of safety for puncture resistance on an energy basis is:

$$F.S. = \frac{36.720}{27,000} = \frac{+1.36}{27,000}$$

When the cask impacts the puncture pin the force imposed upon the cask is estimated as:

$$F_{I} = k_{S} A_{I}$$

 $k_{s} = Dynamic flow pressure of stainless = 45,000 psi *$  $A_{I} = 7/4 (R_{c}) = 7/4 (6.0)^{2} = 28.27 in^{2}$ 

 $F_{I} = (45,000) (28,27)$ = 1.272 x 10<sup>6</sup> 1bs.

This force induces a moment at the midsection of the cask. The moment is estimated as:

 $M = F1 = (1.272 \times 10^{6})(68) = 10.8 \times 10^{6} \text{ in-lb.}$ 

\*Shappert, L.B., "Cask Designers Guide:, ORNL-NSIC-68, Page 64.



Using the section properties from Section 2.7.1.3 to calculate bending stresses and the pressure and temperature stresses reported gives stress intensities and factors of safety of:

		Stresse	s (psi)		
		Bending	<u>5, ī</u>	F. S.	
0	Inner Shell (Containment)	±9958	21130	48/21.13= <u>2.27</u>	
o	Outer Shell	±14014	27857	70.28/27.86= <u>2.52</u>	



## 2.7.3 Thermal

#### 2.7.3.1 <u>Summary of Pressures and Temperatures</u>

The maximum temperatures and pressures resulting from the hypothetical accident conditions presented in Section 3.5.3 and 3.5.4 are summarized below:

Maximum Containment Vessel Pressure = 184.4 psig Temperatures:

- o Cavity (Inner Shell) =  $371.43^{\circ}F$
- o Outer Shell Sides = 434.56°F
  - o Outer Shell Ends = 363.35°F



Revision D

## 2.7.3.2 <u>Differential Thermal Expansion</u>

Differential thermal expansion between the two shells of the cask and the lead shield produces significant stresses. Stresses have been assessed by use of the finite element models discussed in Section 2.6 1, see Figure 2.6.1-1 and 2.6.1-2.

## 2.7.3.3 Stress Calculations

Stress calculations for pressure and thermal loads were performed using the conditions summarized in Section 2.7.3.1, as Cases 2 and 1 of Appendix Section 2.10.3, respectively. These two cases are combined as Case 4 of the referenced section.




•

	Stress Intensity (rol)		
Location	Center	Surface	
Cask Side (Figure 2.6.1-1)			
o Lover Quarter	+		
- Inner Shell	27740 (18)	29042 (18)	
- Outer Shell	53380 (9)	62805 (9)	
o Mid Height			
- Inner Shell	32049 (26)	33611 (25)	
- Outer Shell	34901 (35)	36641 (35)	
o Top Quarter			
- Inner Shell	36322 (30)	44216 (30)	
- Outer Shell	36860 (46)	37917 (46)	
Lid (Figure 2.6.1-2)			
- Base Plate	18760 (5)	33325 (5)	
- Side Cone	37963 (8)	58210 (10)	
- Top Plate	31120 (16)	53494 (14)	

# Maximum Stresses are summarized as follows:

.

\*( ) Denotes element number.

.



..

Ċ

# 2.7 4 <u>Comparison With Allowable Stresses</u>

Allow she stresses based upon the temperatures of Section 2.7.3.1 and Table 2.3-1 are as follows:

(T=371.43°F) · o <u>Containment Vessel (Inner Shell)</u> 5\_ = 19.07 ksi S. = 64.86 ksi - Membrane:  $2.4S_{\rm m} = 45.77$  ksi .75, = 45.40 ksi - Membrane + Bending 3.65 = 68.65 ksi S. = 64.86 ksi  $(T=434.56^{\circ}F)$ o Outer Shell  $S_{11} = 64.09 \text{ ksi}$ For the containment vessel, the minimum factors of safety are:  $F.S. = 45.40 = \pm 1.20$ o Membrane 37.963 o Membrane + Bending:  $F.S. = 64.86 = \pm 1.11$ 58.21 For the outer shell, the minimum factor of safety is:

F.S. =  $64.09 = \pm 1.02$ 62.805









Therefore, it is safe to conclude that the containment vessel can safely react any anticipated loading condition without experiencing detrimental stresses.

### 2.7.4 Immersion-Fissile Material

The requirement of 10 CFR 71.73 (c)(4) is not applicable, since no arrangement of the TMI-2 core debris samples contained in the shielded debris canisters can result in a critical configuration under any normal or accident conditions with the total mass limited to 29kg by the volume restriction. The actual mass of approximately 15 kg will provide a safety factor of five.

# 2.7.5 Immersion - All Packages

10 CFR 71.73 (c) (5) requires an immersion in water with a pressure of 21 psig for eight hours. Review of the stresses in Section 2.6.4 for a 25 psig pressure indicates the stresses are low, and this test will have no significant effect on the package.

#### 2.7.6 Summary of Damage

The structural integrity of the Model 1-13C II Package has been substantiated ' for normal transport conditions as well as hypothetical accident conditions.

Damage to the Model 1-13C II Package that results from the hypothetical accident condition is:

 Impact limiters crush during the 30 foot drop condition producing a maximum bolt load on the containment vessel. Bolt stresses are maintained at or below yield. Vessel stresses are less than those prescribed by NRC Regulatory Guide 7.6.





- 2. Small local deformations to the external shell may result during the 40 inch puncture condition. There will be no loss of shielding and the containment vessel will not be deformed.
- 3. Presence of the overpack and heat shield limit temperatures in the containment vessel walls to less than 372°F, and internal pressures to 185 psig. Geometry and temperature integrity :: the seals is maintained.

# 2.8 Special Form

Not applicable since no special form is claimed.

# 2.9 Fuel Rods

Not applicable since fuel rod cladding is not considered to provide containment of radioactive material.

# 2.10 Structural Evaluation Appendices

This section contains three blocks of informational appendices supporting the structural evaluations of the 1-13C II Cask presented in Sections 2.1 through 2.9.

- Section 2.10.1 Summarizes the methodology of computer programs utilized to demonstrate structural compliance of the package with applicable provisions.
- Section 2.10.2 Contains finite element stress summaries for the cask under normal and accident conditions.
- Section 2.10.3 contains finite element stress summaries for the cask lifting lug support ring.



# 2.10.1 Analytic Methods

This section briefly documents the methodology employed for computer programs used to demonstrate compliance of the package with applicable provisions of 10 GFR 71 under normal and accident conditions. The first three subsections deal with the calculation of external and internal forces imposed upon the package, when subjected to drop events. These three subsections describe techniques and computer programs developed by Nuclear Packaging, Inc., of Tacoma, Washington.

The fourth section describes the E3SAP finite element code employed for detailed evaluation of package stresses under normal and accident conditions. The E3SAP program was developed by Boeing Computer Services Company based upon the well known SAP IV program developed by Dr. E. L. Wilson.

### 2.10.1.1 Overpack Deformation Behavior

The package is protected by foam filled energy absorbing end buffers, called overpacks. For purposes of analysis, the overpacks are assumed to absorb, in plastic deformation of foam, the potential energy of the drop event. That is, the analyses assume that none of the drop potential energy is transferred to kinetic or strain energy of the target (the "unyielding surface" assumption of 10 CFR 71) nor strain energy in the package body itself.

There are three orientations of the package where the potential energy of drop is assumed totally absorbed by plastic deformation of the overpacks. At other orientations, where rotational effects are important, the methods outlined in Section 2.10.1.2 are employed. These three orientations where rotational (or pitch) motions play no role in the evaluation of the impact event are:







o End Brop - on the circular end surface of the overpack.

- o <u>Side Drop</u> on the cylindrical side surface of the overpacks.
- <u>Corner Drop</u> with package center of gravity directly above the struck corner of the overpack.

For these three orientations, the prediction of overpack behavior can be approached from straightforward energy balance principles:

$$E = W(h + \delta) = \int_{c}^{\delta} F_{x} dx \qquad (1)$$

Where: W = Package weight h = Drop height  $\delta = Maximum overpack deformation$   $F_{\chi} = Force imposed upon target and package by the$ overpack at a deflection equal to x.

The left-hand term represents the potential energy of the drop. The right hand term represents the strain energy of the deformed overpack.

Each of these three orientations is treated by an individual computer program reflecting the differing geometry characteristics of each event. All three employ common energy balance techniques to assess maximum overpack deformations. All three employ a common description of the crushable energy absorbing foam.

This foam typically exhibits a stress-strain plateau of nearly constant stress up to a total strain of 40-60%. Above this strain value, pronounced strain hardening effects commence reflecting the collaspe or consolidation of the entrapped bubbles within the foam. Accordingly, a tabular definition of foam stress-strain relations is employed in each of the three computer programs. This tabular definition is taken directly from measured properties and accurately reflects the strain hardening behavior of the foam up to strains of 90-95%.



This discussion of these three computer programs proceeds from the geometrically simple (end drop) to the most complex (corner drop).

# 1. End\_Drop\_(EYDROP)

The force produced by the overpack is simply:

- $F_{x} = A\sigma_{e}$  (2)
- Where:  $A = \pi_{-}D^{2}$ , the end area of the package. 4 D = effective diameter of package  $\sigma_{e} = \beta(e)$ , the foam crush stress at a strain of e (3)  $\beta(e) =$  the tabular definition of foam stress strain properties
  - $e = x/x_{u}$
  - x = deformation
  - $x_{i}$  = end thickness of overpack.

EYDROP performs the calculations outlined in Equations (1) to (3) for a trial range of deformation values,  $\delta$ . For each trial value of total deformation, the energy balance of Equation (1) is monitored and reported. Solution for total overpack deformation is found by an interpolated balance of Equation (1).



# 2. <u>Side Drop (SYDROP)</u>

STDROP differs from the end drop solution only in the fact that both deformation and strain vary from point to point and total force, at a given crush depth, must be found by geometric integration over these points. The details on this geometry are found in Figure 2.10.2-1. For each trial deformation value, the force is found as:

$$F_{\delta} = 2L \int_{0}^{x_{max}} \sigma_{ex} dx$$

Where: L = effective length of overpack

$$x_{max} = [r_0^2 - (r_0 - \delta)^2] \quad 1/2$$
(5)  
$$\sigma_{ex} = \phi(e_x), \text{ tabular definition of foam}$$

stress-strain properties.

 $e_{\rm X}$  = The foam strain at location x. The strain at a point x is found by reference to Figure 2.10.2-1 as:  $e_{\rm X} = \frac{{\rm Crush \ Depth}}{{\rm Original \ Thick.}} = \frac{\delta - r_0 (1-\cos\theta)}{r_0 \cos\theta - r_1 \cos\nu}$ Where:  $\theta = \sin^{-1} ({\rm x}/r_0)$ 

 $v = \sin^{-1} (x/r_1)$ 

3. <u>Corner Drop (CYDROP)</u>

CYDROP is like SYDROP excepting that a two dimensional geometric integration is required to assess the overpack crush force at each deformation. A detailed explanation follows.



. ...









.

· ·· •

-

CYDROP treats the corner impact of a cylindrical package upon an unyielding surface. The package itself consists of a cylindrical payload portion surrounded by a larger cylindrical volume composed of a crushable media. So long as the deformations of the crushable media are modest, the problem may be approximately solved by assuming a uniform crush stress exists over the ellipitcal surface of the crush plane (contact surface). CYDROP was developed specifically to address problems of large deformations of this crushable media and to treat geometries where the cylindrical overpack envelope possesses axisymmetric cylindrical voids (e.g. does not completely cover the cylindrical ends of the payload package).

The large deformation behavior of the crushable media is uccomodated by determining the actual stain of the crushable media at a point. This strain is used to determine the corresponding stress from an implicit tabular definition of media stress-strain characteristics. The total crush force is found by a double integration over the contact area of the crush plane.

Strain energy absorbed by the crushable media is determined by integrating the crush force and its associated deformation. The package is assumed to be at "rest" when the computed strain energy value equals the applied drop energy.

The geometric calculations for the contact surface and the associated strains are carried out using a moving (x, y, z) coordinate system in which the x-y plans corresponds to the crush plane, see Figure 2.10.1-2. The crush plane itself represents a segment of an ellipse. The contact area is this ellipse segment, provided no cylindrical end void exists. When a cylindrical end void exists, the contact area of the crush plane is reduced by the removal of a second elliptical region associated with the projection of this void into the contact plane.



Calculation of strain is somewhat more complex. In principal, the distance from point (x,y) in the crush lane to the payload is found and denoted,  $Z_{top}$ . Similarly the distance to the undeformed external overpack envelope is found and denoted,  $Z_{bot}$ . The strain represents deformation divided by original thickness, or:

At any point (x, y), the calculation of Ztop may follow three branches, according to location. The three possible branches relate to the payload surface intercepted. They are:

## The Circular Bottom of the Payload

The bottom of the payload cylinder describes an ellipse in the crush plane. If (x,y) is inside this ellipse, the point is considered "backed" by the bottom of the payload. An exception to this general statement is noted in the discussion of the "Unbacked Region," see below.

#### The Cylindrical Surface of the Payload

The cylindrical surface of the payload describes a rectangular region tangent to the payload bottom ellipse at its major axes. If (x,y) is outside the bottom ellipse yet possesses an x coordinate less than the radius of the payload bottom, the point is considered "backed" by the payload cylinder.

#### Unbacked Regions

Unbacked regions are of two forms - those associated with the cylindrical end void and those near the external surface of the overpack. The unbacked region associated with the end void is a point in the crush plane which lies within the ellipse defined by the void circle lying in the plane of the payload bottom. The unbacked region associated with points near the overpack extrimities is defined by those points (x, y) where the x coordinate exceeds the radius of the payload volume. Points which are "unbacked" employ a nominal crush stress for force integration purposes.





2-124

The calculation of Z<sub>bot</sub>, the distance to the undeformed overpack envelope, may follow two branches. These branches correspond to intercepts with either the cylindrical surface of the overpack or the circular end of the overpack.

The analytics describing the geometry discussed above, consists of the sequential application of a series of geometric transformations of surfaces described in the coordinates of the cylindrical package (X, Y, Z) to the coordinates of the contact plane (x, y, z). The surfaces in package corrdinates are:

 $\frac{\text{Overpack Cylinder}}{\chi^2 + \chi^2 = R_c^2}$ Overpack Bottom Circle  $x^2 + y^2 = R_c^2$  $= -1_{c}/2$ Z  $\frac{Payload Cylinder}{X^2 + Y^2 = R_p^2}$  $\frac{Payload Bottom Circle}{X^2 + Y^2} = R_p^2$  $= -1_{p}/2$ Z  $\frac{\text{Void Circle at Payload}}{\chi^2 + \chi^2 = R_f^2}$  $= -1_{\rm p}/2$ Z Void Circle at Overpack Exterior  $x^2 + y^2 = R_f^2$  $= -1_{c}/2$ Z





•

••

**. .** .

•

Revision O

# 2.10.1.2 <u>Collique Impact Dynamic Analysis</u>

Impact at arbitrary orientation angles differ in two major respects from those that occur at angles corresponding to stable or neutral equilibrium (end, side, and c.g. over struck corner). In the neutral equilibrium conditions the entire initial kinetic energy of drop is transformed into strain energy associated with plastic deformation of the overpack. At arbitrary orientation angles, only a portion of this kinetic energy is transformed into strain energy at the impacted end. The remainder of this kinetic energy becomes rotational motion of the package. The solution approach must properly reflect the continually changing transformation of initial translational kinetic energy into rotational kinetic energy and plastic deformation of the overpack energy absorber.

The second major difference between neutral equilibrium impacts and arbitrary angle impacts relates to the rather different load-deflection behavior of the overpacks at low angle  $(10-30^{\circ})$  from horizontal) orientations. Under neutral equilibrium conditions a major portion of the crush footprint is backed by the cylindrical body of the package, allowing strain hardening effects to stiffen the overpack load-deflection relation. At low angle orientations  $(10-30^{\circ})$  from horizontal) much of the overpack crush footprint is unbacked. Thus, the low angle load- deflection relations are initially quite soft, then abruptly harden as portions of the crush footprints approach horizontal attitudes, this terminal stiffening phenomena becomes more pronounced.

There are two potential solution paths to problems of this nature - a momentum formulation or a direct solution of the equations of motion. The momentum approach provides an easy and simple means to assess the transformation of translational initial velocities into rotary velocities; hence, total plastic strain energy absorbed by the overpack energy absorber. Unfortunately, this momentum formulation does not produce intermediate values of crush force and crush deformation needed to assess overpack attachment forces nor does it conveniently provide a means to incorporate the varying load-deflection relationships of the overpack as a function of orientation angle. Thus, a direct solution of the equations of motion was selected. The model is illustrated on the following page.





. ....



.

The three key problem variables (grush force, F, crush depth, ; and prientation angle, 0) all vary with time for a given orientation angle,  $0_0$ . The crush force is assumed to act at the centroid of th elliptical crush footprint. For the model shown in the sketch, three independent second order differential equations of motion can be formed:

$$\begin{split} \dot{HX} &= F_{\chi} \\ \dot{HY} &= F_{\chi} - Mg \\ \dot{I\theta} &= \left[ \left( \frac{\delta}{\sin \theta} \cdot \frac{(a-c)}{c} + t_{B} + L/2 \right) \sin \theta \right] F_{\chi} - \left[ \frac{\delta}{x} - \left( \frac{\delta}{\sin \theta} \cdot \frac{(a-c)}{c} + t_{B} + L/2 \right) \cos \theta \right] F_{\chi} \end{split}$$



Where: M = the package mass = pl

- F = the crush force
- g =the gravitational constant, = 386.4 in/sec<sup>2</sup>
- I = the rotational mass moment of inertia =  $(p1^3/12)$
- R = the radius of the body
- L = the length of the body
- $t_{R}$  = overpack bottom thickness
- 9 = the instantaneous orientation angle of the package with
   respect to the horizon
- p = the mass per unit length

a, c,  $t_{\rm R}$ ,  $\bar{x}$  are footprint geometry quantities defined in Section 2.10.2.3.



These differential equations are integrated subjected to initial conditions, associated with the moment of impact, t = 0, of:

$$X = 0, Y = 0, \theta = \theta_0$$

$$\dot{X} = 0, \dot{Y} = \dot{Y}_0, \dot{\theta} = 0$$

$$\theta_0 = \text{impact angle, varies}$$

$$\dot{Y}_0 = \sqrt{2gh}$$

$$h = \text{drop height}$$

Each of the above differential equations requires a continuously updated value of force, F, reflecting both crush depth and package orientation, or:

$$\Sigma = \Psi(\xi_0; \theta)$$



This continuously updated value of force, F, is supplied to the integration process by means of a two dimensional Langrangian interpolation of crush depth,  $\delta_y$ , and orientation angle, 9. The tabular data used in this interpolation consists of a series of complete force-deflection relations for separate orientation angles developed via the CYDROP (and SYDROP) computer programs, described in Section 2.10.2.1. The deflection,  $\delta_y$ , is expressed in terms of problem variables as:

$$\delta_y = \frac{L}{2} (sin\theta - sin\theta_0) + R (cos\theta - cos\theta_) - y$$



The foregoing analysis process for evaluating impacts at oblique orientations was consolidated in a NuPac developed computer program, OBLIQUE. OBLIQUE integrates the equations of motion for each value of orientation angle versus time until maximum values are found for crush force, crush deformation, shear and body bending moment. At each incremental time step (incremental crush deformation) overpack attachment moments are computed, scanned for maximum "slues and output. By sweeping through a series of initial orientation angles, the maximum values of all internal loads are found.

# 2.10.1.3 Overpack Force Analyses

This section treats both external and internal forces imposed upon the package. Key to the treatment of external force application locations is an understanding of crush footprint geometry.

The crush footprint is a sector of the ellipse shown the next page. The location of the centroid,  $\overline{x}$ , is calculated relative to the ellipse origin.







From the above sketch, the geometric properties of the elliptical crush footprint are:

```
a = r/\sin \theta
c = \delta/(\sin \theta \cos \theta)
```

The area, A, and the centroidal offset,  $\overline{x}$ , of the crush footprint are derived as:





When 
$$c \neq a$$
:  

$$\lambda = 2\int_{a-c}^{a} ydx ; y = \frac{b}{a}\sqrt{a^{2}-x^{2}}$$

$$\lambda = \frac{2b}{a}\int \sqrt{a^{2}-x^{2}}dx$$

$$\lambda = \frac{b}{a}\left[\frac{\pi a^{2}}{2} - (a \cdot c)(2ac - c^{2})^{\frac{1}{2}} - a^{2}sin^{-1}(\frac{a-c}{a})\right]$$

$$\lambda \overline{x} = \frac{2b}{a}\int_{(a-c)}^{a} x\sqrt{a^{2}-x^{2}}dx$$

$$\overline{x} = \frac{2b}{3a}(2ac - c^{2})^{\frac{3}{2}}/\lambda$$
When  $a \leq c \leq 2a$ :  

$$\lambda = \pi ab - \lambda^{a}$$

$$\overline{x} = \frac{\lambda^{a} \overline{x}^{a}}{\lambda}$$

$$\lambda^{a} and \overline{x}^{a} are as defined for  $\lambda$  and  $\overline{x}$ , except that  $c^{a}$  replaces  $c$ .  

$$c^{a} = 2a - c$$
When  $c > 2a$ :  

$$\lambda = \pi ab$$$$

·

<del>x</del> = 0

•

· ·

.



· ·

F

# Overpack Attachment Forces

For most orientations and crush depths, the overpack crush force is transmitted to the cask body in direct compression; hence, the forces transmitted to the circumferential overpack attachments are near zero. This is not true for near vertical and near horizontal orientations of the package, at very modest crush deformations and crush forces. In these very limited situations, the center of pressure of the crush force can lie beyond the outer extremeties of the cask body and exert a resultant moment force upon the overpack attachments. Significantly, these moments exist only for very modest crush deformations and crush forces, regardless of orientation angle. This is because larger crush deformations move the center of pressure toward the cask body. At maximum crush depth and maximum crush force, for all angles of orientation, there are no overpack attachment moments because the overpack interface forces are all direct compression. The near vertical and near horizontal orientations where attachment moments exist are sketched below:







Attachment Moment = M = F.e. or F.e.

"There: e = Moment Arm about adjacent corner

e = Moment Arm about opposite corner

The location of the crush force can be approximated as the centroid of the crush footprint area. This approximation is consistently conservative. Specifically, for both near vertical and near horizontal orientations, foam strain hardening effects tend to move the center of force from the geometric center of the crush footprint toward the cask body. In both instances, this tendency reduces the actual moment arm of the crush force to less than that predicted by the location of the crush footprint centroid. The moment arm, as defined by crush footprint geometry is found below.

The location of the center of pressure relative to the opposite and adjacent corners of the cask body can be obtained from the geometry of the sketch shown on the following page.





Revision 0 20 4 , s - 95"  $q = r_0 \sin \hat{\tau}$ x -| F Previously defined a C f f = g cos ÷ \**ē**;+  $g = t_{B} (a-c) \cos f$  $\overline{e} = \overline{x} - f$ 



2-136

The location of the center of pressure measured from a normal to the crush plane passed through the intercept of package center line and body baseplate (point c) is:

 $\overline{e} = \overline{x} - \left[ t_{a} + (a - c) \cos \theta \right] \cos \theta$ 

The moment arms,  $e_0$  and  $e_a$ , representing the distance from the center of pressure to the corners of the cask body, are thus given as:

 $e_{0} = - (e+c)$ ; Moment Arm about opposite corner

e\_ = (e-q) ; Homent Arm about adjacent corner

Sign convention for these arms is such that the moment (F.e<sub>o</sub>) produces a clockwise (separation) moment about the opposite corner and moment (F.e<sub>a</sub>) produces a counter-clockwise (separation) moment about the adjacent corner. In other words, a positive moment must be resisted by overpack attachment bolts whereas a negative moment implies that the center of pressure is totally resisted by compressive interface forces and there are no attachment bolt loads.

In summary, the attachment moment interface forces between the overpack and body have been derived in terms of package geometry and three problem variables: orientation angle, crush force and crush deformation.



2-137

# Internal Forces

The cask is idealized as a beam impacting on the lower end. The equations of motion are formed and used to define state-wise accelerations. These accelerations, in conjunction with the unit mass of the package, form forces which vary along with the length of the package. When integrated, these forces provide a complete definition of internal thrusts, shears and moments for the package as a function of total impact force and orientation angle. The derivation is as follows:



e





For a planar rigid body system, the behavior is totally defined by a solution of the three equations of equilibrium written at the c.g. if the rigid body. In the above sketch, local coordinates are defined at the c.g. with axes parallel and normal to the beam. The end impact force is resolved into components parallel to these local axes. Summation of forces at the c.g. leads to three rigid body equations of motion:

<u>Sum of Normal Forces</u> -  $\dot{MY} = F \sin \alpha$ , <u>Sum of Longitudinal Forces</u> -  $\dot{MX} = F \cos \alpha$ , <u>Sum of Moments</u> -  $I\bar{\partial} = -F1/2 \sin \alpha$ .

i 🏟

Where:

M = pl, the mass of the body  $I = Ml^2/12 = pl^3/12$ ; the mass moment of inertia of the body

p = the mass per unit length of the body

 $\alpha$  = the vertical orientation angle

Substituting for the mass and inertia terms:

 $\ddot{Y} = F/pl \sin \alpha$  $\ddot{X} = F/pl \cos \alpha$  $\ddot{\Theta} = -6F/pl^2 \sin \alpha$ 



2-139

The normal and longitudinal acceleration at a point r are:

$$\ddot{S}_{n} = \ddot{Y} + (r - 1/2) \ddot{z}$$

$$= \frac{F \sin \alpha}{c t^{2}} \left[ i - 6 (r - 1/2) \right]$$

$$= \frac{2F \sin \alpha}{c t^{2}} \left( -3r + 2i \right), \text{ (varies with } r)$$

$$\ddot{S}_{r} = \ddot{X} = \frac{F \cos \alpha}{c t^{2}}, \text{ (a constant)}$$

The lateral inertial force acting on the body at the rth location is:



dv drr = -ssn

The corresponding expression for shear is found by integrating this lateral force from the free end to the rth location:

$$V_{r} = \frac{2F \sin \alpha}{r^{2}} \int_{t}^{r} (-3r - 2r) dr = \frac{2F \sin \alpha}{r^{2}} \left[ -\frac{3}{2} (r^{2} i^{2}) + 2i (r - i) \right]_{t}^{r}$$
$$= \frac{2F \sin \alpha}{r^{2}} \left[ -\frac{3}{2} r^{2} + \frac{3}{2} r^{2} + 2i r - 2r^{2} \right]$$
$$V_{r} = \frac{F \sin \alpha}{r^{2}} (3r^{2} - 4r + r^{2})$$

.

• •

Similarly, the corresponding moment is found by integration of the shear expression:

$$\frac{di!}{dr}r = v_r$$

$$M_r = \frac{F \sin 2}{r^2} \int_{1}^{r} (3r^2 - 4ir + i^2) dr$$

$$M_r = \frac{F \sin 2}{r^2} \left[ (r^3 - i^3) - 2i (r^2 - r^2) + i^2 (r - i) \right]$$

$$= \frac{F \sin 2}{r^2} \left[ r^3 - i^3 - 2ir^2 + 2i^3 + i^2 r - i^3 \right]$$

$$M_r = \frac{F \sin 2}{r^2} \left[ r^3 - 2ir^2 + i^2 r \right]$$

In order to verify these expressions for shear and moment, they are evaluated at the boundaries, r = 0, 1.

$$\frac{r = \ell}{v_r} = \frac{F \sin \alpha}{t^2} (3r^2 - 4ir + t^2)$$
$$= \frac{F \sin \alpha}{t^2} (3t^2 - 4t^2 + t^2) = 0;$$

(

$$H_{r} = \frac{F \sin \alpha}{t^{2}} \left[ r^{3} - 2tr^{2} + t^{2}r \right]$$
$$= \frac{F \sin \alpha}{t^{2}} \left[ t^{3} - 2t^{3} + t^{3} \right] = 0;$$

$$\frac{r = 0}{v_r} = \frac{F \sin \alpha}{t^2} (3r^2 - 4ir + i^2)$$
$$= \frac{F \sin \alpha}{t^2} (i^2) = F \sin \alpha;$$
$$M_r = \frac{F \sin \alpha}{t^2} [r^2 - 2ir^2 + t^2r] = 0;$$

Maximum moment occurs when the shear term,  $V_{r}$ , equals zero. For this to occur

 $3r^2-41r+1^2 = 0$ , and the location of the moment mini/max is found as:

$$r = \frac{42^{\pm} \sqrt{16z^2 - 4(3)(z^2)}}{6} = \frac{4z^{\pm} 2z}{6} = z, \frac{2}{3}$$

Substituting r = 1/3 into the moment expression:

$$M_{r} = \frac{F \sin \alpha}{t^{2}} \left[ r^{3} - 2tr^{2} + t^{2}r \right]$$

$$M_{hax} = \frac{F \sin \alpha}{t^{2}} \left[ \frac{t^{3}}{27} - \frac{2t^{3}}{9} + \frac{t^{3}}{3} \right]$$

$$= Ftsing \frac{1 - 6 + 9}{27}$$

$$M_{Max} = \frac{4}{27} Ftsing, at r = t/3$$

•

The location of minimum shear can be found where the lateral force expression equals zero:

$$\frac{dv_{r}}{dr} = \frac{-2F \sin 2}{t^{2}} (-3r+2t)$$
$$r = \frac{2}{3} t.$$

The magnitude of axial forces can be found as a function of location as:

$$\frac{dT}{dr} = -\rho S_r (a \text{ constant})$$
$$T = -\rho S_r \int_1^r dr = -\rho S_r (r-\ell)$$
$$T = F \cos 2 (1-r/\ell)$$

For convenience, the package internal forces are summarized as:

Force	Expression	Maximum	<u>Minimum</u>
• Thrust	F cos a (1-r/1)	F cos a (r=0)	0(r=1)
* Moment	$\frac{F \sin \alpha}{t^2} (r^3 - 2tr^2 + t^2 r)$	4 27 Fl sina (r=1/3)	0(r=0, 1)
• Shear	$\frac{F\sin\alpha}{t^2}(3r^2-4tr+t^2)$	F sin a (r=0)	0(r=2/31, 1)

These forces are graphically illustrated on the next page.





2-143

.



**(** 

Revision 0

# PACKAGE INTERNAL FORCES UNDER END IMPACT





••••

•



# E<sup>3</sup>SAP

# GENERAL INFORMATION

 $E^3SAP$  (the Easy. Efficient, and Enhanced Structural Analysis Program) was originally developed as SAP IV at the University of California.  $E^3SAP$ , pronounced E-three-SAP, is the BCS version of the program incorporating extensive modifications and enhancements which provide

- Greater convenience
- Improved utility
- Lower cost solutions

# ANALYSIS CAPABILITIES

The purpose of  $E^3SAP$  is to perform linear, static, and dynamic analysis of engineering structures and piping systems  $E^3SAP$  offers comprehensive structural capabilities while providing a problem solving method that is easy to learn and easy to use. The analysis capabilities and assumptions inherent in  $E^3SAP$  may be broadly categorized as.

- Linear static analysis
- Natural frequency and mode shape determination
- Response spectrum analysis for earthquake shock studies
- Time history response to arbitrary force or acceleration functions
- Solution approaches based on proven state-ofthe-art finite element methods
- Orthotropic and temperature dependent material properties for a number of situations

# MAINSTREAM:EKS

Theoretical derivations assuming linear and reversible materials

# TYPICAL APPLICATIONS

 $E^3SAP$ , one of the most widely applicable structures programs available today, is used to evaluate a variety of structures, including

- Framed Structures
  - Buildings
  - Bridges
  - Off-shore oil rigs
  - Piping systems
  - Railcars
  - Transmission towers
- Plate and Shell Structures
  - Nuclear containment structures
  - Pressure vessels
  - Storage tanks
  - Ships and barges
- Solid Bodies
  - Dams
  - Tunnels
- Axisymmetric Structures
  - Rotating machinery
  - Pressure vessels
  - Storage tanks





# BCS ENHANCEMENTS

ICS enhancements provide important benefits. For example:

- Greater Convenience
  - Problems may be solved interactively for quick real-time answers or in batch for minimum cost
  - Simplified control card procedures are provided for both normal and restart runs
  - Concentrated loads may be specified in arbitrary order
  - Nodal data may be specified using the mesh generation feature
- Comprehensive Debug Aids
  - Exhaustive error checking is performed to lessen manual debugging



- Automatically generated model details are "tagged" for easier identification
- Stiffness matna printout is provided to assist in debugging
- Auxiliary element characteristics (length, area, volume, weight) are provided to aid in interpretation and debugging
- Improved Reliability
  - = Known SAP IV errors have been removed
- Enhanced Output
  - Output formats have been reorganized for clarity
  - Simplified result file has been incorporated for ease of post-processing
  - Interactive and hardcopy plots of the model and output results are available for checking and interpretation. Users are free to select om a variety of devices, such as Tektronix

interactive graphics terminals and CalComp. Gerber, or Comp80 plotters.

- Lower Cost and Greater Efficiency
  - Automatic bandwidth resequencing features assure an efficient set of equations-solved at a low cost
  - During execution, core memory is dynamically allocated to minimize user cost and simplify the job control statements
  - Improved solution algorithms and file management for lower run cost

# FEATURES

# Input/Output

 $E^3SAP$ , data follows a fixed format and consists of a description of the model, its properties, and the loading conditions relative to the selected analysis. Model geometry may be expressed in either cartesian, cylindrical or sphencal coordinates. Input to  $E^3SAP$  is generally a deck of punched cards or a disk file containing card images. CMEDIT, the powerful editor offered on the interactive (KIT) system of EKS, provides the means for conversationally creating and modifying data files for use with  $E^3SAP$ .

Output is tabular, and the input data, as well as the "tagged" generated and computed quantities, are listed. Results, such as displacements, stress, reaction, frequencies and mode shapes, and dynamic time histories are conveniently displayed for easy interpretation. The output may optionally be scanned and spot-checked at a low speed terminal prior to routing to an RJE terminal or to the BCS Data Center for printing. Additionally, graphic displays of displacements and mode shapes are available if desired.

# Multiple Job Processing

Multiple jobs can be processed within a single computer run by appropriate stacking of the input data. This time saving feature is particularly useful for the parametric evaluation of a structure.





<sup>3</sup>SAP may be executed interactively or in batch .ode Interactive execution of  $E^3SAP$  directs solution results to the user's output file in real-time. When time is not a factor, however, economical solutions may be achieved by executing  $E^3SAP$ in the batch mode.

Input requirements for both modes are identical, and only minor differences occur within the job control statements. As a result, users may alternate between the two modes with case

# Extensive Element Library

 $E^3SAP$  maintains a comprehensive, state-of-the-art inventory of finite elements needed to perform most linear structural analysis, including.

- Three-dimensional truss
- Three-dimensional beam
- Two-dimensional membrane



Two-dimensional axisymmetric solid

.hree-dimensional colid

- Thin-plate or thun-shell
- Three-dimensional variable-number-of-nodes solid
- Boundary (or foundation) springs
- Three-dimensional pile
- Three-dimensional pipe (tangent and bend)

# SPECIAL FEATURES

# **User-Defined Motion Features**

 $E^3SAP$  permits the linking of "slave" and "master" degrees of freedom. This allows the realistic and accurate treatment of rigid body portions of a structure such as rigid or very stiff beams, diaphragms, and slabs.

In addition, skewed boundary degrees of freedom may easily be specified using the boundary spring 'es.

# **Diagnostic Aids**

The  $E^3SAP$  program may be run in a data check mode. In this mode, the complete data input stream is checked and error messages are printed to indicate prohable causes of any detected errors

As an additional diagnostic aid,  $E^3SAP$  logs important solution job-step summary information in the user's EKS system day file. A quick examination of this dayfile via an interactive terminal not only indicates whether or not the run is successful, but also what and where the probable cause of the error might be

# Comprehensive Result File

Frequently, there is a need to perform postprocessing analyses using  $E^3SAP$  output To facilitate this need.  $E^3SAP$  features an easy to read result file containing complete geometry information, as well as computed quantities such as displacements, stresses, and mode shapes

# PROGRAM CAPACITIES

Because its memory requirements are completely dynamic, there are no fixed capacity limits for  $E^3SAP$ . Utilizing the maximum computer capacity of MAINSTREAM-EKS, static analysis problems as large as 6000 nodes with a virtually unlimited number of elements or load cases can be solved with  $E^3SAP$ . For dynamic analyses, the maximum problem size is governed by the product of the total number of nodes and the number of frequencies.

# MAINSTREAM-EKS OPERATING ENVIRONMENT

 $E^3SAP$  and MAINSTREAM-EKS offer the flexibility for selecting the operational mode which best satisfies your solution needs. Both processing modes, interactive and batch, offer specific benefits. Interactive solutions provide real-time answers, whereas batch solutions provide results at a minimum of cost.





# PROGRAM SUPPORT

BCS provides in-depth technical support for users of  $E^3SAP$ . This support is provided by personnel with structural engineering degrees who are familiar with both the program code and real-life engineering problems.

# DOCUMENTATION

The following documentation is available through your local BCS Representative

E<sup>3</sup>SAP User's Manual 10205-043

Contact your BCS Representative for further information regarding E<sup>3</sup>SAP.







# STARDYNE\*

# **GENERAL INFORMATION**

The STARDYNE Analysis System consists of a - series of compatible programs for the static and dynamic analysis of structures.

The static capability includes the computation of structural deformations and member loads and stresses caused by an arbitrary set of thermal, nodal applied loads and prescribed displacements.

Using the direct integration or the normal mode techniques, dynamic response analyses can be performed for a wide range of loading conditions, including transient, steady-state harmonic, random and shock spectra excitation types. Dynamic response results can be presented as structural eformations (displacements, velocities or accelerations) and/or internal member loads and stresses.

To aid the user in the interpretation of results, plots of stress contours, deformed and undeformed model geometry are available. Complete time histories of stresses, internal member loads, displacements, velocities and accelerations may also be plotted.

Automated node and element generation capabilities reduce the time and effort required for data input.

# ANALYSIS CAPABILITIES

The STARDYNE system provides the following capabilities:

# Static Analysis

The static analysis is based on the "Stiffness" ("Displacement") method and conforms to small displacement theory.

# Substructure Analysis

An unlimited number of substructures may be individually modeled and subsequently tied together. Hence, there is no limit to the size of a complete structure for static analysis.

MAINSTREAM: EKS

# Lift-off

The static response of structures containing lift-off and bottom-out points, as well as tension only and compression only members, may be solved in a non-iterative exact solution, with the static analysis.

# Extraction of Eigenvalues and Eigenvectors for the Structural System

Procedures include LANCZOS, Householder-QR, Inverse Iteration and H-QR Guyan.

# Complete Seismic Analysis

Earthquake response may be analyzed as a transient response in the time domain or as a random response using Power Spectral Density inputs or utilizing the shock response capability. Many standard earthquakes are built into the program.

# Piping System Analysis

A curved pipe and a straight pipe are available as finite elements and can be subjected to any of the static or dynamic analyses. The stiffness matrix can be formulated using the ASME flexibility factors at the user's option. Piping elements may be combined with any other elements in the STAR-DYNE library.

Dynamic Response Analysis (DYNRE) Programs Program modules for the various dynamic analyses include:



2-149


# DYNRE1-Transient Response Analysis, Modal Method, Time Domain Solution

The stresses, internal loads, and coordinate respon-; are calculated for any time-dependent forces \_pplied to the structure. Base motion response can be computed in either relative or absolute coordinates. Sinusoidal forcing functions may be automatically generated.

# DYNRE2-Steady State Harmonic Response Analysis, Frequency Domain Solution

The coordinate displacement, velocity and acceleration and stress responses are computed for a steady state sinusoidal forcing functions or sinusoidal base motion. Point-to-point transfer functions are calculated for unit sinusoidal excitations.

# DYNRE3-Stationary Random Vibration Analysis

Calculates RMS nodal responses, element forces, stresses and Nodal Response Power Spectral Density (PSD) due to a user supplied PSD input forcing function.

# DYNRE4-Shock Analysis

esses, internal loads and coordinate responses and calculated for either user-supplied shock spectra or standard earthquake spectra internally supplied by the program. The program will do an absolute upper bound or a root sum square or let the user select from a variety of other modal combination methods currently in use by the engineering community.

# DYNRES-Shock Spectra Calculation

Using a DYNRE acceleration time history or one supplied by the user, the program will calculate shock spectra which can be used as input to DYNRE4.

# DYNRE6-Transient Response Analysis, Direct Integration, Time Domain Solution

In addition to the linear elastic structural components, the model may contain non-linear, onedimensional elements to simulate such phenomena as gapping, bottoming out and soil.

# TYPICAL APPLICATIONS

- · Gearbox analysis and design
- · Building frames subjected to seismic loads
- Industrial piping systems
- Nuclear pressure vessels
- Solar collectors
- Offshore drilling platforms
- Electronic component housings
- Heavy equipment design and manufacture

# FEATURES

# Input Data Generation

Input data generation features are available for node points and finite elements on curved or flat planes.

# Graphic Output

Plots of the original model as well as the deformed structural shape and stress contours are available on many different plotting devices. Complete time histories of stress, internal member loads, displacements, velocities and accelerations may also be plotted.

## Load Case Combination

A post processor for complex load case combinations provides searches, output by load case or by element and stresses for "worst direction" of wind or wave loadings.

## Extensive Finite Element Library

STARDYNE provides a comprehensive, state-ofthe-art inventory of finite elements needed to perform most structural analyses including:

- Beam and Pipe elements with shear stiffness in 3-D space
- Two Triangular Plate Elements (thick plate and thin plate)

Plate Bending Sandwich (thick plate only) In-plane (constant strain) Shear Only (thick plate only)

• Quadrilateral Plate Element (isoparametric inplane)

2 10208-138



Revision 1 August, 1991





ings, non-standard elements or substructures y be entered in numerical form by direct alterations to the stiffness matrix.

- Hexahedron (Brick) Solid Element (isoparametric)
- Wedge Solid Element (isoparametric)
- Tetrahedron Solid Element (constant strain)
- Nonlinear Springs

### Calculation of Hydrodynamic Forces

STARDYNE may be used to compute hydrodynamic forces on the tubular and circular beam members contained in the submerged portion of the model. The fluid forces can result from both wave motion and a steady current. The wave motion is defined by z Stoke's 5th order theory.

### Nonlinear Foundation Analysis

The program may be used to determine loads and deformations in a linear elastic structure supported

a nonlinear foundation and subjected to a static loading.

#### Extensive Program Checks

Numerous error, consistency and validity checks are performed throughout the program.

### Automatic Bandwidth Reduction

Nodes are reordered internally so as to produce a minimum bandwidth. This does not effect either input or output of data and can produce a substantial savings in run time and cost.

### Geometry Phase Checks

The user may complete an entire analysis in a single run; however on larger problems, it is advisable to terminate the run after the geometry phase in order to check the run time estimates and to inspect node and element data. Additional validation performed during this phase includes checks for problem size, duplicate or badly shaped elements and data inconsistency.

### PROBLEM SIZE

There is no practical limit to the number of elements which may be used in a single model or substructure nor in the number of nodes used in most dynamic analyses. Limitations on program capacity are imposed only by available memory storage.

# MAINSTREAM-EKS OPERATING ENVIRONMENT

STARDYNE on MAINSTREAM-EKS offers the flexibility for selecting the operational mode which bests satisfies a user's solution needs. Both processing modes, interactive and batch, offer specific benefits. Interactive solutions provide real-time answers, whereas batch solutions provide results at a minimum of cost.

### SUPPORT

#### User Documentation

Manual for the use of STARDYNE on MAINSTREAM-EKS is:

MAINSTREAM-EKS STARDYNE User Information Manual 10208-136

Contact your BCS local representative for this document and for other information on MAINSTREAM-EKS.

#### **Technical Support**

BCS provides in-depth technical support for users of STARDYNE both at National Support Headquarters and in a large number of BCS sales offices throughout the nation. The support staff consists of engineers who have a thorough understanding of STARDYNE and its application to a variety of practical engineering problems. Staff members can help you select an appropriate solution process from several options and assist in debugging and in the interpretation of results.

"YRIGHTE The Booing Company 1980

• • •



2-151

Revision O

# 2.10.2 Finite Element Stress Summary

This section provides results of cask and lid stress analyses using models depicted in Figures 2.6.1-1 and -2. Results were developed using the EBSAP program described in Section 2.10.1.4. A common organization of results is used for both cask and lid analyses. Each analysis consists of two runs for a total of seven loading cases. The two runs were used to reflect both accident and normal temperature distributions, respectively. The seven cases are described as follows:

• Case 1 - Accident (Fire) Thermal Stresses

Decay Heat = 800 watts Stress-free temperature =  $70^{\circ}$ F Other controlling temperatures:

Component	Temperature		
	(Table_3.5.3-1)		
Cavity	371.43°F		
Outer Shell	343.56 <sup>0</sup> F		
Bottom Outer Shell	363.35 <sup>0</sup> F		

o <u>Case 2 - Accident (Fire) Pressure Stresses</u>

Internal Pressure = 184.40 psig (Section 3.5.4)

Case 3 - Accident (30 foot drop) Impact Streases

(Section 2.7.1)  $n_g = 2.704.956 = 100.184g's$ 27,000







0

0

0

<u>Case 4 - Accident (Fire) Combined Temperature &</u> <u>Pressure Stresses</u>

Case 1 + Case 2

<u>Case 5 - Normal Thermal Stresses</u>

Stress Free Temperature =  $70^{\circ}$ F Decay Heat = 800 vatts Ambient Air Temperature =  $130^{\circ}$ F Body Temperature =  $214.4^{\circ}$ F (Table 3.4.2-1)

<u>Case 6 - Accident (Impact) Combined Stresses</u>

Body Temperature = 214°F Internal Pressure= 19 psig Axial Load = 100.184g Case 6 = Case 5 + (19/184.4)° Case 2 + Case 3

<u>Case 7 - Normal Combined Temperatures and Pressures</u>

Body Temperature = 214.4°F Internal Pressure = 19 psig



## 2.10.2.1 Cask Stresses

Cask stresses are provided in Microfiche form in Section 2.12. These data include complete results of finite element stress analyses of cask and lid for all cases described in Section 2.10.1. The information is contained on the microfiche entitled "2.10.3 Finite Element Stress Analysis 1-13C II Cask, and on the Microfiche entitled "Appendix 2.10.4 Lift Lug Analysis, August 1981.









•

.



	JO BASE HD	DE LOAD				
QUAD-PLATE C	ORNER FORCES F	OR CASE 1				
HODE		FORCES			MOMENTS	
	FX1	FX2	FYN	HY 1	HY2	HX3
HAQUAD-PLATE ND	1 44		123	110.4	1146	1
GLOBAL J1 8	.1504435483	- 1495185484	2228/25184	- 4974875187	1537738187	1998785184
12 10	- 4194545101	1375100704	.2227625.04	42/402E+02	.0732/36.42	- 1// 7775106
	7/00145403		./723486103	.0/3841E+02	4961516103	-,1002//2/04
	.2070176703	·147512F+04	298197E+04	.582780E+02	278100E+03	2004/12:09 12
Ja I	.208128E-08	308117E+04	.150030E-07	521726E+02	.261934E-09	.2743942+04 18
<b>HRQUAD-PLATE HO</b>	5 H H					
GLOBAL JI 9	.456252E+#3	778929E+#3	.287237E+83	.112274F+03	~.271#53E#83	.139334E+04 (答
J2 10	101720E+04	211801E+04	- 180285E+04	- 911765F+02	2254115+01	- 346288E103
<b>Z Z</b>	781094F+03	1076616407	- 4141475183	17/07/050-08	- 312#116-08	- 102061F+01 X
14 7	- 2201445103	- 11/01010402	~. 4363472403	.1/60/02-00		2212255406 0
WEATTE NO	2201962703	-11112256484	·503130E+84	1400135401		
CLUBAL JI IN	3/9390E+02	•221213E+#3	601418E+02	.98D495E+#1	-,127781E+02	.00187/6102 0
JZ 11	624792E+02	336476E+03	521904E+03	1744598+03	.227359E+03	.465548E+D3 W
	.901532E+#3	204771E+03	.145697E+03	.168882E-08	337604E-08	.819648E+D3
J4 ° 3	7810946+#3	102661E+02	.436347E+03	276486E-08	.465661E-08	.302041E+03
**QUAD-PLATE NO	4 88 -					l⊶
GLOBAL JI II	877742F+#7	.4238136483	- 5691975183	- 1855738483	810688F402	497301E+03 []]
12 12	4898865403	- 4544405181	3335845403	- 7878645403	2282575483	A14899E+D3
	49447164#1			- 35/3805-09		1180625404
		.2007835792	.2022105403	/J6/00E-08	.2783140-08	- #10/475401 0
	JUIJJZE+UJ	.204//16+03	145699E+93	2684/9E-08	·2149026-09	
RAQUAD-FLATE HU	2 44					2
OLOBAL JI IZ	321902E+U3	~34682IE+83	133345E+#3	211008E+03	.874023E+02	103042E+04
N 72 13	.6625600+03	399780E+03	.376144E+03	364750E+03	.151084E+03	.1028962+04 1-
년 <b>J</b> 3 6	.153763E+#3	.790057E+02	.1405J2E+03	.183718E-08	851237E-09	.126163E+04
5. J4 5	494421E+#3	260763E+02	383310E+03	749424E-08	4214605-08	118062E+04
W. RRQUAD-PLATE ND	6				• • • • • • • •	• Iv
GLOBAL JI 13		.1755435+83	.1898845+82	3117665483	.409789E+D2	113945E+04
12 14	6231235101	- 9651755187	1214035403	- 1718555101	4895575102	1128705+04
	4 18 18 75487	- 7873075-04	. 1210032103		- 3414345-11	1781285+04
						- 1761615406
	122/076463	/yuus/E+u2	190212E+03	.1919266-07	· 7996916-01	1201032104
HRQUAD-PLATE NU	/ **					
GLOBAL JI 64	9928692199	1229565442	.268265E*84	.464727E+#3	-,2255686+04	.2939702009
JZ 16	.428032E+04	127822E+05	J484J4E+DZ	377636€+03	.183216E+04	.3424345403
4 EL	.135683E+04	311521E+04	211225E+#4	.984254E+02	4775276+03	1144821+01
J4 90	117895E+ <b>#</b> 4	.451486E+94	455554E+03	103839E+03	.503794E+03	.183956E+04
<b>XXQUAD-PLATE HO</b>	8 N.N					
GLOBAL JI 16	344635E+#4	· .697641E+04	.162393F+D4	.5#7776E+D3	122588E+#4	.2004798+04
12 17	2239195404	416113F+84	- 401025F+84	- 177905F+#1	912342E+03	127225E+04
11 10	1635605406	- 2716895484	- 7118155481	4117655482	- 2254115+03	968240E+01
	- 4384475187	- 7838055107		- 1122745481	7710535403	1195445106
		/030936/02	.2100105+04	1122/46403	. 2/14236.42	.13/3402.04
REQUAD-PLATE HU	7 **					
_GLOBAL JI 17	141380E+04	.1443726+44	-11221126+04	.5318486403		.1212236+04
JZ 18	.110229E+84	117323E+04	175293E+#4	312475E+03	.407225E+03	.1151026.04
JZ ]]	.872043E+#3	338060E+03	.112913E+#4	.174459E+03	227359E+03	.566417E+03
J4 10	560444E+83	.673642E+92	.257683£+04	980495E+01	.127781E+02	,275780[+03
WHQUAD-PLATE HO	18 XM					
GLOBAL JI IN	- 3356248+13	1383616483	7791175+81	.7786185+83	- 597449E+03	.963141E+03
17 14	1711755141		478433516743	_ 1708766181	2511225401	1210625104
JE 17 17 17		JUCCJDETUJ	. 72071JE VJ	JE 70/JETTJ		8351385401
13 15	.370/376783	.1010/96.03	.300,305,43	.20/0112/03	~,22023/6703	
J5 _11	~./26/90E+03	.2207246+03	3803428402	.102213E+03	-,810038E+02	
- REQUAD-PLATE HO	11 AM					
GLOBAL JI 19	.232050E+#3	374880E+83	.97588558482	.742551E+#3	-,3158598+03	.486924E+03
JZ 20	766720E+80	.123674E+03	.587456E+03	198716E+03	.823109E+02	,102925E104
72 · 72	.327439E+03	.144390E+03	983377E+02	.364750E+03	151864E+03	.103189E+04



				C7 #1			
	. 12	358722E+#3	. 1258365483	- 534494	2118885481	- 8740315107	- 1996765101
WNOU.	LATE NO	12 88			511880C*83	0/10236102	
CI 084'	1 28	515182F+81					
	1 21	- 4873465103	5600385107		. 4002136143		0/00232102
	13 14	1489865101	6/51755/02	.23013/2.3	10//906403	1418436102	1011005104
	14 11	- 1067615161	.7633/3E+02	121603E103	.3/1855E+03	- 489557E102	.1031902004
wweitte	PLATE NO		.//89688+02	296715E+03	.311266E+03	4097896+02	4513446103
CLOBAL	- CLAIC RU			<b></b>			~~~~~~
OLOBAL	17 74			.718795E+#4	716967E+#2	.173071E+03	2//1102+03
	JZ 24	2553806+04	.462336E+04	110214E+05	.697454E+02	168380E+03	6645911+03
	JJ 1/		106193E+05	818503E+04	.833384E+02	201197E+03	.61/3251+03
WYALLA			382293E+03	.120185E+05	108126E+03	.2610390403	.1068892+04
	TLAIE HU	12 ##					
ULUDAL	JI 21	4946106+04	.573172E+44	.108522E+05	.4680758+01	610033E+01	875:138+02
	12 25		.505106E+04	110309E+05	.129802E+03	169161E+03	405556[+03
	12 18	-601565E+04	679319E+04	113183C+05	.137355E+03	179005E+03	.730270E+03
	J9 17	.248406E+04	418959E+04	.115470E+05	.759469E+#1	989759E101	.649211E+03
REQUAD	-PLATE HO	16 XX					
GLOBAL	JI 25	3335558484	.2932578+04	.]#6223E+85	.886462E+#2	689207E+02	156510E+03
	JZ 26	682521E+04	.4758558+04	974826E+84	.143650E+D3	110227E+03	2440826103
•	JJ 19	.492402E+04	SD9171E+04	108248E+05	153258E+03	117599E+03	.7931792+03
	J4 18	.523675E+04	461741E+04	.795082E+04	.450215E+02	345462E+02	.425868[+03
MMQUAD	-PLATE HO	17 XK					
GLUBAL	JI 26	4402798+03	.4#8372E+#3	.7456778+84	. 9729182+02	482996E102	184271E+03
	JZ 27	776374E+#4	.292269E+04	4 48 326 E+04	1 1 + 03	564005E+02	104837E+03
	J3 20	.204125E+04	431335E+03	737783E+04	11.458E+03	720972E+02	.7352362+03
	J4 19	.616277E+04	- 289974E+04	482433E+04	509161E+02	210985E+02	.108432E+03
NNQUAD	-PLATE HO	18 #4	• • • • • • • • •		19079032002		
GLOBAL	J1 27	338797E+14	391181E+03	.243882F+#4	.1056525+83	139893F+82	130227E+03
	J2 28	652446E+#4	.739910F+03	- 2716965+04	1223015403	- 1610125+02	242393E+02
N	J3 21	170875E+#4	.383203E+03	2368835484	.164734E+#3	216877F+02	.543841E+03
ï.	J4 21	484324E+04	731932E+03	2646965104	118071F+03	144911E+02	245132E+03
WHQUAD	-PLATE HO	19 XX					•
O GLOBAL	JI 91	.896482E+#3	256518E+84	.1550995+#3	19478#E+03	.9450898403	128387E+04
	JZ 30	- 127891E+04	468379F+04	- 7484825+84	1542785483	- 748505F+03	533344E+03
	JJ 23	309236E+84	.8003702+04	101205F+84	- 466149F183	226160F104	1163010+04
	J4 65	346679E+04	- 101223F+05	826177F+#4	1995826401	- 193864F+04	189256E+04
MMQUAD	-PLATE NO	20 XN	1101000000				
GLOBAL	J1 30	125348E+02	_110448E+04	.5834685403	1495495183	.341043F+03	
	JZ 31	147458E+04	260285F+84	- 481444F+04	1475235+03	356151F+03	173289E+03
	J3 24	757034E+03	897685E+13	200937E+04	432473F+03	105508F+05	1436860+04
	J4 23	224414E+04	441493E+84	636841F+04	412785F+03	996550F+03	127764E+04
MMQUAD	-PLATE HO	21 44				••••••••••••	
GLOBAL	JI 31	1797868+03	.873403E+83	.2738788+84	372389F+82	.420041E+02	319276E+#2
	J2 32	118250E+04	856413E+03	703937F+#3	1551475403	2021910+03	360197E103
	JJ 25	.323304E+03	205951E+#3	318568E184	13930AF+03	.181550E103	685911F+03
	JA 24	.103898E+04	154387F+04	1150545404	- 987471E402	- 128690F+03	335875E+03
WWQUAD	-PLATE HO	• 22 #4			1707 1712:02	110070000	
GLOBAL	J1 32	2509678+83	.1997465+83	2308245484	9689465182	728158F#02	.2299665403
	12 33	- 120298F+04	395274F+#3	1162645403	2780028403	- 1749525+03	- 5410415103
	JJ 26	952208E+#3	957463F+#3	- 207641F+06	10424F4#3	- 8473145102	- 4111655401
	JA 25	1904225+04	- 15524XF+04	- 1470415101	. 3085796+63	236781F+D3	- 4421085+01
XXQUAD	-PLATE NO	23 88				************	
GLOBAL	J1 11	983929F+81	722638F+83	. ********	2858825183		444289FID1
<b>U</b> U U U U U U	12 16	157400F+84	2761765401	- 4751865481	1167715101	- 130176F+03	6887795401
	JJ 27	717422E+04	1418615404	- 5169695103	1729616461	- 1546855103	203555F+03
	JA 26	356427F+84	- 1472165104	152441Fint	5251115461	- 2175175101	4187041403
- XXQUAN	-PLATE HO	24 KK					
GIONAL	JI 34	1580235484	- 7211655483	- 4186835103	1814175181	- 3994825187	.636416E+D3
	12 35	- 178420F104	1206205405	- 8646465403	1687786101	- 4501645107	- 7795551101
	11 28	- 4447595104	7761666142	5513110105703	.JTU//VETVJ SL0777EINY	- 7581726187	3561075107
		* * * * * 2 2 5 5 * 8 7	***********	・ フ フ コ ビ フ イ ビ マ V ブ		** ? * * C * * *	*******

71	:	-

¥•	J4 27	465156E+14	335049E+#3	.727 483	.624769E+#3	822525E+02	258268E4
<u></u>	11 11					<b></b>	
010	21 28	.10/0302+12	.218087E+#3	-,486/00E+04	895286E+#2	.514685E+03	233648E+04
	JZ 3/	//39//2+02	714337E+03	3089938+04	.165509E+02	.985136E+01	.19156JE+04
	77 41	.120517E+03	138432E+04	.618225E+04	.627990E+8Z	371654E+03	.143866E+04
	J4 29	599024E+02	-158057F+84	1775485+04	1461655102	259196E+33	2093746+04
N×QU/	ND-PLATE H	0 26 ##					•••
GLOBAL	JI 37	- 124128F+#1	2186565484	- 4114575141		7871515183	- 1869965104
•••••	.12 38	- 9777116441	1116565184		2707236743		4117556401
	11 11			-, 3763736707		1406776443	200/01/00
	14 10		2034576404	/8843/2403	14/5232.03	. 3201215103	.2//////////
wwnit i	07 JU		12833/E+84	.6685822+04	.1495496483	2010435+03	1024696404
	AD-FLAIC A	U 2/ 44					
ULUBAL	L 11 79	230014E+03	.125402E+04	,153834E+84	262171E+03	.341668E+D3	37/1/6[103
	JZ 39	127602E+84	.7626448+05	233360E+04	.211104E+03	275116E+03	3944948103
		.766682E+03	554981E+03	2080842+04	155147E+03	.2021918+03	·291281E+03
	J4 31	.739350E+03	146168E+04	2884105+04	1223095402	420041E+02	942738E+02
XXQU)	ID-PLATE H	D 28 HN					
61 0 8 A 1	J1 11	- 5848888482	<i></i>	17/0155104		<i></i>	1945845481
	17 4	- 7156975441	- 2882105107	- 5403335101			- 101017E404
•	11 11		2002170102		.2/79362903	2144202003	_ / # ] / # ] [ + 0 ]
	JJ 33		1166146443	115/18E+04	228002E+03	1/4952E+U3	
-	J9	.1698986+#3	501178E+03	.476540E+03	948944E+02	.728150£+02	.4213466+03
XXV/	AD-PLATE H	0 29 88					
GLOBAI	L JI 40	769998E+#Z	.374712E+#3	.204044E+03	.721197E+#2	298730E+#Z	.103249E+04
	JZ 41	144607E+03	261269E+03	5373672+03	.286377E+03	118621E+03	141311E+04
	J3 34	.611815E+03	574142E+02	.102101E+03	314273E+03	1301760+03	971273E+03
	J4 33	- 390208E+03	568294E+02	231222F+03	- 205002E+03	349145E+02	7783550+03
HHQU/	D-PLATE H	D 30 KH			12030022.00		
AL CRAA	11 41	- 137525F483	1392795483	- 7787765483	1816955483	- 23928685402	.1414926+04
	17 47	1848975483	- 104/155101	- B24144F141	-1010735743	- 1114745107	- 1568385104
	11 18		- 10/01/01/07		.2333042483		- 1001716104
			3206806402	.8696966783	348//02+03		
- L	J7 J7		.2403066+01	·/83/885.03	707471F+07	.3994826102	
REQU/	AD-PLAIE H	0 31 44		•			
SLUBA	L 11 43	.846034E+0Z	.11#820E+#4	-,231892E+#4	916416E+0Z	.696087E+#3	322706E+04
	JZ 44	255527E+#3	.342990E+03	418138E+#4	375743E+02	.285405E+03	.260323E+04
	J3 37	.1877#8E+03	146298E+04	_163250E+#4,		303673E+03	.220919E+04
	J4 ·36	167835E+#2	.197916E+0Z	.486780E+84	.677596E+12	514685E+ <b>03</b>	264396E+04
x x q U /	ID-PLATE H	D 32 XX					•
GLOBAL	L J1 44	.605762E+02	.114595E+04	564717E+03	294364E+03	.710658E+03	260980E+04
		- 755698E+13	.531668E+P3	363923E+04	577238F+82	139356E+03	.1211708+04
	J3 38	440723E+83	- 125362E+84	4540375+03	- 4159255+02	148697F403	.836466F+03
	.14 . 17	756399F+83	- 4219985101	3749916484	2989665401	- 7217205403	- 2252051104
<b>VX611</b>	10-PLATE 4				.2707402403		
C1 (1 1 1 1			********	1178665181		4947365481	- 1218865104
GLOBAL			.1115302404	.3120335403	3/17316783	.404/302/03	- 1210012104
	JZ 10	-12/834E444		-,3042962+04	.281434E+03	366//2E+03	350416E+03
	72 22	.544018E+#3	580175E+03	750146E+03	211104E+03	.275116E+03	395214E+03
	J4 38	.766521E+#3	113999E+#4	.348105E+04	.262171E+03	341668E+03	850543E+03
X Y Q U A	ID-PLATE H	0 · 34 ××					-
CLOBAL	. JI 46	.127224E+#2	.735991E+03	.559317E+#3	~.2#1153E+03	.154350E+0'	.341181E+03
	J2 \ 47	109747E+84	.817915E+02	172769E+04	473507E+#3	363335E+03	156034E+04
	J3 40	294335E+03	.112999E+#2	- 455024F+#3	- 2794 \XE40 \	214420E+03	- 1232661104
	.14 . 39	7904#8F+#3	- 829082F481	1823405404	868665E102	- 466550F+02	.393121E+03
K×6114	D-PLATE N	0 15 88					
01 0141	11 47	7781875187	<b>3886765183</b>	- 1715885191		- 3637875407	1552045104
ULUBAL			_ 338831E144	- }}}//////////////////////////////////	+0/007DC7#4	- JUJEVELIVE	
	JZ 10		««00/1E*UJ	113/406404	.31330/6403	2237302403	- 177/4/544
	13 11	2323136+02	11112256403	.2602/26.03	~.2863//EI03	.1100216103	
<b></b>	J4 40	. 498362E+03	321140F+83	.1000692+04	721197E+0Z	·548130F+05	.1230342404
	AD-PLATE H	D 26 xx					<b></b>
GLOBAI	L J1 48	.701119E+0Z	.1#9671E+#3	-,835059E+03	.348628E+83	-:448445E+02	.227843E+04
	J2 49	167923E+#3	143890E+03	109045E+04	.497389E+03	6548258+02	252054E+04
	J3 42	207573E+03	.109615E+03	.928188E+03	253564E+03	.333824E+02	189419E+04

Revision O

.

2-157

•



.



			1-4	N		
	1 1853848285			<u>[]</u>	01000/F100	1774655164
	но 37 жж	-1123323F+#5	. <b>77</b> /321E	181693E+#3	.2392082+02	.1/29036149
GLOBAL 5	0 .153730E+02	.1726848+04	155994Ev	132928E##3	.188969E+04	412533E+04
. Je 5	1 .492765E+#1	128746E+04	236452E+04	865436E+#Z	.657364E+03	.347176E+04
J3 4	4 .622828E+02	128389E+04	.168554E+04	.375743E+#2	285405E103	.290578[+04
	3846034E+82	.844515E+03	.231892E+04	.916416E+#2	696087E+03	~.357658E104
		.1/15/46404	7079548403	409910E+03	.9896102403	
	5	- 310079102403	2733938+04		110141401403	1446615404
JĂ Ă	4 .13266AE+03	2#504\F+0\	3140565484	)//2306402	710658E+03	2899210+04
HHQUAD-PLATE	HO 39 HR			.2/13012.13		
GLOSAL JI 5	2 .110292E+02	.120117E+04	191158E+#3	469599E+#3	.611994E+#3	181533E+04
JZ 5	3 903651E+03	.578505E+01	272216E+04	.2382785+#3	310531E+03	241477 <u>L</u> +03
J3 4	6 .440567E+03	660780E+03	.894674E+02	281434E+#3	.366772[+03	303684E+03
A 4	5 .452054E+03	546179E+03	.282385E+04	.371951E+03	484736E+03	1439488404
RAQUAD-PLAIE				•		2285125485
GLUBAL JL J		,6185548493	2117826+03	2210478+03	.1676132+03	- 2059545+06
JZ J 4 JT 4	7 1777165463	.1235392403	-,2270/92909		900779270J 161115F#81	-174453E104
A AL	6 .825045F+83	- KX0076F403	2194185404	4/330/2703 201151F401	- 1543505+03	312919E+03
- WHQUAD-PLATE	HD 41 XX			.2011/32.03		
GLOBAL J1 5	4 .751564E+#3	.194439E+03	576473E+03	.23225E+#3	961908E+02	.204719E+04
J2 5	5138794E+84	.684503E+01	175462E+04	.863688E+03	357751E+03	J24149E+04
JZ 4	8216854E+03	.205071E+03	.588255E+03	545507E+03	.225956E+03	263381E+04
J4 4	7 .853230E+03	406355E+03	.174284E+04	876848E+02	.363202E+02	.175313E+04
WANNAD-PLATE	HD 42 MM					
			1064586+04	.6690912403	868708E*V2	- 1460795104
N JC J	-502400F403		1410146404	- 497189ELD1	4548255402	291127E+04
	8 .6938728483	858719F+82	.1334275404	140%2%E+D1	.448445E402	.263823E+04
- ** ** QUAD-PLATE	HO 43 ##		11001010101			•
OLOBAL JI 5	7 .139698E-08	.7#5843E+#3	148250E+04	.100772E-08	424914E-08	447888E+04
JZ 5	8197378E+03	687300E+03	153922E+04	152795E-08	.9429648-08	.383418E+04
12 2	1 .212771E+03	142548E+04	.146178E+04	.865436E+02	657364E+03	.3/15526+04
		.1986792489	.1009946+04	.132928E+03	100363E+04	
		*********	- 1475785184	- 1484185-88	883766F-08	383418F+04
.12 5	9 - 365600F+03	641640F+03	1484145484	475209F-08	.1152518-07	.211334E+04
ž žL	2 .311030E+#3	104286E+14	.141723E+#4	493529E482	119148E+03	.198452E+04
. JĀ 5	1 142808E+03	.997202E+03	.161269E+04	.409910E+03	989610E+03	371112E+04
HHQUAD-PLATE	HO 45 ××					
OLOBAL JI 5	9 .365600E+03	.641640E+03	136886E+84	.250293E-08	75670DE-09	211334E404
JZ	•520682E+13	311280E+03	164674E+04	13#385E-#7	.137952E-07	133158E+03
13 2		3819391+33	.1299746404	2382/81403	.3103316+03	- 1076875464
	2 ·	.231377.03	.1/0/052709	.4693996+03	0117742703	17/40/5/44
	8 .5206825483	.311280F+83	1318261484	*****	8672945-88	.133158F+03
	1 629403E+03	357084F+03	160631E+04	.1868475-07	104774E-07	224477E+04
ĴĴ Š	4 373651E+#3	.88180.5E+02	.129037E+04	608598E+03	466994E+03	217626E+04
Ĵ4 Š	3 .482373E+#3	24238LE+03	.163420E+04	.221847E+83	169615E+03	.201064E+03
#¥QUAD-PLATE	HO 47 ××	-				
GLOBAL JI 6	1 .629403E+#3	.3571B0E+13	135869E+#4	222062E-87	.127693E-07	.224477E+04
JZ 6	Z672134E+03	183517E+03	155136E+04	483124E-18	.8440118-09	3/0114E+04
13 2		.2346902403	.1333202404	8636886493	.35//516+03	3202332104
- VNAILED-PLATE		4082336+03	.13/6635.09	2322236+03	. 101 1005 02	
		1835175183	- 1413245484	1671455-88	- 276464F-09	378114Fin4
J7 4	3 709583F+#1	.861473F-DR	148250F+04	117325F-09	144382E-10	421385E+04
2 2 2	6 121024E+04	562604E+02	.141014E+04	889747E103	.117137E+03	39662/1:04

the second particular

**1** 

t.



13

181

.849334E+04

-.341177E+01



.620938E+04

APPENDIX 2.10 ... E LUG ANALISIS cont n.

Revision

Ċ,

-.582712E+03

.140679E+04

.6154120+04



	•	80	151018E+05	.3310632+04	634038E+.	.4004878403	165887E+03	.102366E+04
**QUAD	-tial	TE HO	67 HH			•		
GLOBAL	J1	88	132043E+04	.627872E+03	204827E+04	.866824E+#3	114119E+#3	113567E+04
	J2	39	.379465E+04	146406E+#4	2163725104	\$30733F403	- 122531E+03	.274971E+03
	73	82	.133479E+05	437297E+03	.2138692404	.133154E+04	175301E+03	.450470E+04
	_J4	81	158222E+05	.127348E+04	- 2254145104	884928E+03	- 114503E+03	196476E+04
MAGAYD	-PLA1	E HO	68 XX			10017202.05	11109032005	••••
GLOBAL	JI	15	539953E+03	.562827E+04	.178844E+03	.361634E+#2	478841E+03	.133386E+04
	JZ	64	.525487E+03	636781E+04	2592715+04	1547448+02	213455E+03	980046E+03
	73	90	.164908E+03	176926E+04	541928E+03	188155E+#2	.259542E+03	1550810104
	J4	8	150443E+03	-250880E+04	- 277967E+84	473590F101	- 651271F+02	.147240E+04
**QUAD	-PLAT	TE NO	69 XX		1000000		10992192102	••••••
GLOBAL	JI	29	.599024E+82	.183294E+04	177548E+04	187904E4#2	.259196E+03	153978E+04
•	JZ	91	160479E+03	323505E+02	798228E+04	743951F+01	.103311E+03	.107067E+04
	<b>J</b> 3	65	417001E+03	.371233E+04	227280E104	375423E102	517860E+03	.353435E+03
	J4	22	.717577E+03	571291E+04	.748497E+04	450269E+02	-,621105E+03	124416E+04

•

•

•

•

٠

•

٠

•





30 E NODE	τανο			
Huues	QUADRI ELEMENT FORCES	LATERAL STK ELEMENT STRESSES	SSES AND FORCES FOR O PRINCIPAL STRE MAX MIN	UTPUT VECTOR 1 SSES MAX SHEAR AHGLE
X2. 1 . /QUAD / 2	FX = 5.624E+02 FY = -8.057E+02 FXY = -3.825E+02 HX = -5.521E+02	+ZFACE SX = -1.225E+04 SY = -1.556E+03 SXY= -3.218E+03	+ZFACE -6.615E+02 -1.314E+04 VM	6.239E+03 ~74.474 1.282E+04
8×1	MY = 2.324E+00 HXY = -1.022E+02	SY = -1.667E+03 SXY= 1.688E+03	-ZFACE 1.467E+04 -1.842E+03 VH	8.236E+03 5.901 1.567E+04
XZ. 2 /QUAD / 3	FX = 2.688E+92 FY = 3.217E+02 FXY = -2.927E+02 MX = -2.781E+02	+ZFACE 5X = -6.136E+03 SY = 5.119E+02 SXY= -4.835E+03 -ZFACE SX = 7.211E+03	+ZFACE 3.056E+03 -8.680E+03 VM	5.868E+03 -62.253 ].054E+04
9XI	HY = -5.480E+00 HXY = -1.771E+02	SY = 7.749E+02 SXY= 3.465E+03	-ZFACE 8.870E+03 -8.839E+02 VH	4.877E403 24.355 9.343E403
X2. 3 /QUAD / 4	FX = -1.229E+02 FY = 1.173E+02 FXY = -4.918E+01 MX = 5.902E+01	+ZFACE 5X = 1.171E+03 5Y = 8.891E+02 5XY= -4.769E+03 -ZFACE 5X = -1.662E+03	+ZFACE 5.801E+03 -3.742E+03 VH	4.771E+03 -44.155 8.328E+03
10:X1	HY = 2.727E+01 HXY = -1.946E+02	SY = -4.197E+02 SXY* 4.573E+03	-ZFACE 3.574E+03 -5.656E+03 VM	4.615E+03 48.869 8.060E+03
X2. 4 N · · /9UAD / 3	FX = -1.521E+02 FY = 4.791E+01 FXY = 9.345E+01 MX = 2.309E+02	+ZFACE 5X = 5.237E+83 SY = 1.293E+03 SXY= -3.323L+83 -ZFACE 5X = -5.845E+83	+ZFACE 7.129E+03 -5.993E+02 VM	3.864E+#3 -29.659 7.467E+03
17:×1	MY = 4.989E+01 MXY = -1.463E+02	SY = -1.101E+03 SXY= 3.697E+03	-ZFACE 9.192E+02 -7.866E+83 VM	4.393E+03 61.341 8.364E+03
XZ. 3 /QUAD / 6	FX = -1.146E+02FY = -4.895E+01FXY = 6.754E+01MX = 2.942E+02	+ZFACE SX = 6.832E+03 SY = 1.410E+03 SXY= -1.865E+03 -ZFACE SX = -7.291E+03	+ZFACE 7.412E+03 8.304E+02 VH	3.291E+03 -17.261 7.033E+03
12X1	MY = 6.282E+01 MXY = -8.333E+01	SY = -1.606E+#3 SXY= 2.135E+#3	-ZFACE -8.931E+02 -8.003E+03 VM	3.555E+03 71.345 7.596E+03
XZ. 6 /QUAD / 7	FX = -9.846E+01 FY = -2.833E+01 FXY = 1.590E+01 MX = 3.144E+02	+ZFACE SX = 7.350E+03 SY = 1.610E+03 SXY= -6.070E+02 -ZFACE SX = -7.744E+03	+ZFACE 7.413E+03 1.547E+03 VM	2.933E+03 -5.971 6.774E+03
13X1	HY = 6.946E+01 HXY = -2.662E+01	SY = -1.724E+03 SXY= 6.706E+02	-ZFACE -1.650E+03 -7.817E+03 VM	3.084E+03 83.720 7.137E+03
XZ. 98 /QUAD / 9.	FX = -2.444E403 FY = -1.155E403 FXY = -2.807E402 HX = -3.494E402	+ZFACE SX = -1.327E+04 SY = -4.789E+03 SXY= -3.150E+03 -ZFACE SX = 3.506E+03	+ZFACE -3.747E+03 -1.431E+04 VM	5.281E+03 -71.690 1.285E+04
64X1	MY = -1.033E+02 MXY = -1.079E+02	SY = 1.6A3E+02 SXY= 2.028E+03	-ZFACE 4.463E+03 -7.884E+02 VM	2.626E+03 25.272 4.905E+03
X2. 9 /QUAD / 10	FX = -1.027E+03 FY = 4.811E+02 FXY = -6.175E+02 HX = -1.397E+02	+ZFACE SX = -5.407E+03 SY = 2.170E+01 SXY= -5.480E+03 -ZFACE SX = 1.299E+03	+ZFACE 3.423E+03 -8.808E+03 VM	6.115E+03 -58.175 1.093E+04





•

			<b>I</b>				
16	17 MY X1 MXY	= -3.919E+#1 = -1.769E+#2	5Y = 1.903 5XY= 3.010L .	-ZFACE 4.62	6E+83 -1.424E+83 VH	3.025E+03 5.479L+03	47.
X2. 10	FX FY 11 FXY	= -3.138E+02 = 7.472E402 = -8.154E+01	+ZFACE 5X = -2.678E+02 5Y = 1.278E+03 5XY= -4.965E+03	+ZFACE 5.52	9E+03 -4.519E+03 VII	5.024E+03 8.717E+03	-49.423
17	18 HY X1 HXY	= -9.028E+00 = -2.001E+02	-2FACE SX = -9.875E+02 SY = 1.711E+03 SXY= 4.638E+03	-ZFACE 5.19	2E+03 -4.469E+03 VM	4.831E103 8.375E103	53.110
X2. 11	FX FY 12 FXY	= -1.198E+02 = 7.065E+01 = 1.068E+02	+ZFACE SX = 2.078E+03 SY = -2.983E+01 SXY= -3.794E+03	VZFACE 4.96	2E+03 -2.913E+03 VM	3.938E+03 6.896E+03	-37.238
18	19 HY X1 HXY	= 7.131E+#0 = -7.131E+#0 = -1.670E+02	-2FACE SX = -2.557E+03 SY = 3.124E+02 SXY= 4.221E+03	-ZFACE 3.33	6E+03 -5.381E+03 VH	4.458E+03 .803E+03	54.386
XZ. 12	FX FY 13 FXY	= -2.604E+81 = -1.381E+82 = 6.394E+91	+ZFACE SX = 3.515E+03 SY = -2.450E+02 SXY= -2.402E+03	+ZFACE 4.68	5E+83 -1.415E+83 VM	3.050E+03 5.530E+03	-23.974
. 11/	20 HX X1 HXY	= 1.486E+02 = 1.391E+00 '= -1.854E+82	-ZFACE SX = -3.619E+03 SY = -3.874E+02 SXY= 2.657E+03	-ZFACE 1.16	8E+03 -5.894E+03 VM	3.131E+#3 5.767E+03	60.963
XZ. 13	FX FY 14 FXY	= 2.050E+01 = -8.434E+01 = 1.419E+01	+ZFACE SX = 4.228E+03 SY = 4.403E+01 SXY= -8.269E+02	+ZFACE 4.38	6E+83 -1.135E+02 Vri	2.230E+43 4.443E+03	-10.783
· / 12/ · / /	21 HY X1 HY	= 1.745E+02 = 8.863E+00 = -3.564E+01	-2FACE SX = -4.146E+03 SY = -3.814E+02 SXY= 8.836E+02	-ZFACE -1.84	3E+02 -4.343E+03 VM	2.079E+03 4.254E+03	77.427
×2. 16	FX FY 17 FXY	<pre>x -8.190E+82 = 7.729E+82 = -2.491E+93</pre>	+ZFACE SX = -2.897E+83 SY = 1.401E+03 SXY= -8.858E+03	+ZFACE 7.59	2E+03 ~9.088E+03 VM	8.348E+83 1.446E+04	-52.466
23	24 HY X1 HXY	= -5.244E+01 = -6.016E+00 = -1.357E+02	-ZFACE SX = -3.795E+02 SY = 1.490E+03 SXY= -1.545E+03	-ZFACE 2.31	3E+03 -1.204E+83 VM	1.837E+03 •3.287E+03	-61.908
X2. 17	FX FY 18 FXY	= -2.916E+02 = 4.611E+01 = -2.800E+03	+ZFACE SX = -9.386E+82 SY = 1.338E+02 SXY= -8.597E+03	+ZFACE 8.21	1E+03 -9.016E+03 VM	8.614E+03 1.493E+04	-46.783
. / 15/	25 HY X1 HXY	= -1.481E401 = 1.733E400 = -1.249E402	-ZFACE SX = -2.279E+02 SY = 5.062E+01 SXY= -2.602E+03	-ZFACE 2.51	8E+03 -2.695E+03 VM	2.606E403 4.515E+03	-46.531
X2. 18	FX FY 19 FXY	= 3.154E402 = -1.762E+02 = -2.572E+03	+ZFACE SX = 1.050E+03 SY = -4.562E+02 SXY= -7.623E+03	+ZFACE 7.95	7E+03 ~7.363E+03 VH	7.660E+#3 1.327E+04	-42.178
. / 16/	26 ny x1 nxy	= 1.748E+01 = -4.322E+00 = -1.033E+02	-ZFACE SX = 2.111E+02 SY = -2.487E+02 SXY= -2.664E+03	-ZFACE 2.63	5E+03 -2.693E+03 VM	2.674E+03 4.631E+03	-42.534
XZ. 19	FX FY 28 FXY	= 7.800E402 = -1.116E402 = -1.785E403	#ZFACE         SX         =         2.619E+03           SY         =         -2.453E+02           SXY=         -5.231E+03	+ZFACE 4.61	1E+#3 -4.237E+03 VM	5.424E+D3 9.469E+D3	-37.344
./ 17/	27 HX X1 HXY	$= 4.414E+01 \\ = -9.231E-01 \\ = -6.921E+01$	-ZFACE SX = 5.007E+02 SY = -2.010E+02 SXY= -1.909E+03	-ZFACE 2.09	1E+03 -1.791E+03	1.941E+03 3.365E+03	-39.793
X2. 21	FX	= 1.039E+03	+ZFACE 5X = 3.493E+03				

Revision 0

•

				-				
	.AD / 21 18/ 27	FY = 5.608E+01 FXY = -6.357E+02 HX = 5.896E+01 HY = 4.764E+00	-ZFACE	SY = 2.26 SXY= -1.85 SX = 6.62 SY = -2.17b_ J	+ZFACE -ZFACE	4.333E+03 -6.143E+0; V? 1.091E+03 -4.310E+02	2.474E483 1 4.671E403 2 7.610E482	-2'
	X2. 65	FX = 1.530E+03	+ZFACE	SXY= -6.846E+02 SX = 6.572E+03		۲۷ ا	1.359E+03	
	/RUAD / 23	FXY = -9.675E+02 HX = 1.464E+02 HY = -2.904E+01	-ZFACE	SY = 5.826E+03 SXY= -4.002E+03 SX = -4.537E+02	+ZFACE	1.022E+04 2.180E+03	4.019E+03 9.321E+03	-42,337
	¥1X1 X2. 23	HXY = -8.611E+01 FX = 4.603E+02	+ZFACF	SX = 1.220E+03 SXY = 1.318E+02 SX = 1.455E+03	-ZFACE	7.2222403 -4.539E+02 VP	3.837E+03 7.460E+03	89.016
	/QUAD / 24	FY = 8.732E+02FXY = -7.866E+02HX = 3.060E+01	-ZFACE	SY = 1.635E+03 SXY = -5.096E+03 SX = 1.864E+02	+ZFACE	6.741E+03 -3.451E+03 VH	5.096E+03 8.979E+03	-44,943
	38,X1 X7, 24	HY = -4.653E+00 HXY = -1.468E+02		SY = 1.858E+03 SXY= 1.950E+03	-ZFACE	3.143E+03 -1.099E+03 VH	2.121E+03 3.814E+03	56,603
	/QUAD / 23	FY = -4.102E+02FXY = -4.471E+02MX = -3.881E+01	-ZFACE	SX = -6.926E402 SY = -1.470E403 SXY= -5.109E403 SX = 1.170E403	+ZFACE	4.843E+03 -6.286E+03 VM	5.124E+03 8.941E+03	-42.824
	31X1	HY = -2.707E+01 HXY = -1.756E+02		5Y = -1.708E+02 5XY= 3.321E+03	-ZFACE	3.888E+#3 -2.888E+03 VM	3.388E+03 5.890E+03	39.292
,	/QUAD / 26	FY = -4.888E+92 FXY = -2.253E+92 HX = -4.368E+91	-ZFACE	3X = -4.659E402 5Y = -6.845E402 5XY= -4.311E403 5X = 1.431F403	+ZFACE	3.738E+03 -4.888E+03 VH	4.313E+03 7.492E+03	-44.274
~	32X1	HY = 1.221E+01 HXY = -1.609E+02	5	5Y = -1.271E+03 5XY= 3.410E+03	-ZFACE	3.886E103 ~3.326E103 VM	5.706E+03 6.421E+03	33,478
	/QUAD / 27 / 23/	FY = -7.349E+02 FY = -1.140E+02 HX = -5.275E+01	+ZFACE S S -ZFACE S	5X = -1.510E+02 5Y = 8.450E+02 5XY= -2.895E+03 5X = 2.381E+03	+ZFACE	3.285E403 ~2.591E403 VH	2.938E+03 • 5.100E+03	-49.880
	33X1 X2. 27	MY = 4.133E+01 MXY = -1.111E+02 FY = 7.215E+02	5	SY = ,-1.139E+03 SXY= 2.439E+03	-ZFACE	3.629E+03 -2.387E+03 VH	3.008E+03 5.246E+03	27.092
	/QUAD / 28	FY = 2.571E+02 FXY = -3.579E+01 HX = -6.161E+01	-ZFACE S	XY = -3.570E+01 XY = 1.838E+03 XY = -1.022E+03 X = 2.922E+03	+ZFACE	2.287E103 -4.850E402 VM	1.386E+#3 2.564E+03	-66.260
	34XI'	HY = 5.514E+01 HXY = -3.958E+01	5	Y = -8.092E+02 XY= 8.784E+02	-ZFACE	3.118E+03 -1.806E+03 VH	2.462E+#3 3.724E+D3	12.607
	/QUAD / 91	FX = -2.180E+02FY = 2.067E+03FXY = 1.173E+00MX = 4.540E+02	+ZFACE S S -ZFACE S	X = 1.846E+04 Y = 7.423E+03 XY= -4.939E+02 X = -1.133E+04	+ZFACE	1.054E+04 7.344E+03 VH	1.597E483 9.360E+03	-9.006
	36X1	HY = 1.370E+02 HXY = -2.068E+01	5	Y = 8.468E+02 XY= 4.986E+02	-ZFACE	8.672E+02 -1.135E+04 VM	6.110E+03 1.181E+04	87.660
	/QUAD / 31	rx = -7.230E401 FY = 1.189E403 FXY = -7.188E402 HX = 2.187F407	+ZFACE S S -7FACF S	X = 5.064E+03 Y = 3.841E+03 XY = -4.130E+03 X = -5.431E+03	+ZFACE	8.628E+03 2.769E+02 VM	4.176E+03 8.493E+03	-40.787
	./ / 38	HY = 6.094E+01	5	Y = 9.153E+02	-ZFACE	1.1548+03 -5.6738+03	3.413E+03	79.213

ĸ

.

2-163

37X1	HXY = -1.122E+82	\$XY= 1.255E	VH	6.3296+03
X2. /\0/ 32	FX = 5.458E+01 FY = 1.619E+02 FXY = -5.074E+02 MX = 3.262E-01	+ZFACE SX = 1.170E SY = 3.835E SXY= -4.868EFUJ -ZFACE SX = 1.013F+02	+ZFACE 4.925E+03 -4.770E+03 VM	4.8488403 -44. 7.3978403
38X1	$\begin{array}{rcl} HY &=& -1.190E+01 \\ HXY &=& -1.597E+02 \end{array}$	SY = 6.094E+02 SXY= 2.818E+03	-ZFACE 3.185E+83 -2.474E+03 VH	2.829E+03 47.576 4.914E+03
X2. 32 /QUAD/ 33	FX = -4.704E-01 $FY = -1.433E+02$ $FXY = -1.996E+02$ $HY = -1.96E+02$	+ZFACE SX = -3.490E+03 SY = -1.85AE+03 SXY= -4.145E+03	+ZFACE 1.550E+03 -6.899E+03 VM	4.225E+03 -50.568 7.791E+03
37X1	$\frac{HY}{HXY} = -6.350E+01$ $\frac{HXY}{HXY} = -1.561E+02$	SY = 1.285E+03 SXY= 3.346E+03	-ZFACE 5.910E+03 -1.136E+03 VM	3.523E+03 35.890 6.552E+03
X2. 33 /QUAD/ 34	FX = -6.363E+01 FY = 6.720E+01 FXY = -5.003E+01 MY = -2.411E+07	+ZFACE SX = -5.914E+03 SY = -2.164E+03 SXY= -2.716E+03 -2FACE SY = 5.659F403	+ZFACE -7.385E+02 -7.339E+03 VM	3.300E+03 -62.307 6.9990+03
41 41X1	$\frac{1}{1000} = -9.578E+01$ $\frac{1}{1000} = -1.890E+02$	SY = 2.433E+03 SXY= 2.516E+03	-ZFACE 7.#35E+03 1.#57E+03 VH	2.989E+03 28.668 6.570E+03
X2. 34 /QUAD / 35	FX = -8.823E+81 FY = 3.323E+02 FXY = -7.304E+00 HX = -2.922E+02	+ZFACE SX = -7.190E+03 SY = -1.989E+03 SXY= -9.440E+02 -7FACF SX = 6.837F+03	+ZFACE -1.823E+03 -7.356E+03 VM	2.766E+03 -80.024 6.635E+03
42 41X1	$\begin{array}{rcl} HY &=& -1.106E+02 \\ HXY &=& -3.872E+01 \end{array}$	SY = 3.318E+03 SXY= 9.148E+02	-ZFACE 7.060E+03 3.095E+03 VH	1.983E+03 13.737 6.130E+03
N X2. 36 Q . /QUAD / 37	FX = -1.266E+02 FY = 1.311E+03 FXY = -2.930E+02 MY = 140F+02	+ZFACE SX = 1.448E+04 SY = 7.013E+03 SXY= -1.482E+03 -7FACE SX = -1.489F105	+ZFACE 1.477E+04 6.730E+03 VM	4.018E+03 -10.823 1.280E+04
44 43X1	$\begin{array}{rcl} HY &=& 1.830E+02\\ HXY &=& -3.734E+01 \end{array}$	SY = -1.771E+03 SXY= 3.103E+02	-ZFACE -1.764E+03 -1.500E+04 VM	6.616E+03 88.656 1.420E+04
X2. 37 . / QUAD / 38	FX = -6.281E+01 FY = 8.476E+02 FXY = -3.661E+02 MY = 3.971E+02	+ZFACE SX = 9.405E+03 SY = 4.692E+03 SXY= -3.307E+03 -7ELCE SY = -0.637E+03	+ZFACE 1.111E+04 2.988E+D3 VM	4.061E+03 -27.262 9.958E+03
44X1	MY = 1.249E+02 MXY = -1.073E+02	SY = -1.301E+03 SXY = 1.843E+03	-ZFACE -9.132E+02 -1.005E+04 VM	4.566E+03 78.099 9.621E+03
X2. 38 . /QUAD / 39.	FX = 5.363E+01 FY = 5.506E+02 FXY = -4.360E+02	+ZFACE SX = 1.933E+03 SY = 1.664E+03 SXY= -4.548E+03	+ZFACE 6.348E+03 -2.751E+03 VH	4.330E+03 -44.132 8.083E+03
46 45X1	HY = 2.345E+01 HXY = -1.532E+02	SY = 5.385E+02 SXY= 2.804E+03	-ZFACE 2.432E+03 -3.613E+03 VM	3.023E+03 55.962 5.268E+03
X2. 39	FX = 7.975E+01 FY = 2.356E+02 FXY = -2.739E+02 HX = -2.027E+02	+ZFACE 5X = -4.786E+83. SY = -9.467E+02 SXY= -4.356E+03 -ZFACE 5X = 5.025E+03	+ZFACE 1.918E+03 -7.570E+03 VM	4.744E+03 -56.670 8.690E+03
46X1	HY = -5.908E+01 HXY = -1.587E+02	SY = 1.889E+03 SXY= 3.260E+03	-ZFACE 7.#75E+03 -1.610E+02 VM	3.618E+03 32.159 7.157E+03
X2. 40	FX = 5.919E+01 FY = 2.542E+02	+ZFACE SX = -9.249E+03 SY = -2.088E+03	+ZFACE -9.937E+02 -3.034E+04	4.675E+03 -69.991



.



41 AU / 41	FXY = -1.008E+02	5XY= -3.006.	NV	9.885E+03
47X1	HX = -3.903E102 HY = -1.082E102 HXY = -1.169E102	-ZFACE SX = 9.48( SY = 3.105. ) SXY= 2.603E+03	-ZFACE 1.041E+04 2.178E+03 VM	4.118E+03 19.23 9.513E+03
X2. 41	FX = 4.228E+01 FY = 3.882E+02 FXY = -1.865E+01	+ZFACE SX = -1.153E+04 SY = -2.331C+03 SXY= -1.059E+03	+ZFACE -Z.218E+83 -1.163E+84 VM	4.718E+03 -83.514 1.071E+04
48X1	HY = -1.295E+02 HXY = -4.257E+01	SY = 3.884E+03 SXY= 9.845E+02	-ZFACE 1.182E+04 3.761E+03 VH	4.028E03 7.073 1.046E04
X2. 43	FX = -2.941E+82 FY = 7.912E+82 FXY = -8.724E+01	4ZFACE 5X = 1.883E+04 5Y = 8.046C+03 5XY= -1.454E+03	+ZFACE 1.893E+04 7.943E+03 VM	5.495E+D3 -5.531 1.447E+04
50X1	HX = 8.092E402 HY = 2.693E402 HXY = -3.666E401	-ZFACE SX = -2.001E+06 SY = -4.881E+03 SXY= 7.056E+02	-ZFACE -4.848E+03 -2.004E+04 VM	7.597E+D3 87.336 1.811E+04
X2: 44 /QUAD / 45	FX = -1.580E+02 FY = 7.347E+02 FXY = -2.794E+02	+ZFACE 5X = 1.296E+04 . SY = 5.869E+03 . SY = -3.860E+03	+ZFACE 1.410E+84 4.732E+03 VH	4.6848483 -20.389 1.2438404
51X1	HY = 1.833E+02 HY = -1.042E+02	-2FACE SX = -1.360E+04 SY = -2.930E+03 SXY= 1.942E+03	-ZFACE -2.588E+83 -1.394E+84 VM	5.675E+03 79.995 1.284E+04
XZ. 45	FX = -2.733E+01 FY = 5.874E+02 FXY = -3.026E+02	+ZFACE SX = 3.683E+03 SY = 2.454E+03 SXY= -4.186E+03	+ZFACE 7.299E+03 -1.162E+03 VM	4.231E+03 -40.824 7.944E+03
52X1	MY = 5.329E+01 MY = -1.492E+02	-2FACE SX = -3.7926+03 SY = -1.043E+02 SXY= 2.975E+03	-ZFACE 1.332E+03 -5.449E+03 VM	3.500E+03 60.894 6.368E+03
X2. 46	FX = 1.102E+02 FY = 5.045E+02 FXY = -2.508E+02 NY = -2.498E+02	+ZFACE 5X = -5.774E+03 SY = -1.003E+03 SXY= -4.263E+03 -7FACE 5X = 4.215E+03	+ZFACE 1.479E+03 -8.257E+03 VM	4.868E+D3 -59.670 9.087E+D3
54 53X1	MY = -8.385E+01 MXY = -1.559E+02	SY = 3.021E+03 SXY= 3.240E+03	-ZFACE 8.234E+03 1.086E+03 VH	3.612E+03 31.884 7.776E+03
XZ. 47 . /QUAD / 48	FX = 1.797E+02 FY = 4.700E+02 FXY = -1.341E+02 HY = -5.61E+02	+2FACE 5X = -1.297E+04 SY = -3.588E+03 SXY= -3.093E+03 -7FACE SX = 1.371E+04	+ZFACE -2.662E+03 -1.391E+04 VM	5.425E+03 -73.325 1.279E+04
55 54X1	MY = -1.837E402 MXY = -1.377E402	SY = 5.468E+03 SXY= 2.556E+03	-ZFACE 1.443E+04 4.739E+03 VH	4.847E+03 15.914 1.274E+04
XZ. 48	FX = 2.803E+82 FY = 4.989E+02 FXY = -3.675E+01	+ZFACE 5X = -1.676E+04 5Y = -4.829E+03 5XY= -1.122E+03	+ZFACE -4.724E+03 -1.687E+04 VM	4.872E483 -84.677 1.507E+04
./ 42/ ./ / 56 53X1	MX = -7.152E+02 MY = -2.428E+02 MXY = -4.368E+01	-2FACE SX = 1.757E+04 SY = 6.823E+03 SXY= 9.748E+02	-ZFACE 1.765E+04 6.737E+03 VM	5.458E+03 5.144 1.543E+04
X2. 50 . /QUAD / 51	FX = -2.410E+02 FY = 6.092E+02 FXY = -8.901E+00	+ZFACE 5X = 2.213E+04 5Y = 5.286E+03 5XY= -9.946E+02	+ZFACE 2.219E+04 5.228E+03 VM	8.479E+03 -3.368 2.009[+04
. / 43/ ./ / 58 57X1	TX = 9.420E+02 MY = 1.695E+02 MXY = -4.070E+01	-ZFACE 5X = -2.309E+04 SY = -2.850E+03 SXY= 9.590E+02	-ZFACE -2.804E+83 -2.314E+04	1.017E+04 87.294 2.187E+04

2-165

.

x2. // / 32	FX $= -1.765E+82$ $+ZFACE$ SX $= 1.571E+44$ FY $= 6.109E+02$ SY $= 4.125E+4$ $+ZFACE$ $1.634E+04$ $3.697E+03$ $6.419E+03$ $-12.7$ FXY $= -2.148E+01$ SXY $= -2.771E+4$ VM $1.490E+04$ MX $= 6.691E+02$ $-ZFACE$ $= -1.641E+4$ VM $1.490E+04$ MY $= 1.218E+02$ SY $= -1.642E+03$ $-ZFACE$ $= -1.687E+03$ $= -1.687E+03$ $= 79.986$ MXY $= -1.317E+02$ SY $= -1.687E+03$ $= -2FACE$ $= -1.687E+03$ $= -1.687E+03$ $= 79.986$	
X2. 52 /QUAD / 53 / 45/ / 60	FX = -3.100E+01 $+ZFACE$ $SX = -2.035E+03$ $+ZFACE$ $7.09E+02$ $4.204E+03$ $FY = 6.064E+02$ $SY = 2.132E+03$ $+ZFACE$ $7.09E+02$ $4.204E+03$ $-34.687$ $FXY = -3.988E+01$ $SXY = -3.934E+03$ $VM$ $8.128E+03$ $HX = 2.165E+02$ $-ZFACE$ $SX = -5.298E+03$ $VM$ $8.128E+03$ $HY = 3.831E+01$ $SY = 2.936E+02$ $-ZFACE$ $2.195E+03$ $-7.197E+03$ $4.697E+03$ $63.262$ $HXY = -3.66E+02$ $SY = 2.936E+02$ $-ZFACE$ $2.195E+03$ $-7.197E+03$ $63.262$	
X2. 53 / QUAD / 54 / 46/ / 61	FX = 7.265E+01 $4ZFACE$ SX = -6.414E+03         FY = 5.896E+02       SY = -2.588E+00 $4ZFACE$ $1.959E+03$ $-8.375E+03$ $5.167E+03$ $-64.171$ FXY = -3.632E+01       SXY = -4.053E+03       VH $9.507E+03$ $VH$ $9.507E+03$ HX = -2.733E+02       -ZFACE       SX = 6.704E+03 $-ZFACE$ $9.803E+03$ $6.238E+01$ $4.470E+03$ $30.468$ HY = -4.924E+01       SY = 2.361E+03 $-ZFACE$ $9.803E+03$ $6.238E+01$ $4.470E+03$ $30.468$	
X2. 54 /QUAD / 55 / 47/ / 62 61	FX = 1.783E+82 + ZFACE 3X = -1.571E+04 $FY = 5.867E+02 SY = -1.696E+03 + ZFACE -1.867E+03 -1.634E+04 7.638E+03 -78.291$ $FXY = -2.508E+01 SXY = -3.036E+03 VH 1.584E+04$ $HX = -6.690E+02 - ZFACE 5X = 1.640E+04$ $HY = -1.196E+02 SY = 4.043E+03 - ZFACE 1.706E+04 3.351E+03 6.838E+03 12.709$ $HXY = -1.244E+02 SY = 2.935E+03$	•
X2. 33 N /QUAD / 36 48/ 6 42 / 43	FX = 2.196E+02       +2FACE 5X = -2.079E+04         FY = 5.839E+02       SY = -2.614E+03       +2FACE -2.545E+03       -2.086E+04       9.159E+03       -86.476         FY = -8.317E+00       SY = -1.124E+03       +2FACE -2.545E+03       -2.086E+04       9.159E+03       -86.476         HX = -8.847E+02       -2FACE 5X = 2.167E+04       -2FACE 2.174E+04       4.879E+03       8.432E+03       3.716         HY = -1.576E+02       SY = 4.950E+03       -2FACE 2.174E+04       4.879E+03       8.432E+03       3.716         HXY = -4.614E+01       SXY= 1.091E+03       -2FACE 2.174E+04       4.879E+03       8.432E+04       3.716	
X2. 15 /QUAD / 64 / 47/ / 66 68	FX       = -4.922E+03       +2FACE       SX       = -1.635E+04         FY       = -4.319E+03       SY       = -6.836E+03       +2FACE       -3.919E+03       -1.927E+04       -7.676E+03       -64.165         FY       = -3.334E+03       SY       = -6.021E+03       VM       1.766E+04         MX       = -9.262E+03       -2FACE       SX       = 1.143E+04         MY       = -3.117E+03       SY       = 2.516E+03       -2FACE       1.218E+04       1.768E+03       5.206E+03       15.542         MXY       = -2.903E+03       SXY=       2.688E+03       -2FACE       1.218E+04       1.768E+04       1.140E+04	
X2. 68 /9UAD/ 66 / 50/ / 67	FX = 3.120E+03 + 2FACE 5X = -1.543E+03 FY = 5.106E+03 + 2FACE 5X = -1.543E+03 + 2FACE 7.675E+03 - 2.280E+03 4.777E+03 - 74.438 FXY = 1.719E+02 5XY = -2.573E+03 VM 9.033E+03 MX = -2.882E+03 - 2FACE 5X = 4.683E+03 - 2FACE 5.683E+03 - 2.852E+03 4.267E+03 20.013 MXY = -1.772E+03 5XY = -2.754E+03 - 2FACE 5.683E+03 - 2.852E+03 4.267E+03 20.013 MXY = -1.772E+03 5XY = 2.754E+03 - 2FACE 5.683E+03 - 2.852E+03 4.267E+03 20.013 MXY = -1.772E+03 5XY = -2.754E+03 - 2FACE 5.683E+03 - 2.852E+03 4.267E+03 20.013 MXY = -1.772E+03 5XY = -2.754E+03 - 2FACE 5.683E+03 - 2.852E+03 4.267E+03 20.013 MXY = -1.772E+03 5XY = -2.754E+03 - 2FACE 5.683E+03 - 2.852E+03 4.267E+03 20.013 MXY = -1.772E+03 5XY = -2.754E+03 - 2FACE 5.683E+03 - 2.852E+03 4.267E+03 20.013 MXY = -1.772E+03 5XY = -2.754E+03 - 2FACE 5.683E+03 - 2.852E+03 4.267E+03 20.013 MXY = -1.772E+03 5XY = -2.754E+03 - 2FACE 5.683E+03 - 2.852E+03 4.267E+03 20.013 MXY = -1.772E+03 5XY = -2.754E+03 5XY	
X2. 66 /QUAD / 64 / 51/ / 65 67	FX       *       4.449E403 $4ZFACE$ SX       =       8.797E403 $4ZFACE$ 8.853E403 $-5.793E403$ $7.323E403$ $3.523$ FY       *       2.217E403       SY       = $-5.738E403$ $4ZFACE$ $8.853E403$ $-5.793E403$ $7.323E403$ $3.523$ FXY       = $-1.944E403$ SXY $8.982E402$ VH $1.278E404$ HX       = $4.382E403$ $-ZFACE$ $SX$ $= -4.349E403$ $SX$ $= -7.955E403$ $-77.603$ HY       = $-4.564E403$ SY $= 7.955E403$ $-ZFACE$ $8.579E403$ $-4.974E403$ $6.776E403$ $-77.603$ HXY       = $1.247E403$ SXY $-2.862E403$ $VH$ $1.187F404$	
X2. 64 /QUAD / 16 / 52/	FX = -1.223E+03 + 2FACE SX = -3.866E+03 FY = 8.804E+02 SY = 3.294E+03 + 2FACE 7.530E+03 - 8.101E+03 7.815E+03 - 58.630 FXY = -2.351E+03 SXY = -6.947E+03 VM 1.354E+04 MX = -5.917E+01 - 2FACE SX = -1.025E+03	

.







.





.

٠

				•
19UAD / 67/	FY = -2.423E+81 FXY = 5.378E+82 MX = 4.925E+02	SY = 2.418E+0; SXY= -6.426E+0; ~ZFACE SX = -5.111F+0	2 +2F 1.221E+03 -1.798E+02 7.805E+02 -33. 2 VM 1.320E+03	272
X1	HY = 4.435E+81 HXY = -1.967E+82	SY = -2.903E10 SXY= 1.718E+0	2 -ZFACE 2.395E+02 -5.660E+03 2.960E+03 72. 3 VM 5.794E+03	257 2 Pr
	FX = -1.067E+03 FY = -1.009E+03 FXY = 2.681E+02 HX = -3.688E+02	+ZFACE SX = -1.051E404 SY = -5.987E402 .SXY= -4.218E402 -ZFACE SX = 6.237E40	4 3 +ZFACE -3.948E+83 -1.855E+84 2.299E+83 -84. VM 9.158E+83	adrila
X1	MY = -1.654E+02 MXY = -3.991E+01	SY = 1.952E+0 SXY= 1.494E+0	3 -ZFACE 6.707E+03 1.483E+03 2.612E+03 17. VH 6.102E+03	445 n c.
22 29UAD / 65 2 69/	FX = 4.513E+02 FY = 3.551E+03 FXY = -6.563E+02 MX = 2.418E+02	+ZFACE SX = 6.707E+01 SY = 9.142E+01 SXY= -2.114E+01 -7FACE SY = -4 002E+01	3 3 +ZFACE 1.036E+04 5.484E+03 2.440E+03 -59. VH 8.981E+03	1 0 5 5 769 Fr 3
×1	HY = 8.498E+81 HXY = -3.342E+01	SY = 5.063E+01 SXY= -5.106E+02	5 -ZFACE 5.089E+83 -4.928E+83 5.809E+93 -87. VM 8.676E+03	075 es
		•		and .
				Forces
N				for
-168				Outpu Outpu
				st Ve
				:

.

٠

### 2.11 APPENDIX

# ENGINEERING DROP TEST FOR 1-13C II Shipping package

### 2.11.1 <u>SCOPE</u>

This appendix summarizes the plans for, conduct of, and results from the engineering drop test for the 1-13C II shipping package.

### 2.11.2 <u>REFERENCES</u>

Prior to performing the drop test, numerous planning documents were generated detailing data collection procedures and acceptance criteria. These reference documents are included in microfiche form in Section 2.11 on the microfiche entitled "2.11 Appendix: Drop Test Report Reference Documents 1-13C II Shipping Package." The documents include:

- 2.11.2.1 CNSI Procedure NE-TI-001, Rev. -, Drop Test Procedure for CNS-1600 Shipping Package 1-13C II: This document summarizes the planned sequence of events for the package drop test.
- 2.11.2.2 NuPac Procedure DT-O1, Rev. 2. Drop Test Requirements Hypothetical Accident Conditions (Type "B") for the 1-13C Model 1600 Package: This document summarizes the drop test objectives, general strategy, and performance criteria. Included in this document are detailed procedures for obtaining bolt force measurements and dimensional survey data.
- 2.11.2.3 CNSI Procedure, 5-QP-004, Rev. -, "Liquid Penetrant Inspection Procedure: This inspection procedure, in conjunction with the Industrial NDT procedure QCP-401 reporting format, was used to perform dye penetrant pre- and post-tests.





- 2.11.2.4 NuPac Document CL-01, Rev. 0, Gamma Scan of GNSI Model 1600 Cask: This document, in conjunction with references 2.11.2.6 and 2.11.2.7, was used to perform the post-test gamma scan of the 1-13C II cask body.
  2.11.2.5 X-Ray, Inc. Procedure No. VT-001 Visual Inspection of Weldments and Adjacent Materials.
  2.11.2.6 NuPac Document GS-002, Rev. 0, General Procedure for Gamma Scan of Shielded Containers - Field Calibration Method.
- 2.11.2.7 Industrial NDT Procedure 601, Ultrasonic Testing for Metal Thickness and Soundness.





### 2.11.3 DETAILED PROCEDURE: CONDUCT OF DROP TEST

### 2.11.3.1 Overview and Summary

The 1-13C II drop test was conducted in September, 1980, at the Chem-Nuclear site in Barnwell, South Carolina. Numerous pre-tests were performed to establish shielding integrity, and package dimensions of a 1-13C II cask. In the actual drop sequence, the cask was loaded with a dummy payload, sealed, and fitted in its transport overpack. The package was then lifted to a height of 30 feet and dropped vertically onto a concrete pad. After this bottom end flat drop, a new overpack pair was installed and the package was again dropped from a height of 30 feet, this time with the top (lid) end down and a package centerline orientation of approximately  $40^{\circ}$  with respect to horizontal. Following this top end oblique drop, post-test inspections were accomplished to document package changes and to verify post-drop shielding integrity. The package survived both drops, meeting all post-test shielding accpetance criteria. The overpacks, although deformed, did not fail. The cask itself was intact except for one cracked lid bolt boss which occurred because of interference between bolt force measurements transducers and the overpack tiedown plate. This interference allowed the overpack to load lid bolt heads in bending introducing a severe and unrealistic moment into the bolt boss. Furthermore, a dimensional review showed that if the maximum bolt head diameter (i.e., width across corners) was used, the edge of the 3 inch clearance hole in the overpack would contact the bolt head before the internal side wall of the overpack would contact the side of the cask. Modifications have since been made to the overpack to enlarge this 3 inch clearance hole to 3.548 inches. This dimensional change allows for sufficient clearance to eliminate interference between overpack tiedown plate and lid bolts. This modification is expected to have no effect on the energy absorbing capabilities of the overpacks. Also, a 1/8 inch thickness of neoprene has been added, as an option, to the inner vertical walls of both the upper and lover overpacks. This neoprene was added exclusively to minimize any chafing between the overpacks and cask,







and is not required to eliminate any previous tolerance problem in the area of the cask lid bolts. In addition, flat washers of 2 1/2 inches outside diameter have been added under each of the twelve lid bolts to eliminate any potential of galling between the bolt heads and the contact surface of the lid.

The remainder of this section provides a summary narrative of test procedures, with an emphasis on those areas which deviate from the planning documents (references 2.11.2.1 through 2.11.2.8).

### 2.11.3.2 <u>Pre-test Liquid Penetrant Inspection</u>

On September 23, 1980, a certified inspector from Industrial NDT Company, Inc., performed a liquid penetrant inspection of all package weldments. The liquid penetrant inspection was observed by a CNSI Quality Assurance Representative to verify the integrity of the data and adherence to the written inspection procedures.

# 2.11.3.3 Pre-test Gamma Scan

A pre-test gamma scan was performed on the 1-13C II cask on September 26, 1980, with the intention of comparing the results with post-drop data to determine shielding integrity. A Cobalt-60 (9 curies) source certified by Pittsburgh Testing Laboratory was used in the test. Unfortunately, following the conclusion of testing, the gamma scan technician discovered that the scintillator being used had a cracked crystal, thus invalidating all of the pre-test gamma scan data. The original strategy was therefore revised so that post-drop gamma scan data were compared with a calibration curve equal to nominal cask lead thicknesses. Acceptance criteria were also revised so that any obtained readings exceeding nominal lead thickness minus 10% were designated failure.



#### 2.11.3.4 Package Preparation

Following dimensional measurements of the package, the package was loaded and sealed. A 3,000 pound dummy payload of sand and lead shot was loaded into the cask. The lid gasket and lid were then put on the cask and the lid bolts torqued to 330 ft.-lbs. (lubricated). After installing bolt force measurement instrumentation, the cask was inserted in its upper and lower overpacks. Ratchet binders connecting the overpacks were torqued to 70 ft.-lbs. and the entire package weighed. Final package weight, verified by CNSI Quality Assurance representatives, was 26,500 pounds. Photo F1 shows the package being put into the lifting sling just prior to the flat end vertical drop.

### 2.11.3.5 Photographic Documentation

The drop test was documented in great detail photographically. Actual drop sequence (bottom end vertical and top end oblique) were recorded using a high speed l6mm movie camera shooting 500 frames per second on negative color film. The camera was placed so that it would record the package falling in front of a plywood backdrop with a 6 inch grid pattern. Drop sequences were also recorded at normal speed (24 frames per second) on l6mm negative color film. In addition to the movies, still pictures were taken of the drop sequences (35mm, ten frames per second). Finally, numerous 35mm color stills were obtained before and after each drop to provide detailed documentation of pre- and post-drop package condition.

### 2.11.3.7 <u>Bolt Force Measurement</u>

Prior to installing the upper overpack, the instrumentation to measure bolt tension was implemented. Bolt tension was monitored by custom fabricated load cells. The cells were designed to fit under bolt heads as washers. They were compressed between the bolt head and lid top surface when the bolts were torqued down. Strain gage



2-173





PREPARATION FOR BOTTOM END VERTICAL DROP



I F1 ATTACHING THE LIFTING SLING TO UPPER OVERPACK.



L F2 SLING HARDWARE BEING ATTACHED TO GUICK PELEASE HOOK ON CRANE.



FINAL CASK DRIEVIATION AT 30" JUST PPIDE TO BOTTOM END VERTICAL DROP.





#### 2.11.3.8 First Drop: Vertical Bottom End

The cask and overpacks were put in a lifting sling which was attached to a quick release hook on the crane (see photo F2). The entire package was then raised to a height of 30 feet above the concrete dop pad as shown in photo F3. Also shown are the instrumentation cables and tag lines Photos F4 to F7 depict the package during the actual drop: the cask impacted on the target, rebounded to height of about 8 inches, then impacted again at a slightly oblique angle and came to rest.



2-175

Photos F8 to F19 illustrate in greater detail package behavior during the first drop. Under secondary impact, the package rotated slightly and impacted on the corner between quadrants I and IV, causing slight mushrooming confined to the lower five inches of the lower overpack (see photo F16). No general crush of the lower overpack was evident at this time. The upper, surface of the lower overpack demonstrated ballooning due to the compressive forces of the foam (see photos F14 and F15). The impact in quadrants I and IV caused the ratchet binders on that side to deform and to buckle (see photo F17). On the opposite side (quadrants II and III), the ratchet binders were loaded with tensile forces, but there was little evidence of distress except in the attached clevis bolts, as seen in photo F18. The upper overpack showed corresponding compression and tensile loading, although there was no distress to ratchet binder lugs or to the overpack shell itself.

Although not evident in these photos, force washer #2 had its lead lines severed 80 milliseconds after impact and experienced undetermined internal damage. Bolt force measurement data showed that the pre-test load (15,700 pounds on force washer #1 and 9,600 pounds on #2, with 200 ft.-lbs. of torque on the bolts) were not permanently reduced by the impact of the drop. In fact, they were slightly increased. Temporary reduction of the initial clamping force by approximately 20% was seen for less than 4 milliseconds at impact. Otherwise, the clamping force was increased slightly and remained so after impact perturbations had ceased.

In summary, visual examination of the package indicated absolutely no damage to the 1-13C II cask body or lid. The overpacks and ratchet binders did not fail, although the lower overpack sustained moderate damage.

### 2.11.3.9 Drop II: Top End Oblique Drop

In preparation for this drop, the overpacks used in previous tests were removed and new ones installed. During this process, force









۰.





. -

OVERALL PACKAGE CONDITION AFTER ROTTOM END VERTICAL DROP



COMPRESSIVE CRIPPLING OF OVERPACK PATCHET BINDERS IN CUADPANT I; NO DAMAGE TO CASK.



COMPRESSIVE CRIPPLING OF PATCHET BINDERS IN QUADRANT IV; NO DAMAGE TO CASK.







TENSILE LOADING OF PATCHET BINDERS IN DUADPANT 11; NO DAMAGE TO CASK.





VERTICAL DROP DAMAGE: DETAILED VIEWS









2-179



#### BOTTOM END VERTICAL DROP DAMAGE DETAILS (CONTINUED)



→ F15 (LEFT) FINAL LOVER OVERPACK GRIENTATION FOLLOWING VERTICAL DROP. PAINT PATTERN ON DROP PAD SHOWS PACKAGE POSITION AT INITIAL IMPACT.

F17 (RIGHT) → PATCHET BINDER LUG (#2 BETWEEN OUANDRANTS 1 AND IV) SHOWING CRIPPLING OF RATCHET BINDER; ALSO, DUCTILE TEAR OF LUG TO OVEPPACK SHELL WELD; LUG REMAINS FIRMLY ATTACHED TO VENTICAL CYLINDRICAL WALL OF OVEPPACK SHELL.





- F18 (LEFT) RATCHET BINDER (13, IN QUADRANT 11) SHOWING FLEXUPAL YIELDING OF BOLT TO LUG UNDER PATCHET BINDER TENSILE LOADING; LUG REMAINS FIRMLY ATTACHED TO OVERPACK SHELL.

F19 (RIGHT) -UPPER OVERPACK LUG IN COMPRESSIVE PEGION; NO DISTPESS TO LUG OR TO OVERPACK SHELL; NO DAMAGE TO UPPER TWEPPACK OR CASK.





washers #3 and #4 were installed at  $90^{\circ}$  angle from force washers #1 and #2. Although the #2 force washer had been damaged during the first drop, as attempt was made to use it as well as force washer #1. The specially made top impact lift fixture, orienting the package of  $40^{\circ}$ , was installed and the force washer instrumentation cables were secured so that they sould not be damaged or loosened during the cask drop. The package was lifted to a height of 30 feet and oriented at approximately  $40^{\circ}$  with respect to horizon. Photos T1 and T4 depict preparation for the oblique drop.

When the package was dropped, initial impact was on the Quadrant IV corner of the upper (lid end) overpack as shown in photo T5. It then rebounded, and the lower overpack experienced initial slapdown impact (see photo T6). The upper overpack impacted a second time while the lower overpack was still in rebound (photo T7).

The lower overpack experienced second and third slapdown impacts, and the package came to rest (see photo T8). At initial impact, all four force washers were destroyed and no significant data was transmitted. Visual inspection of the package (discussed at length in the next section) after the oblique drop revealed some crushing in both the top and bottom overpacks, but no bottom-out. Bottom-out was defined as a deformation of the overpacks exceeding 95% of the overpack foam thickness measured perpendicular to the crush plane. Had this occurred, a third (bottom end oblique) drop would have been necessary. The conditional third drop was vaived and post-test inspections commenced.







# PREPARATION FOR TOP END CELLOUE DROP





1 T3 2411:120 TASUPETENTS VEPTEVING 40<sup>00</sup> 0011:NTATION.



1 14 PACKAGE BEING RAISED TO NOT HEIGHT PRIOR TO SECOND DOOD.

• •



2-182

n::\_\_\_\_\_\_

**.** . . . .

. .



2-183

• • • •

### 2.11.2.10 Post-test Visual Inspection

Photos T9 and T20 illustrate post-test visual examination findings. It should be noted that the drop orientation was selected to maximize potential damage to the cask lid and to the upper overpack shell. In spite of this "worst case" strategy, the upper overpack shell evidenced only moderate compression (photo T9). Furthermore, although the tiedown lugs deformed to conform to the overpack shell shape, there was only moderate distress in the vicinity of the tiedown lugs and to rupturing or tearing (photo T10).

The interior upper overpack shell likewise suffered little damage, although it caused problems with the force washer instrumentation. Photo T16 shows that bolt relief holes fashioned to accomodate the force washer instrumentation cables which had to be routed through the upper overpack. Unfortunately, inadequate clearances in the relief holes caused the vertical inner walls of the overpack to impact on the sides of the force vashers. This lateral impact greatly increased the loads applied to the cask closure bolts. In the most extreme case, this resulted in a crack around 3/4 of the circumference of a lid bolt boss (photo T17). The boss was also raised 1/8 inch above the cask top surface. It should be noted that in service, no such loads would be used. In addition, the clearance holes in the overpack which receive the lid bolts have been enlarged from 3 inches to 3.548 inches, and a 1/8 inch thickness of neoprene has been added, as an option, to the inner vertical valls of both the upper and lover overpacks. Hence, the failure can be considered the result of an unrealistic over-test. The lateral impact also destroyed the force washers (see photo T18).

Visual inspection of the lower overpack, which experienced three slapdown impacts, showed only minor damage. In photo T14, the cask registry can be seen running along the circumference of the inside lower plate. There was no rupturing or tearing of the interior shell, however. Also shown is slight paint distress indicating some yielding along the vertical weld attaching ratchet binders in







· • •

• • • • • •

.


TOP END CELICUE DROP DAMAGE: DETAILED VIEWS



T14 (R15HT) 🛏

TVEPPACK PLATE.

- 113 (1257) REMOVAL OF UNDAMAGED CASK FROM LOWER CVERPACK AFTER DELIGUE DROP: UPPER CVERPACK USED IN SECOND DROP IS SHOWN IN BACKGROUND.





- 115 (11970) SLIGHT PAINT DISTRESS AT LOVER OVERPACK PATCHET BINDER LUS ATTACHMENT INDICATES MODEST STRAINS AFONE MATERIAL ELASTIC LIMITS: NO PUPTURES OF DUTER OVERPACK SHELL OCCUPPED.



2-186





TOP END CELIQUE DPOP DAMAGE DETAILS

I TIS (ABOVE) INTERIOR OF UPPER OVERPACK, WHICH EXPERIENCED INITIAL IMPACT; BURN MARKS APE FROM TORCHING OF CASK LID BOLT RELIEF HOLES TO ACCOMPODATE FORCE WASHER INSTRUMENTATION CABLES.



1 T17 (ABOVE) CIPCUMPERENTIAL CRACK ON CASK LID BOLT BOSS CAUSED BY CATERIAL IMPACT OF OVERPACK LID BOLT RELIEF HOLE ON FORCE MASHER.



FORCE WASHER VS. DESTROYED BY LATERAL IMPACT OF DWERPACK BOLT RELIEF HOLES; INADECHATE CLEARANCES IN BOLT RELIEF HOLES RESULTED IN DWERTEST OF CASK CLOSURE BOLTS.

#### T19 (BELCV)

CASK LEAK CHECK AFTER OBLIQUE DROP; DESPITE DAMAGE TO FORCE WASHERS AND CASK LID BOLT BOSS, LEAK RATES DID NOT EXCEED ACCEPTANCE LIMITS.







TZO (LEFT) CASK LID AND BODY FOLLOWING SECOND DROP; CASK PEMAINS UNCHANGED EXCEPT FOP CFACKED LID BOLT BOSS AND OVER-PACK PAINT PEGISTEY MARKS.





Quadrants II and III to the overpack inner shell. Photo 715 shows additional paint distress on the upper surface of the lower overpack at the ratchet binder lug attachment. Although this indicates modest strains above material elastic limits, there were no ruptures or tears of the outer overpack shell.

## 2.11.3.11 Post-test Dimensional Checks and Bolt Retorquing Values

The cask lid bolts were retorqued to their original values of 330 ft.-lbs. and the number of degrees required to obtain this value was documented. Not surprisingly, the two bolts requiring the most retorquing to obtain original values were those closest to the impact plane after the second drop. Lid bolts #5 and #9 required  $25^{\circ}$  and  $15^{\circ}$  respectively to obtain 330 ft.-lbs.; both were adjacent to the torn boss insert under Force Washer #2.

Post-drop cask diametrical measurements were also obtained. These measurements were identical to tee pre-drop measurements within error tolerance (1/8 inch).

## 2.11.3.12 Dve Penetrant Post-test

On October 2, 1980, the post-test liquid penetrant inspection was performed by Industrial NDT Company, Inc. Results were identical to pre-test findings with the exception of the cracked bolt hole boss discussed earlier.

## 2.11.3.12 Post-test Gamma Scan

Due to equipment difficulties noted earlier, the post-test gamma scan was not concluded until February, 1981. The test was performed by the Quality Assurance Manager of Nuclear Packaging, Inc., of Tacoma, Washington. The following preliminary inspection: were performed and verified:

- A. Exterior surface wipe test.
- B. Held integrity of external heat shield welds.
- C. Cask dimensions



The cask surface was gridded in four inch squares, and the probe and source were calibrated in accordance with Gamma Scan Procedure No. GS-002, Rev. O. The cask was first scanned without the 9 curie Gobalt 60 source, and all readings were recorded on a grid map corrsponding to the cask surface. The source was then inserted into the cask, calibration rechecked, and a normal gamma scan completed. The "background" readings (no source in cask) for each grid were subtracted from readings obtained with the source in the cask. Obtain differences were the actual gamma readings in millirems per hour for each grid. Cask lead integrity was shown to be fully acceptable with no detectable lead settlement.

Gamma scan of the cask lid was not practical, so it was subjected to ultrasonic soundness inspection, performed by a certified Industrial NDT technician. No relevant indications were detected.

#### 2.11.4 TEST\_RESULTS/DATA



#### 2.11.4.1 <u>Summarry</u>

Inspections and tests of the 1-13C II cask provided dramatic confirmation that the package is structurally adequate to survive accidential drops from 30 feet, in any orientation. The overpacks, although deformed, did not fail. The lid was not deformed. The cask itself was not deformed except in areas deformed by testing instrumentation. Post-drop package shielding capabilities, as measured by gamma scan data, were clearly acceptable. The damage which was sustained by the lid bolt boss noted previously is not expected to occur if the cask is subjected to hypothetical accident conditions in the future.

A potential interference problem existed between the bolt head (i.e., width across corners) and the edge of the 3 inch clearance hole in the overpack. The overpacks have since been modified to enlarge the lid bolt clearance holes from 3 inches to 3.548 inches. This eliminates any possibility of the overpack contacting the lid bolts when the package is subjected to hypothetical accident conditions.



2-189

Because of this design modification no damage is expected if the cask is again subjected to the hypothetical accident conditions. Therefore, the pre-accident cask condition is actually representative of the post-accident cask condition. Moreover, the pre-accident cask condition may be tested to demonstrate compliance with the post drop-test containment requirements of NUREG. 7.4 and ANSI N 14.5. This also affords the flexibility of making package design improvements without requiring additional testing. Additional package design improvements are noted as follows: An O-ring has been added to the lid to facilitate leak testing in accordance with NUREG. 7.4 and ANSI N 14.5 for assembly verification. There was no consideration given to this O-ring to enhance the sealing capabilities of the package. However, the addition of this O-ring does augment the existing sealing competence. The 3/8 inch thick silicone gasket has been reduced to a thickness of 1/4 inch to ensure metal to metal contact between the lid and the cask. This will result in more repeatable results when performing future leak tests.

#### 2.11.4.2 Dye Pentrant Inspections

Figures 2.11.4-1 and 2.11.4-2 show the pre-drop and post-drop dye penetrant inspections of the 1-13C II cask.

#### 2.11.4.3 <u>Dimensional Checks</u>

Figure 2.11.4-3 shows the pre- and post-test cask diametrical measurements. Pre-test figures are in the upper level and post-test measurements in the lower level. It was determined from a post-test dimensional survey completed at Barnwell, South Carolina, by CNSI Quality Assurance personnel that all pertinent areas of the cask lid which might have been affected by the drop test remained within fabrication tolerances.

Figure 2.11.4-4 is a sketch depicting the lower overpack after the bottom end vertical drop. Deformation of the upper overpack after top end oblique drop is shown in Figure 2.11.4-5.



#### 2.11.4.4 Gamma Scan Data

Figure 2.11.4-6 shows the calibration curve obtained prior to gamma scan of the 1-13C II cask. Also calculated in this figure are the gamma scan acceptance criteria derived by subtracting 10% from cask nominal lead thickness (which was 5.00 inches except at the lifting ear support bond and in the tapered lid insertion area) and correlating these figures with calibration readings obtained with lead plates of the same thickness. The reader will note that the acceptance thresholds were 49.5 mR./hr. at lifting ear support band and 26.0 MR. for the rest of the cask. The 49.5 mR/hr. figure was projected, since no calibration lead plate with 3.82 inches thickness was available.

Figure 2.11.4-7 (3 pages) shows the post-test gamma scan with the 9 curie Cobalt 60 source inside the cask. No reading at lifting ear support band exceeds the maximum allowable limit of 49.5 mR./hr. No reading for the rest of the cask body exceeds the maximum allowable limit of 26.0 except in row 14. These higher readings are due to the taper present in the top area for insertion of the mating taper of the lid. At this taper, the cross section lead thickness is only about 50% of that in the main cask body.

#### 2.11.4.5 Bolt Force Measurement Data and Bolt Retorquing Data

Figure 2.11.4-8 shows a top down view of the cask lid bolts and force washers. Included in this sketch is the number of degrees to achieve original lid bolt torque values (330 ft.-1b.) after the two drops.

Bolt force time history for the first drop impact can be seen in Figure 2.11.4-9 (two pages).









10 11 11 12 1/16<sup>+</sup> WIDE PINHOLE PINHOLE

5

9

IMPACT PLANE

FIGURE 2.11.4-2 FOST TEST DYE PENETRANT FINDINGS

9

Revision 0

BOLT HOLE CRACKED 3/4 OF DIA. RAISEI

EXTENDING INTO BASE

1/8" NO CRACKS

BASE METAL.

BOLT HOLES 11-12

1/32\* PINHOLE DIRECTLY

1

DUTER WELD.

UNDER 13 BOLT HOLE

3/32" WIDE PINHOLE -

8

7

THNER WELD-

CORRESPONDING ORIENTATION OF HOLE LOCATION NUMBERING SYSTEM USED ON ALL OTHER INSPECTIONS

\$

È

ရ







1





FIGURE 2.11.4-3

CASK DIAMETRICAL MEASUREMENTS

SUMMARY DATA SHEET



Γ	<del>ار معانی</del>				.•	C	IRCUMFER	ENTIAL P	OSITION	- DIAMET	ER (1N.	+ 1/8")		
:			1'	2.	3	1.4 1	5 1	6	<u> </u>	8	<u></u>		<u>  11 '</u>	12
<sup>:</sup> [		e c	39-3/16"	39-3/15	39-3/16"	39-3/16"	39-3/16-	39-3/16-	39-3/16"	39-7/32-	39-3/16"	39-3/16-	39-7/32*	39-9/32"
	A	- 01 1	39-1/16-	39-3/32"	39-1/16"	39-1/8"	39-1/8"	39-1/16-	39-1/16-	39-1/8-	19-5/32"	39-3/16"	12-1/4-	139-5/32-
	B	EXTER	39-1/4"	39-1/8	39-1/8"	39-1/8"	<u>39-1/8"</u>	39-9/32	39-7/32"	39-1/16-	39-1/8"	39-1/8	39-1/8-	39=3/16*
Ļ		/	35-1136	37-1/0	13-1/32	19-3/16	39-5/32-	39-1/4"	39-1/4-	39-3/16-1	39-5/32-	39-1/16-	39-1/16-	39-1/16-
	r	, 1	26-7/16-	26-7/16-	26-11/32-	26-3/8-	26-3/8-	¢6-11/32−	26-11/32	26=3/8"_	26-11/32	26-7/16	26-7/16=	26-15/32
	~	, }	26-7/16*	26-3/8"	26-3/8"	26-3/8-	26-3/8-	26-5/16-	26-5/16-	26-11/32	26-3/8"	26-7/16-	26-1/2"	26-15/32
	n	IOR	26-11/32	26-11/32	26-7/16"_	26-7/16-	26-15/.12-	?e-15/3?*	26-15/32:	26-3/8"_	26-3/8"_	26-5/16.	26-9/32-	26-13/12
	۲ ا	1 H	26-3/8*	26-3/8*	26-13/32-	26-7/16*	26-15/12-	<u>p:6-15/32"</u>	26-13/32-	26-3/8-	26-5/16-	26-5/16-	26-1/4-	26-1/8"
	F	<u> </u> Ξ'	26-7/16-	26-7/16-	26-7/16-	26-15/32-	26-7/16-	26-1/2-	26-1/2-	26-7/16-	26-3/8"	26-3/8-	26-3/8-	26-7/16"
	۲ J	1'	26-3/8-	26-7/16*	26-3/8-	26-7/16*	26-1/2"	26-1/2"	26-15/32-	26-3/8"	26-3/R*	26-3/3*	26-3/8"	26-1/2*
ſ	1				NOTE:	FIGURES FIGURES	IN UPPER	LEVEL A	RE POST-	EST MEAS TEST MEA	UNCHENTS SUREMENT	: 5.		











LOWER OVERPACK AFTER FLAT END DROP

1



)



.

.







Revision O

•

## FIGURE 2.11.4-7 GAMMA SCAN READINGS IN MR OF 1-13C II CASK WITH 9 CURIES COBALT 60 SOURCE

												<u></u>			•	**			_
		Ē						ľ		•					·				 
				1				1											
		}-		<u> </u>	+	<u> </u> 			! 	<u> </u>							{		
		}_		1		 	 	{	 		<u> </u>	1.00	1 FAD		<b> </b>	1	<u> </u>		ł
1	OP						<u> </u>			<u> </u>	<u>[</u>			<u> </u>					
1	114		7.0	26.0	25.0	\$5.0	26.0	25.0	25.0	24.0	22.0	24.0	240	24.0	2:.0	23.0	3:0	2:.0	
	13	1	6.0	16.0	15.0	25.0	16.0	16.0	16.0	17.0	17.0	17.0	17:0	17.0	17.0	h7.c	<u>12.0</u>	19.0	
	1.2		8.0	19.0	17.0	18.0	17.0	15.0	10.0	10.0	17	21.0	21.0	2.0	10.0	h9:0	21.0	21.0	
~ /		T,	0.0	2.0	25.0	<b>B</b> .0	2.0	30.0	32.0	////	11,	33.0	11 0	22.0	122 0	22.0		34.0	3
	1								12.00	777	17	1,55.0	55.0	135.0			22.0	:	
	10	$\frac{1}{1}$	£.0	1 27.0	27.0	<u>7.0</u>	125.0	x <u>25.0</u>	132.0	123.0	134.0	133.0	33.0	132.0	<u>0,[5]</u>	<u>132.0</u> 1	13.0	132.0	Г Г
	9	1	7.0	16.0	16.0	<u>þ7.0</u>	18.0	17.0	19.0	20.0	21.0	킨.0	21.0	19.0	15.0	<u>hε.o</u>	12.0	19.0	 
	"8	1	7.0	16.0	16.0	<b>p6.0</b>	17.0	17.0	18.0	19.0	20.0	21.0	21.0	20.0	19.0	15.0	17.0	19.0	i
	7	1	7.0	16.0	<u>16.0</u>	.7.0	17.0	18.0	19.0	20.0	20.0	21.0	20.0	19.0	18.0	17.0	18.0	18.0	
	6	21	5.0	17.0	<b>17.0</b>	7.0	17.0	18.0	19.0	20.0	21.0	20.0	20.0	19.0	19.0	19.0	17.0	18.0	
	ĸ	11	3.0	16.0	17.0	17.0	18.0	19.0	20.0	20.0	20.0	20.0	x, o,	19.0	19.0	17.0	17.0	18.0	
	2				16.0	2.0	18.0	19.0	20.0	21 0	21 0	21 .0	21 0		10 0		18.0	17 0	
	7				10.0		10.0	17.0	20.0		2.0	2.0	<u> </u>	20.0	17.0	47.0	10.0	11.0	
	3	19	.0	18.0	17.0	19.0	20.0	20:0	21.0	22.0	22.0	22.0	22.0	21.0	<u>20.0</u>	<u>9.0</u>	18.C	عبعد	
	2	18		17.0	17.0	18.0	19.0	20.0	21.0	22.0	22.0	22.0	21.0	20.0	19.0	1 <del>7</del> .0	10.0	معد	
ړ	-	15	i.d	15.0	15.0	15.0	16.0	17.0	17.0	17.0	17.0	16.0	15.0	12.0	12.0	1.0	15.0	15.0	
-				2.	3	4	5	6	•7 •	-8	9	10	ນ	12	13	Ц	15	16	
•	7 1				DAAIN	P-1	2												
	Y		- <u> </u>	– B							~ •								

.

.

## FIGURE 2.11.4-7 GAMMA SCAN READINGS IN MR OF 1-13C II CASK WITH 9 CURIES COBALT 60 SOURCE (continued)

													•				
								•		T	1		T	1	T	]:	7
		]									1.						1
	}	]		1		]	1	LIF	~~~	PAD				1	1	1	1
				1			1		1.	]	Ī				1	1	1
ц	29.0	0 30.0	34.0	33.0	33.0	30.0	27.0	26.1	26.0	27.0	26.0	25.0	25.0	25.0	25.0	, ,	
13	20.1	d 20.0	0 19.0	19.0	19.0	18.0	217.0	27.0	17.0	17.0	17.0	17.0	17.0	17.0	18.0		]
12	21.0	)20.0	.o	.9.0	19.0	18.0	170	117.0	15.0	18.0	17.0	17.0	17.0	19:0	19.0		]
	34.0	35.0	24.0	\$3.0	32.0	32.0	23.0	23.0	32.0	32.0	\$1.0	32.0	30.0	30.0	31.0		
	31.0	31.0	31.0	\$2.0	32.0	32.0	23 6	////, 22.5	31.0	31.0	31.0	31.0	30.0	30.0	30.0	:	1
9	18.0	18.0	με.ο	\$7.0	17.0	17.0	17.0	17.0	17.0	17.0	17.0	17.0	17.0	29.0	15.0		T
8	19.0	18.0	17.0	17.0	17.0	16.0	16.0	16.0	16.0	17.0	17.0	16.0	17.0	17.0	17.0		]
7	19.0	17.0	17.0	h7.0	16.0	16.0	15.0	·/// 15.0	16.c	17.0	17.0	17.0	17.0	18.0	P		]
6	17.0	17.0	16.0	16.0	15.0	15.0	ц.0	11.c	1///	16.0	17.0	17.0	17.0	15.0	15.0		] .
5	17.0	17.0	16.0	15.0	15.q	15.0	15.0	15.0	15.0	16.0	17.0	16.0	17.0	18.0	19.0		
,	17.0	16.0	15.0	15.0	ц.q	ц.0	ц.0	ц.0	• 15.0	15.0	15.0	16.0	16.0	17.0	19.0	•	
•	27.0	16.0	15.0	15.0	13.0	14.0	15.0	<u>и.</u> о	и.0	и.o	15.0	15.0	16.0	17.0	19.0		
ر	17.0	16.0	15.0		14.0		14.0	17.0	13.0	14.0	27.0	15.0	16.0	7.0	18.0	·	
2						100							10.0				
	17	15.0	19.0	20	<u>13.u</u> 21	22	23 .	24	25	26	<u>2.0</u> 27	28	29	<b>3</b> 0	31		











2-200



OBLIQUE IMPACT Degress indicate turn required to restore bolt preload torque.



.

.

FIGURE 2.11.4-9

	•		· == •		• .					
	· · ·	<u>EIRST: TMA</u>	367:27		* ~. *	ふへいやうと	بعثا ومبسده	م کام خطی ا		· · · · ·
ļ	-		CALIA	Nation	·Frac	D Was	side	71=	13334	URFIN
		·····		それていい	ما يعتر منه .		1.21	1221	34601	'allin
	:	-بيب سي والمره	المالي المراجعة المراجع	. منهمه و د و بی			IL	 	2000 10 12 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	20,000
1		المراجع وجودية عالي المراجع	····	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	Ters		دريند م ( مرب	مند مان داند مانتی ا	يتبع الأزون ما	
4	•	·	•••••						r +. 4. 5	7. 5.
3	•	ارد و جربهم باد مدد ا		· · · · · · · · · · · · · · · · · · ·	37		7:1.	7 ****?	۱۵ بروز مدور	
-7	-	والمجموعة والمطار الجروانية		7.1 TYPE !!	41.25	··· • • • • • • • • • • • • • • • • • •	154 6			4 : :
		· · · · · · · · · · · · · · · · · · ·			wro ti		·	***. • * * *		
3			1. * *	· · · • • • • · · · ·	. • • .	1: n.	J	· ··: •	~~~	= { == ;- ;
ġ		<u>ما المراجعة ما الميلو حمد المد سور</u>	م بيند و و منه من م	Jere and a start	رسند	).001.	~~~~	c	,010	sec-
ğ			1Hz	) with the			·	·: ·: *: *1.	· · · · · · · · · · · · ·	~
-	1	14111.4 1	1.1.1.	-1-17-1	<b>[</b> ]: <b>[</b> '	1 .1 17	:1.11	1.211	1 1.1	•
Ц	11						-11 -11	•    •		·    ·    ·
ž		_11.11.11-11.11	. . .	~11 -11 -11			:11 - 11		11.11	•11 •11
- 7	_1]	•   •   •   •		-   -   -			11.11			•    •    •
- P							11.11		$\Pi$ $\Pi$ $\Pi$	11 11
7	!!	1 1 1 1 1 1 1 1 1	4 1 1 1 1 1					1111		. 11 1.1
		11111111		1.1.1.1	11.1	1114	11.11			1111
		1 1.1.1.1.1						111		1111
		• • • • • • • • •								
					IL Ed	rldel 4	Ubsh		744	
	1.					II III	1117	iIIT	111.9	11117
				11 11 11 1		<u>4      : </u>	I-ILI			111.7
	: •			11 11.11	1 10 11	VIII	·    ·		11 11	111/1
ł			11 11 11	11 11 . 11 1		4.1		1-11-11		11-12-1
	•	Y Y Y Y	IY Y Y	Y . Y . Y I	171-1	-V . V	NU	1 1 1	1 1 -	VZVEV
	•	1.1. 1. 0.23		···	· .; is ;	·· · · · · · · · · · · · · ·	7.5	···· ·· ·· ·· ·· ·· ·· ·· ·· ·· ·· ·· ·	بو د *=٠	11:1
			allieft starting							
	-	FORCE	Washe	× #2-	S MA	X +ORL	-=6	1000	HUN -	
	-		1	• • • • • • • • •		1/2			• • • • • •	-10,030
		· mintulus si		•	17	hu.	W. Jak	A AN	• • • •	4
			الدونيد مبلد سه	A THE ALL AND A THE ALL AND A			-1/		14, 15	Y
		••,*	•	· · · · · · · · · · · · · · · · · · ·	#1	TAVA Z	EERO'	7 .	11:20	1.06
			ī					<b>y</b>		
		····	••	· · · · · · · · · · · · · · · · · · ·	\$27	RVEFO	ROIT	£ ·· ·	•••••••	ور: ۱۰۰ تار
			نحيعنما	Sile and in	معيت		12			مزبنداج
	•			- <u>e</u> . è	· · · · · · · · · · · · · · · · · · ·	:- 5-	••	• • •		



.



FIGURE 2.11.4-9 (continued)

- Finisti Tripact Test to	مر بسر بسر المراجم الت المراجم	- יאדר בראי בין אין בראי	الندا تشاعكما
	المالية المرادسين مدا	· · · · · · · · · · · · · · · · · · ·	المراح وبالمسلم
المرجع ا	باشتاد منامرك الاحالة تسعا	ייד גיל אביני גייוידי איד איד איד	[145].K.H
: ٢٠ ٢٠ ١٠ ٢٠ ٢٠ ٢٠ ٢٠ ٢٠ ٢٠ ٢٠ ٢٠ ٢٠ ٢٠ ٢٠ ٢٠ ٢٠	· Laber Tertinter and i		Jules . 1. 1. 1. 2. 4.
ومداري محالية مرود والمرود المرود والمرود والمرود والمرود المرود المرو	57.5. a K E.	いいないといいにエナン	177 55777
6	· ?	·	4-1-121-1-12
Times and the end of the state of the state of the		·	
こうちょうないのでに、うい あちょうしいろいないないないで	באי אישריאווירבבויי	الحدويات أمي المنحود المنامد	->*:
- 11, 5 = A. + C. 212 + + + + + + + + + + + + + + + + + +		الاستين و الاسماع الذي المنتقب <u>الم</u>	1×2
	···		· · · · · · · · · · · · · · · · · · ·
A line - The miles and the sector with the miles of	this never to here to		1
			ा ग न न
	╎┊╴┧╴┧╴┥╴┥╶┥╴┥╴┥	<u>╎</u> ╎ ╎ ╎ ╎ ╎ ╎   	╫┽┽┽┼┼┼
<u>┥╷╷╷╷╷╷╷╷╷╷╷╷╷╷╷╷╷╷╷╷╷</u>	╟╾╏┇╺┨┧╼╏╎╾╏╎		
┝╾┼┼╤┼┼╾┼┼╼┼┼╼┼┼╼┼┼╼┼┼╼┼┼	╟╾┧┟╾┧┟╾┨┟╴┨╽╸╎┟╸┼		
الله المالية في الله المدينة في المولية في المربعة في المالية			
يتحدا وليجنبك فرابع بالحدال وبالبيالية المعال مبالستين النهر		THE PARTY NEWS TO	7.7.16 4.4
<u> </u>	· · · · · · · · · · · · · · · · · · ·		<u> </u>
	provershin) - (-	1 - ··· · · · · · · · · · · · · · · · ·	1. A
	·	1. 2 7 1. 1	·····
	no lin novi valvi	I FOR	re_mas
- The approximation of the second state of the			hur and the
HI TAVE FELOT	··· glool of Tox	avelare 2	4.7
P2 Take Zero			
		۲۰۱۱ و ۲۱۱۱ میلاده امور ملسوم مرجوع وارد م ۱۰ و ۲۱۱۱ میلاد موجود موجود مرجوع وارد م	····





#### 2.11.5 INTERPRETATION, CORRELATION, AND CONCLUSIONS

This section briefly summarizes significant comments and conclusions. Where appropriate, correlations and meaningful interpretations are provided.

#### 2.11.5.1 Cask Behavior

The drop tests conclusively demonstrated the capability of the 1-13C II cask to survive the 30 foot drop provisions of the hypothetical accident segments of CFR 71. Successful survival of two sequential 30 foot drops demonstrates the high degree of conservatism embodied in the 1-13C II design. Shield integrity was directly demonstrated by appropriate performance tests. Structural integrity was similarly demonstrated by the unchanged condition of the cask following these two drops, with the exception as previously noted.

#### 2.11.5.1.1 Shield Interrity

The post test gamma scan described in Section 2.11.4.4 demonstrates that the lead shield auffered no damage or change during two successive drop tests. There was no detectable lead settlement. All provisions of 10 CFR 71 relative to survival of shielding under hypothetical accident conditions are fully satisfied.

#### 2.11.5.1.2 <u>Weld Interrity</u>

Pre- and post-test dye penetration examinations fully demonstrated sufficient weld integrity to satisfy hypothetical accident conditions as defined in 10 CFR 71. However, there was one cracked lid bolt boss which occurred because of interference between bolt force measurement transducers and the overpack tiedown plate. Under normal shipping conditions, this interference situation would not prevail since force washers would not be present. Moreover, the diameter of the clearance area of the overpack has been increased from 3 inches to 3.548 inches.







## 2.11.5.1.3 Structural Integrity

Dimensional surveys before and after drop test show no change in cask geometry, see Paragraph 2.11.4.3. Diametrical measurements were taken at two outside locations (top and bottom) and three inside locations within the cask cavity or containment vessel (top, middle and bottom). None of these pre- and post-test measurements differed by more than the 1/8" measurement accuracy. Data fits, based on the very conservative assumption that any measurement differences were true physical changes, show that maximum possible strain cannot exceed:

> Bending .23% Level 'A', Figure 2.11.4-4 Membrane .19%

This is well below the 40% ultimate strain of the stainless steel cask materials, and well below the permissable strains set for cask requalification following the test, see Paragraph 6.0 of reference 2.11.2.2. Thus, the cask meets the strain level requirements for return to service.

Closure bolt strains under two successive 30' drops were conservatively determined using a "turn of the bolt" method with "turn" values extracted from Figure 2.11.4-3, Maximum probable closure bolt strains are as follows:

Bolt Nr. (Fig. 2.11.4-9)	Tyrn ()	Strain (%)
5	25 <sup>0</sup>	.44
10,7,11,6	10 <sup>0</sup>	.18
8	5 <sup>0</sup>	.09

"Strain,  $c = \Delta t/t$ ;  $1t^{n} - 7tNC \times 2t^{n}$  $\Delta t = \frac{\binom{0}{360}}{\frac{1}{360}} \cdot \frac{1}{7}$ ;  $t = 2.25 (1^{n}+1d)$ 





With one exception, bolt #5, all closure bolts experienced strains less than the definition of engineering yield, 0.2% offset. The maximum possible strain in bolt #5 is just slightly more than this value. These predictions are very conservative since a major fraction of the retorque applied is probably related to the instantaneous release of friction forces upon impact.

Additional very useful data regarding closure bolt performance can be extracted from the end drop bolt force transducer #1 output, Figure 2.11.4-9. Essentially no hig frequency response components may be observed. Additional frequency decomposition analyses of transducer data confirm this conclusion. Thus, for the 1-13C II package, bolt dynamic response effects due to wave propagation effects in the cask body, may be safely ignored. The primary reason for the absence of elastic body dynamic responses within the cask may be attributed to the decoupling of excitation and response afforded by the energy absorbing overpacks.





#### 2.11.5.2 Overpack Behavior

Where not directly measured or assessed by test, structural integrity of the package is demonstrated using stress values based upon analytic predictions or overpack performance. This section presents a comparison of test and analysis demonstrating a high degree of agreement.

The end drop test provides an ideal vehicle to compare the prediction accuracy of overpack analyses since the geometry details are minimized as are complications arising from rotational motions during impact. Figure 2.11.5.2-1 represents a pair of high speed motion picture photos at the instant of initial impact and at maximum overpack deformation. Total test deformation, as measured from these photos is 4.46 inches. Corresponding analytic predictions, assuming a 50% effectiveness for unbacked foam, amounted to 5.5 inches. These values differ by approximately 23%.

Careful examination of data and comparison with probable causes leads to the conclusion that these modest differences relate to "effectivess" assumptions for unbacked foam regions, i.e., that portion of the impact cross section that is not backed by the cask body. This unbacked "over-hang" clearly is capable of resisting loads up to some value where the foam fails in shear. Examinations of the post-test results for the impacted overpack show a shear failure of the foam along the bond surface of the inner overpack shell. This is matched by tensile ruptures of the overpack shell as shown in photos F14 and F15, Section 2.11.3. Thus, the unbacked foam generates a significant portion of the total impact force.





FIGURE 2.11.5.2-1



CASK AT INSTANT OF IMPACT



CASK AT INSTANT OF MAXIMUM DEFORMATION





To assess this effectiveness of unbacked regions, a series of end drop analysis (EYDROP), see Section 2.10.2.1, were performed on varying effective impact areas. Results are as follows:

Diameter (in.)	$\frac{\lambda rea}{(in^2)}$	Deflection (in)	
40.	1257	7.96	
42.5	1419	7.29	
45.	1590	6.68	
47.5	1772	6.13	
50.	1963	5.63	
52.5	2165	5.19	
55.	2376	4.79	
57.5	2597	4.44	
60.	2827	4.13	
57.36"	2584	4.46 (See next s	heet)

The total end area of the overpack is 2584  $in^2$  (60" 0.D.). The end area backed by the cask body is 1257  $in^2$  (40" 0.D.). Thus, the unbacked area of partial effectiveness is:

$$2827 - 1257 = 1570 \text{ in}^2$$



Revision O

...



# 1-13C II OVERPACK END DROP ANALYSIS 16.5 PCF FOAM

## IETOKOP(ENI)

•

PACKAGE VEIGHT		27000.	(1)5)
PACKAGE DIAMETER	æ	57.31	(1x)
OVERPACK DEPTH		15.00	(IX)
DROP HEISHT		30.10	(FT)

		++++ Inta	6T ++++	*****	EXERGI ++++	**	
Ckush							
DEPTH	STRAIX	FORCE	ACCEL.	XIXETIC	SIRAIN	KATIO	
(1)		(L)\$)	(2)	(1x-L3)	(1X-L3)	(SE/KE)	
.50	.033	1312561.	4 <b>8</b> .8	1713504	279282		
1.00	.017	2285804.	84.7	\$747000	1710714		
1.50	.100	2454889.	40.9	\$740500	7115100	-110	
2.00	.133	2511278.	\$3.0	\$774000	3454851	- 4 - 7 - 7 - 7 - 7 - 7 - 7 - 7 - 7 - 7	
2.50	-147	2508414.		\$787500	4011074	.3/4	
3.00	.200	2480279.	11.0	9901000	1150710	106	
3.50	.213	2509120.	97.4	9814500	7104400	.6/8	
.46 1.00	.247	2533049.	93.4	9R2R000.	P147787	./	
4.50	.300	2584093.	15.7	9841500.	V117095		1.000
5.00	.333	244487.	98.7	9855000	11754730	1.1.7	
5.59	.347	2747851.	102.5	9848500	17117714	1.114	
4.00	.400	2894184.	107.2	<b>1882000</b> .	14077871	1.420	
4.50	.433	3037745.	112.5	1895500.	15510804	1.547	
7.00	.447	3232988.	119.7	9909000	17078489	1 794	
7.50	.500	3488524.	129.2	\$\$22500.	19258417	1.729	
8.00	.533	3798298.	140.7	931000	20190523	7 071	
B.50	.547	4171875.	154.5	1941300.	77573114	2.0/1	
7.00	. 400	4451348.	122.3	9943000.	24278927	7 487	
7.50	.433	5309833.	194.2	9974500.	27210222	2 71	
10.00		4252842.	231.4	9990000.	10159102	2.733	
10.50	.700	7545552.	279.5	10003500.	33484507	1 1.4	
11.00	.731	1147052.	338.8	10012000.	17747458	3.300	
11.50	.717	11510380.	424.3	10030500.	47947614	J.//4	
12.00	. 100	14997740.	555.1	10044000	49571544	7.202	
12.50	. 113	19723895.	210.5	10057500	51010145		
13.00	.1.17	27145239.	1005.4	10021000	10011017	3.772	
13.50	. 100	37777740.	1107 1	10071000.	#7106083, 94185330	\$.74/ 0.5./	
14.00	.911	51054574	1991 0	10001300.		8.241	
•••••	. 947	44474511	14714V 9447 A	10078000.	149391255	10./33	
	1 000	83883038	434747 7116 F	10111200.0	1094.	13.821	
	1*448	BJIEJNKE.	4118*2	10132000	∽ 4479.	17.329	





Test deformation are matched with an effective end area of 2534  $in^2$  (57.36" 0.D.). The effective area of the unbacked region is thus calculated as:

$$2584 - 1257 = 1327 \text{ in}^2$$

The effectiveness ratio for this unbacked region is:

Effective 
$$= \frac{1327}{= 84.5\%}$$
  
1570

An alternate approach is the assessment of unbacked foam effectiveness would assume that full section effective up to the point where shear failures of the foam no longer permits additional load transfer. For the 1-13C II cask overpack, the shear plane shown in Photos F14 and F15 is idealized as follows:







Allowable shear stresses for 16.5  $1b/ft^3$  foam, as provided by suppliers, are estimated as:

...

$$\tau = 13.27 p^{1.2492} = 440 psi$$
  
 $F_{max} = (440)(25)(\pi)(40) = 1.383 \times 10^6 lbs.$ 

The area of unbacked impact cross section is

$$\lambda_{\rm u} = \frac{\pi}{4} (60^2 - 40^2) = 1570 \, {\rm in}^2$$

Maximum crush stress in unbacked regions is:

$$\sigma = \frac{F_{max}}{\lambda_{11}} = \frac{1.383 \times 10^6}{1570} = 880 \text{ psi}$$

The stress in backed regions at a deflection of 4.46 inches is found as approximately 1000 psi. Thus, the effectiveness of unbacked foam is estimated as:

The lesser of these two unbacked foam effectiveness values has been employed for all side, end and oblique orientation cask load analyses used in this report. This adjustment corrects the analytic "over prediction" of overpack deformations observed in test measurement.



## 2.12 ATTACHMENT

INFORMATION IN MICROFICHE FORM

- 2.10.3 Finite Element Stress Analysis (1-13C II Cask)
- 2.11 Appendix: Drop Test Report Reference Documents (1-13C II)

Appendix 2.10.4 Lift Lug Analysis - August 1981.





## 3.0 THERMAL EVALUATION

This chapter identifies and describes the principal thermal engineering design . aspects of the 1-13C II package important to safety and to compliance with the performance requirements of 10 CFR 71.

## 3.1 Discussion

The CNS 1-13C II package is designed with a totally passive thermal system. The principal physical characteristics of this thermal system consist of an external thermal fire shield surrounding a cylindrical stainless steel shell. The 1/2 inch cylindrical stainless steel shell surrounds a lead shield of approximately 5 inch thickness. The inner surface of the package consist of an 26-1/2 inch diameter stainless steel shell. The cylindrical package ends are similarly constructed of stainless steel and lead materials. These ends are enclosed in, and surrounded by, cylindrical foam impact absorbers (also called overpacks) that effectively serve as adiabatic boundaries for these end surfaces. The above components are identified on the drawings as noted in Appendis 2.10.1

The principal operating features of the 1-13C II package thermal system are as follows:

- 1. Normal Transport Conditions
  - a. The decay heat of the payload is transferred to the inner surface of the package by radial radial heat transfer means.
  - b. This decay heat is transferred to the exterior of the cylindrical stainless steel shell by radial conductive heat transfer through the lead shield.
  - c. Next, this decay heat is transferred to the exterior fire shield by radiant heat transfer.









- d. The fire shield dissipated this decay heat to the ambient environment by combined radiation and convection heat transfer.
- e. Essentially no heat is disspated through package ends. The foam impact energy absorbers, enclosing these package ends, enforce abiabatic boundary conditions on all end surfaces.

## 2. <u>Hypothetical Thermal Accident</u>

- a. The heat of a hypothetical fire is imposed upon the exterior fire shield by radiant heat transfer means, as prescribed in 10 CFR 71.
- b. This fire heat is transferred to the stainless steel cylindrical shell by radiant heat transfer.
- c. Conductive heat transfer conveys this heat to the lead shield.
- d. Both the lead shielding material and the steel end assemblies store heat during the period of the fire.
- e. Subsequent to the fire, the stored fire heat is dissipatd to the ambient environment as described, above, for normal transport conditions.
- f. During fire exposure portions of the foam end impact absorbers char but retain a cellular structure to trap a dead air void of equivalent dimensions.

The import maximum temperatures of the 1-13C II package for both normal transport and hypothetical thermal accident conditions are summarized below.



-- ....

## Normal Transport Conditions

0

	TEMPERATUR	<u> </u>
	<u>Cask Bedy</u>	Fire Shield
100°F Ambient Air*	190.6°F	156.3 <sup>9</sup> F
130°F Ambient Air*	214.4°F	182.1°F
-40°F Ambient Air/	81.5°F	37.3°7

### <u>Hypothetical Fire Accident</u>

	Node	<u>Temperature</u> F
٩	•	
Fire Shield	9	1174.0°F
Lift Lug	2	508.7°F
Side Outer Shell	10	434.5°F
Top Outer Shell	3	363.4°F
Bottom Outer Shell	16	365.1°7
Case Cavity	22	371.4 <sup>0</sup> F

\*Includes heating due to solar insolation. With maximum decay heat load = 800 watts.

The important conclusions derived from the above results include:

The lead shield does not melt under hypothetical thermal accident conditions (maximum lead temperature =  $508.7^{\circ}$ F at lift lug; melt =  $621^{\circ}$ F).









The components of the closure systems (bolts, seals) at both ends are exposed to temperatures not in excess of  $363.4^{\circ}F$ . Silicon seals retain excellent sealing properties up to  $400^{\circ}F$ . The maximum predicted temperature of the pressurized containment cavity is  $371.4^{\circ}F$ . This temperature is used to derive maximum containment vessel pressure magnitudes.

The package has been evaluated for payload decay heats ranging up to 800 watts. Maximum temperatures reported above correspond to the 800 watt case. Specific steady state (at  $100^{\circ}$ F ambient air) and hypothetical accident transient analyses have been performed for decay heat values of: 150, 200, 300, 400, 500, 600, 700, and 800 watts.





<sup>\*</sup>Shappert, L.B., Cask Designers Guide, Oak Rigde National Laboratory, ORNL-NSIC-68, 1968, Table 2.3, Page 32.

Revision J

#### 3.2 Summary of Thermal Properties of Materials

Metallic thermal material properties used for analyses were taken directly from Shappert, "Cask Designer's Guide - A Guide for the Design, Fabrication and Operation of Shipping Casks for Nuclear Applications", ORML-NSIC-68, page 166. These properties are as follows:

		Stainless	Solid
Property	Symbol	<u>Steel</u>	Lead
Thermal Conductivity (Btu/hr-ft - F	k	11.	18.6
Specific Heat Capacity	C <sub>p</sub>	.125	.0325
Density (1b/ft <sup>3</sup> )	ρ	485	687

Surface emissivities and absorptivities, used in radiant heat transfer calculations, were taken from a variety of sources, as later defined. The specific analysis assumptions are tabulated below:

## o Source Model and Fire Shield Exterior

Properties for the external surface of the cask and the source are taken directly from requirements defined in 10 GFR 71 such that the heat input to the package is not less than that which would result from exposure of the whole package to a radiation environment of  $1,475^{\circ}F$  for 30 minutes with an emissivity coefficient of 0.9 assuming the surfaces of the package have an absorption coefficient of 0.8. Thus, the exterior surface of the package fire shield is assumed to possess an emissivity and absorptivity of 0.8. The emissive power of the hypothetical accident source is assumed equivalent to a grey body with emissivity of 0.9 at a temperature of  $1475^{\circ}F$ . This is equivalent to a black body source with a temperature of  $1424.71^{\circ}F$  as shown in the following calculations:



3-5





$$T_{o} = \left[ \left( .9 \right)^{4} \right]^{1/4}$$
$$T_{o} = \left[ \left( .9 \right) \left( 1475 + 459 \cdot 69 \right)^{4} \right]^{1/4} - 459 \cdot 69$$

 $T_0 = 1424.71^{\circ}F$ 

#### o <u>Fire Shield Interior and Cask Exterior</u>

Both bodies are 304 stainless steel and possess emissivity properties varying with temperature. An empirical equation of the form:

 $E = A + BT^{C}$ 

was fitted to available data from three sources. This fit found coefficients of:

$$A = .430$$
  
 $B = 3.1359 \times 10^{-5}$   
 $C = 1.2850$ 







Numerical evaluation at varying temperatures gives values used in the analyses:

<u>Emissivity c</u>
.43
.46
.50
.54
.60
.65
.71
.74
.78
.80

The data sources used for this empirical fit were as follows:

(1) Kreith, Frank, Radiation Heat Transfer.

International Textbook Co., 1962, Table 2-5. °E Type 301 96 .44 638 .50 1213 .86 1710 .90 Type 316 98 .49 665 .52 1193 .61 1575 .89



(



(2) Holman, J.P., <u>Heat Transfer</u>, McGraw-Hill Book Company, 1963, Table A-10.

.

	Ē	<u>c</u>
Type 301	450	.54
	1725	.63

(3) Baumeister, Theodore, <u>Mark's Standard Handbook for Mechanical</u> <u>Engineers.</u> McGrav-Hill Book Co. Eighth Edition, 1978, Table 2, Page 4-73.

		°E	<u></u>
Type	304	42	8.62
		98	6.73

## 3.3 Technical Specifications of Components

Not Applicable.

#### 3.4 Thermal Evaluation for Normal Conditions of Transport

The thermal evaluation of the 1-13C II package to the Normal Conditions of Transport, defined in 10 CFR 71, has been performed by analytic means. Eight alternate payload decay heat configurations have been considered, as described in Section 3.1.

#### 3.4.1 Thermal Model

A common thermal analysis model was used for both steady state evaluation of the normal conditions of transport and transient evaluation of the




hypothetical thermal accident. The common thermal model consists of a 22 node lumped parameter axisymmetric idealization. Heat transfer is represented by 38 resistor elements of both linear (conduction) and non-linear (radiation and convection) form. Figure 3.4.1-1 depicts this thermal idealization.

The general characteristics of this model may be summarized as:

- o The model consists of three zones two end zones and one mid-body zone.
- Each zone is comprised of 5 lead shield nodes and two stainless steel nodes representing inner and outer shells. Within a zone these nodes are connected by 6 conduction resistors ( $R_7$  to  $R_{7a}$ ).
- o Coupling between each zone is achieved by a total of twelve conduction resistors ( $\mathbb{R}_{25}$  to  $\mathbb{R}_{36}$ ).
- The outer shell is enclosed by a fire shield comprised of a  $1/4^{-1}$ stainless steel plate offset from the cask outer shell by a 16 gauge (.065") wire spacer at 6" centers. This fire shield is thermally linked to the cask outer shell by a radiation resistor ( $R_{_{A}}$ ).
- o The fire shield is penetrated by a pair of lug pads attaching to the lifting ring integrally on the cask outer shell. Conductive resistors coupling this lug pad to the fire shield  $(R_{38})$ , outer shell  $(R_5)$ , and lead  $(R_{37})$  are provided.













3-10



- c The ambient external environment is represented by a single node held at constant temperature, or at programmed temperatures versus time for the hypothetical thermal accident. Every external node of the 1-13C II package is linked to this external environment node by a pair of resistors - one for radiation, one for free convection  $(R_1 \text{ to } R_4)$ . During the hypothetical thermal accident, the convection resistors are "switched off".
- Solutions are achieved by using a conventional thermal analyzer program, THAN, based on the well-known Lockheed Thermal Analyzer.

The separate steady-state conditions were evaluated: eight to establish initial conditions for the hypothetical thermal acident analyses per NRC Regulatory Guide 7.8, paragraph C.l.a (all run at an ambient environmental temperature of  $100^{\circ}$ F): one at the "hot" conditions specified by 10 CFR 71 ( $130^{\circ}$ F) for Normal Transport and one at the "cold" conditions (-40°F) specified for Normal Transport.

The derivation of model properties follows:







# 1. THERMAL CARACITANCE

C<sub>i</sub> = pVCp

NODE (i)	MATERIAL	VOLUME (in <sup>3</sup> )		C <sub>i</sub>
2 3 9 10 16 22	Steel	$(5.5)(8)(2)(2) = 176 \text{ in}^{3}$ $\pi(3/4)(19.25)^{2} = 873.1 \text{ in}$ $2\pi (19.565)(1/4)(68.63)=2$ $2\pi (1/2)(19)(68.6) = 4094$ $\pi(1/2)(19.25)^{2} = 582.1 \text{ in}$ $2\pi (1/2)(13.25)^{2} + 2\pi(55)$	3 109.2 in <sup>3</sup> .8 in <sup>3</sup> .3 (13.5)(5)	6.17 30.63 74.00 143.66 20.42 101.19
4 5 6 7 8	Sclid Lead	$V = \pi R_j^2 t; t = 5.75/5$ $R_j = 13.75 + t(j-1)$ j = i-3	(in <sup>3</sup> ) 683.1 802.1 930.7 1068.8 1216.5	8.83 10.36 12.03 13.81 15.72
11 12 13 14 15		$V_{j} = \pi (R_{j+1}^{2} - R_{j}^{2}) \ell_{j}$ $\ell_{j} = 55 + 2(j-1)t$ $t = 5^{*}/5$ $R_{j} = 13.75 + (j-1)t$ $j = i-10$	5103.5 5653.3 6228.2 6828.3 7453.4	65.94 73.05 80.47 88.23 96.31
17 18 19 20 21		Vj = πR <sub>j</sub> <sup>2</sup> t; t = 6/5 Rj = 13.75 + t (j-1) j = i-16	712.75 842.6 983.3 1134.8 1297.2	9.21 10.89 12.70 14.66 16.76



٠

.





# 2. END CONDUCTION RESISTORS (R7 - R12 and R19 - R24)



RESISTOR	NODE	Ai (in)	LEAD RESISTOR	STEEL RESISTOR	TCTAL RESISTOR
	3, 16	1104.47	585 13-6	740,79-6	1.3259-3
7,19	4, 17	1008,85	642.25-6		
a 21	5, 18	916.49	708.20-6		
10.22	6, 19	829.58	784.33-6		
11 22	7, 20	747.	874.00-6		
	8, 21	667.83	980.00-6	919.34-6	1.0983-3
	22	593.96			



Ĩ

\*Assume Bottom Identical



3. MODEL PROPERTIES - SIDE CONDUCTION RESISTORS (R13-R18)

Side - Through the thickness -



 $\frac{\text{Rij} = \frac{\ln (r_{\eta}/r_{i})}{2\pi K i}}$ 



		RADIUS	LENGTH	LEAD	STEEL	TOTAL
RESISTOR	NODE	(in.)	(in.)	PESISTORS	RESISTORS	RESISTORS
	10	19.00	67.38			
13				76.809-6	67.814-6	144.62-6
	I1	18.08	65.32			
14				82.487-6		
	12	17.17	63.25			
15				90.882-6		
	13	16.25	61.19			
16				99.474-6		
	14	15.33	59.13			
17				108.16-6		
	15	14.42	57.06			
18				120.82-6	116.93-6	237.75-6
	22	13.50	55.00			
1						





(4) COUPLING CONDUCTION RESISTORS - (R25-R36)



RESISTOR	NOD	ES D	RADIUS I	HALF HT. h	MATERIAL	RESISTOR
25, 26 27, 28 29, 30 31, 32 33, 34	3 4 5 6 7	10 11 12 13 14	19.00 18.08 17.17 16.25 15.33	33.69 32.66 31.63 30.60 29.57	Steel Lead	574.71×10 <sup>-3</sup> 283.97×10 <sup>-3</sup> 283.03-3 282.32-3 281.82-3
35. 36	8	15	14.42	28.53	*	281.45-3







5. EXTERNAL RADIATION TO SIDES (R2 and R4)

 $\sigma = .1714 \times 10^{-8}$ Kij = JAE; E = .8 (10 CFR 71) The exposed gross area of the shield is:  $A_{A} = 2 R \Omega$  R = 19.565 in. <sup>4</sup> 1 = 68.63 =2(10) = 48.63 in.  $A_{\Delta} = (2)(19.565)(48.63)(144) - 41.515 \text{ ft.}^2$ The exposed area of the pad is:  $A_2 = (5.5)(8)(2) = 0.611 \text{ Ft.}^2$ 144 The net area of the fire shield is:  $A_{4 \text{ net}} = 41.515 - 0.611 = 40.904 \text{ in.}^2$ The external radiation coefficients are:  $K_{A} = (.1714 \times 10^{-8})(.8)(41.515 - .611) = 56.088 \times 10^{-9}$  $K_2 = (.1714 \times 10^{-3})(.8)(.611) = 837.80 \times 10^{-12}$ 



3-16



6. FIRE SHIELD TO OUTER SHELL RADIATION (R6)

$$\begin{array}{l} \text{Xij} = \frac{c\text{Ai}}{\left(\frac{1}{\epsilon_{1}}-1\right)^{-}\frac{1}{Fij}} + \frac{\text{Ai}}{\lambda j} \left(\frac{1}{\epsilon_{j}}-1\right)^{-} \frac{1}{j = \text{Node 10}} \\ \text{i} = 2\pi (19.315) \quad (68.63)/144 = 57.418 \ \text{ft.}^{2} \\ \text{Aj} = 2\pi (19.565) \quad (68.63)/144 = 58.528 \ \text{ft.}^{2} \\ \text{c}_{i} = .52; \quad \text{c}_{j} = .71 \ (\text{interpolated from Section 3.2}) \\ \text{Therefore: Kij} = 42.334 \times 10^{-9} \\ \end{array}$$

7. CONVECTION RESISTORS (R1 and R3)

The film coefficient, h, depends upon (Gr Pr.) Assume  $T = 130^{\circ} r$   $Tw = 170^{\circ} r$   $GR = \left(\frac{2^2 c_s}{u^2}\right)(Tw - T = ) L^3$   $= (1.305)(40)\left(\frac{68.63}{12}\right)^3 \times 10^6 = 9.7649 \times 10^9$  Pr = .72, see Kreith Principals of Heat Transfer, 3rd Edition  $(GrPr) = \frac{7 \times 10^9}{100}$ 

Therefore; the flow turbulent and we may use McAdams recommendations:

 $\frac{1/3}{Vert Cyl: h = .19\Delta T}$  (sides) The effective convection areas are: Node Area (ft<sup>2</sup>) 2 0.611 9 40.904







.

.

8. CONDUCTION RESISTANCE TERMS FOR THE LUG PAD

· .





R38 =	(1/4"	Fire	Shield)	=	22.576	X	10	 
								_

to Lead
t = (1/12)(19.00-18085)= .035
K = 18.6
A = .611
$\frac{.035}{1.000} = \frac{3.0797 \times 10^{-3}}{.0000}$



•

.



9. SCLAR INSOLATION LOADS (Nodes 2 and 9)

Per NRC Regulatory Guide 7.8, Table #1, the solar insolation load applied to vertical curved surfaces is 1475 Btu/ft<sup>3</sup> per 12 hour period.

The projected gross exposed area of the thermal shield is:

 $P_{9gross} = (68.63-20)(2)(19.565)/144$ = 13.215 ft.<sup>2</sup> The projected area of the lug pad is:  $A_2 = .611/2 = 0.306$  ft.<sup>2</sup> The net projected area of the fire shield is:  $A_{9net} = 13.215 - .306 = 12.910$  ft.<sup>2</sup>

The solar loads are:

 $q_2 = (.306)(1475/12)/3.41 = 11.01$  watts

 $q_0 = (12.910) (1475/12)/3.41 = 465.20$  watts



## 3.4.2 <u>Maximum Temperatures</u>

Maximum steady state temperatures, under normal conditions of transport, for selected locations in the package are shown in Table 3.4.2-1. Essentially all locations within the package, excepting the fire shield, exhibit near constant temperatures, for a given condition. It can also be observed that the predicted temperatures are a nearly linear function of applied decay heat. Detailed analysis results are found in Tables 3.4.2-2 to 3.4.2-10.







•



		AMBIENT						
CASE	DECAY HEAT (WATTS)	AIR TEHP. (F.) (NODE 1)	FIRE SHIELD (NODE 9)	LIFT LUG (NODE 2)	SIDE Outer Shell (Node 10)	TOP OUTER SHELL (NODE 3)	CAVITY (NODE 22)	
1	150	100	131.20	137.26	138.00	138.33	138.35	3.4.2-2
2 3	200		133.28 137.35	141.24	142.23 150.49	142.67 151.16	142.70	- 3
4	400		141.32	156.60	158.53	159.42	159.47	- 5
5	500		145.17	163.96	166.35	167.46	167.51	- 6
6	600		148.96	171.16	173.98	175.32	175.39	-7
7	700	♥	152.66	178.16	181.42	182.97	183.06	- U
8	800	100	156.27	185.00	188.67	190.45	190.55	-9
9	800	130	182.09	209.11	. 212.55	214.34	214.43	-10

\*Includes Solar Insolation per NRC Regulatory Guide 7.8.







150 WATT DECAY HEAT, 100 ° F AMBIENT AIR

STEADY STATE PROBLEM

TOTAL NUMBER OF NEWTON ITERATIONS =8NUMBER OF TEMPERATURE DEPENDENT INTERPOLATIONS =4NO. OF NEWTON ITERATIONS FOR FINAL UPDATE =1

CLASS 2 - TEMPERATURE, T

ID	DLGPEES F	10	VEGREES F	10	DEGREES F	10	DEGREES F
1 5 9 13 17 21	100.00000 138.3375428 131.1**********************************	2 6 10 14 10 22	137.2513193 138.3373297 135.003=330 135.1772072 135.337547F	3 7 11 15 17	138.3357349 136.341.424 135.0410248 135.2337003 138.3393247	4 5 12 16 70	1 38. 3 365007 1 3 3. 3452 5 6 9 1 3 4. 0 6 2 4 4 9 2 1 3 4. 3 3 5 7 3 4 9 1 3 4. 3 4 1 8 9 2 4

i







 TABLE 3.4.2-3 : 1-13 C II NORMAL TRANSPORT .....'ERATURES

 200 WATT DECAY HEAT, 100 OF ANBIENT AIR

# STEADY STATE PROBLEM

Total Number of Newton Iterations =	8
Number of Temperature Dependent Interpolations =	4
No. of Newton Iterations for Final Update =	1

CLASS 2 - Temperature, T

• ---

10	DE FRENS E	ΙJ	0531::: F	t D	DEGR_ES F	tn	Decation
1 5 7 13 17 21	100.0020000 142.5763340 133.2005512 142.542518 142.542518 142.5480164	2 10 14 1r 22	141.2+1444. 142.5707217 142.2112441 142.452223 142.6763340 142.6773421	3 7 11 15 14	142.6734252 142.68213e7 142.2509326 142.5378431 142.67.7217	4 8 12 16 20	142.6749494 142.0166109 142.3360016 142.5739282 142.6021317







TABLE 3.4.2-4 : 1-13 CII NORMAL TRANSPORT .\_\_MPERATURES 300 WATT DECAY HEAT, 100 °F AMBIENT AIR

STEADY STATE PROBLEM

TOTAL NUMBER OF NEWTON ITERATIONS =8NUMBER OF TEMPERATURE DEPENDENT INTERPOLATIONS =4NO. OF NEWTON ITERATIONS FOR FINAL UPDATE =1

CLASS 2 -	TEMPERATURE, T	10		10			
10		10	Drukcij r	10	UCG4213 P	10	DEGREES P
1	100.0000000	2	142.0310249	3	151-1630035	4	151-1645368
5	151.1665223	6	151+1731+68	7	151.1753230	۲	151.1420412
9	137.1524714	10	150.4944184	11	150.5735931	12	150.6574416
13	170.7414464	14	150.8444510	15	150.4:84429	16	151.1630635
17	191.194 360	15	151.1666220	14	151.1701+60	20	1:1.1753230
<i>č</i> 1	151.1 20412	22	151,1921365				







TABLE 3.4.2-5 : 1-13 C 11 NORMAL TRANSPORT 1.... ERATURES

400 WAIT DECAY HEAT, 100 OF AMBIENT AIR

STEADY STATE PROBLEM

TOTAL NUMBER OF NEWTON ITERATIONS =	8
NUMBER OF TEMPERATURE DEPENDENT INTERPOLATIONS =	4
NO. OF NEWTON ITERATIONS FOR FINAL UPDATE =	1

CLASS 2 - TEMPERATURE, T

3-25

[D	D:026253 F	10	DEGREES F	. ID	DEGREES F	10	DÉGREES F
1	100.0000000	ζ	156.6300963	3	159.4202341	4	159.4222808
5	157.4229622	C	1:9.4278248	7	154.4356662	~	157.4456252
4	141.1125473	10	172.5330786	11	154.6343664	. 15	158.7461642
13	151.7451.130	1.5	157.0023521	15	159.1481047	16	157.4202341
17	157.4272dOH	10	157.4259622	19	159.4296248	20	157.4365662
21	151.5479252	22	1,2,4570814				







TAL 3.4.2-6: 1-13 C 11 NORMAL TRANSPORT TEMPERATUR.

500 WATT DECAY HEAT, 1000 F AMBIENT AIR

# STEADY STATE PROBLEM

.

TOTAL NUMBER OF NEWTON ITERATIONS = 5

NUMBER OF TEMPERATURE DEPENDENT INTERPOLATIONS = 3

NO. OF NEWTON ITERATIONS FOR FINAL UPDATE = 1

CLASS 2 - TEMPERATURE, T

10	υξύΡ·Lj F	10	Jeules, 1	::	DEGAEES F	10	JEGPEES F
1 5 13 17 21	177.023300 187.421-516 197.1717103 159.7997749 187.4924834 197.4979703	2 E 10 14 14 27	163.~0JJU43 167.4577728 184.347602 186.7342100 157.4517010 167.5144.22	J 7 11 15 [4	167.4353213 167.4764700 167.4754700 167.4736042 167.4677223	4 12 16 20	167.4524334 157.4276700 164.6133511 167.4559212 167.4764700







IBLE 3.4.2-7 : 1-13 C II NORMAL TRANSPORT TEMPE AES 600 WATT DECAY HEAT, 100° F AMBIENT AIR

STEADY STATE PROBLEM

TOTAL NUMBER OF NEWTON ITERATIONS =4NUMBER OF TEMPERATURE DEPENDENT INTERPOLATIONS =2NO. OF NEWTON ITERATIONS FOR FINAL UPDATE =1

CLASS 2 - TEMPERATURE, T

12 NEROSSE F tŋ N'GOFET F 10 DEGREES F 10 DEGREFS F 1 100.0000000 > 171.15+0155 5 175.3241116 3 175.3164542 4 175.3194359 \* 17:.3212670 7 175.3515258 9 145,0145516 8 173.082-145 175.3549691 10 174-1390075 13 11 174.4106015 174,3057034 14 176, + 20+231 12 17 15 174.9047012 172.3107350 175.3168542 14 1 t 175.3241116 21 19 171, 11, 66,441 175.3312470 20 175.3415254 22 175.3971614





ABLE 3.4.2-8 : 1-13 CII NORMAL TRANSPORT TEMPER, ... ES 700 WATT DECAY HEAT, 100 °F ANDIENT AIR

STEADY STATE PROBLEM

TOTAL NUMBER OF NEWTON ITERATIONS = 6

NUMBER OF TEMPERATURE DEPENDENT INTERPOLATIONS = 3

1

NO. OF NEWTON ITERATIONS FOR FINAL UPDATE =

CLASS 2 - TEMPERATURE, T

ID DELOTES E DEGRERS E In 10 DEGREES F 10 DEGREES F 1 100.0000.00 7 179.1631419 3 182. 3737595 182.9773546 4 5 182.0822283 5 142.4205745 7 193.0025495 8 18 3.0182342 0 157.6543126 10 181.4154/32 11 191.5405626 141.7742072 12 12 157.0014594 1-2.2474073 14 15 1=2.4977727 147.0737595 16 17 192.5773566 **1** H 1-2.0422243 13 142.9905785 22 193.0325446 21 1-7.0132342 22 183.0554071







TABLE 3.4.2-9 : 1-13CII NORMAL TI ... PORT TEMPERATURES 800 WATT DECAY HEAT, 100 OF AMBIENT AIR

STEADY STATE PROBLEM

TOTAL NUMBER OF NEWTON ITERATIONS = 8

NUMBER OF TEMPERATURE DEPENDENT INTERPOLATIONS = 4

NO. OF NEWTON ITERATIONS FOR FINAL UPDATE = 1

CLASS 2 - TEMPERATURE,T

ID	DEGREE. F	10	DFGR123 F	10	Despèrs F	10	DEGREES F
1	100.0000000	Z	132-0012411	3	170.4530011	4	170.4571140
5	179++526371	U	173.4722328	7	190.4859168	4	140.5030474
9	154.7645160	10	150.0629344	11	164.0813742	12	164.1049674
13	1 17.1504435	14	1-4.51-3376	15	104.9066655	16	170.4530011
17	197.4571148	1.	1+0.4525371	17	140.4722323	20	1+0.4059116
71	1-10.563-474	22	140.5457:65	-			





1

TABLE 3.4.2-10 : 1-13 CTT NORMAL TRANSPORT TEMPERATURES

BOO WATT DECAY HEAT, 130 VE AMBIENT AIR

•

STEADY STATE PROBLEM

.

TOTAL HUPMER OF NEWTON ITERATIONS .		8
NUMBER OF TEMPERATURE DEPENDENT INTERPOLATIONS	•	4
NO. OF NEWTON ITERATIONS FOR FINAL UPDATE -		1

. .

.

CLASS 2 - TEMPEPATURE, T

-10	REGPERS F	•10	DEGREES F	I D	DECKEE2 F	10	CEGREES F
1	130.0000000	2	709.1090136	3	214.3410132	4	214.3451503
5 ·	214.3507337	6	214.3602914	7	214.3739886	8	214.3919336
9	182.0957505	10	212.5477853	11	212.7695123	12	212.9931026
13	213.2345757	14	213.5064664	15	213.7970954	16	214,3410132
17	214.3451503	19	714.2507337	19	214.3602914	20	214.3739836
21	714.3910336	22	214.4349639				

.

### 3.4.3 <u>Minimum Temperatures</u>

The minimum temperatures of the 1-13C II package approach  $-40^{\circ}F$  since several of the planned payloads possessivery low decay heats. The stainless steel and lead materials of construction for the packaging are not significantly affected by an ambient temperature of  $-40^{\circ}F$ . For higher decay heats, these minimum temperatures do not exceed  $82^{\circ}F$ . Table 3.4.3-1 summarizes predicted temperatures for an 300 watt decay heat payload.

### 3.4.4 <u>Maximum Internal Pressures</u>

Internal pressure in the cask activity is calculated assuming:

- o The cask is loaded at a temperature of  $70^{\circ}$ F and is unpressurized.
- A small but sufficient quantity of free water is always present.
   This conservatively permits pressure calculations to be based on tabulated thermodynamic properties of saturated water and steam.











LE 3.4.3-1 : 1-13 C 11 NORMAL TRANSPORT TEMPER, 2S

800 WATT DECAY HEAT, -400 AMBIENT AIR

### STEADY STATE PECHLEM

TOTAL YUMAFR CF NEWTON ITLATIINS .	12
PINALD OF TEAPERATURE DEPENDENT INTERPOLATIONS .	5
NO. OF NEWTON ITEPATIONS FOR FINAL UPDATE .	1

### CLASS 2 - TENPERATURE, T

10	LEUDEEZ E	10	ntopics e	1 C	DEGREES F	10	CEGREES F
1	-40.000000	2	74.113738	. <b>1</b>	81.3793387	4	dl.3¤39329
5	41.34944AA	4	81.3049329	7	91.4125500	н	41.4304069
9	37.3291247	10	79.6485395	11	79.0075599	12	40.C31277E
13	FC.2747435	14	20.5446202	15	40.2353213	16	£1.3794397
17	P].3936329	14	41.2494486	19	41.3449329	20	11.4125500
21	AJ, LOCLONG	22	-1.4731874				

.

o The cask cavity serves as the pressure boundary of the system; conservatively no pressure retention capability is assumed for the payload container (liner, drum, etc.)

Pressure calculations are based upon the appropriate partial pressures of air and water. At the time of loading, the total pressure is the sum of partial pressures for air and water and is equal to atmospheric pressure:

$$T_{F} = P_{AF} + P_{WF}$$
(1)

Where:  $T_{p} = Total$  Pressure

 $P_{\mu\nu}$  = Partial pressure of air, at temperature, F

 $P_{up}$  = Partial pressure of water, at temperature, F

The partial pressure of air, at time of loading, is found as:

$$P_{AF} = T_F - P_{WF}$$
(2)

Once steady state equilibrium temperatures are established, the total and gauge pressures in the package are found as:

$$T_{X} = P_{AX} + P_{WX}$$
(3)

$$G_{\chi} = T_{\chi} - T_{F}$$
 (4)

Where: x = the steady state equilibrium temperature of the coolest portion of cask cavity; i.e., the condensation surface.

 $G_{ij}$  = Gauge pressure of cask at temperature, x







The partial pressure of air at temperature, x, is calculated assuming adiabatic constant volume relations:

<u>PV</u> = constant T

$$P_{AX} = P_{AF} + \frac{(460 + x)}{(460 + F)}$$
 (5)

The partial pressures of water,  $P_{AF}$  and  $P_{AX}$ , are taken directly from the thermodynamic properties of saturated water and steam. The tabulated values developed by Joseph H. Keenan and Frederick G. Keyes, "Thermodynamic Properties of Steam", John Wiley & Sons, 1936, are reproduced below:



Temp. J	Aba Franca La Ac Ja	Temp., J I	<u>din Prom</u> ., Li Bajin, P
1111	8 On 54 8 On 65 8 37370 0 36737 6 37813	300 21 0 26 3 254 344	47 012 77 64 10 54 100 54 113 01
TERT	8 2563 6 36.1 6 66.0 9 9962 8 9962		134 63 143 04 173 17 193 77 230 37
110 110 140 140	1 7748 1 66.4 2 77.1 2 8424 3 718	610 610 610 610 610	247_21 174 73 304 61 372 72 341 84
	4 741 8 052 7 815 0 335 81,838	864 660 676 689	427 8 444 9 414 7 344 1 821 4
5-14 211 211 211 211	14 123 14 844 17 186 20 240 24 949	844 84 8 844 84 84 84 84 8	640 8 812 4 947 8 1133 1 1375 8
514 515 510 540 540	70 823 83 425 41 835 49 763 87,854	0	1447 9 • 1144 6 7034 7 1343 6 2106 1
		194 194 4	9083 7 3204 3







Substitution of equations (3), (5) and (2) into equation (4) gives a final expression for gauge pressures,  $G_{\chi}$ , in the package:

$$G_{\chi} = (T_{F} - P_{WF})(\frac{460 + \chi}{460 + \chi}) + P_{WX} - T_{F}$$
 (5)  
(460 + F)

For this evaluation the following values are constant:

 $F = 70^{\circ}F$ , temperature at fill

P<sub>WF</sub> = 0.3631 psi, partial pressure of water at loading temperature

 $T_{\rm F} = 14.7$  psi, at atmospheric pressure

Normal transport pressure results are summarized for the 1-13C II package, as follows:



Case No.	Decay Heat (Watts)	Air Temp. ( <sup>0</sup> F)	Solar Insol. Present	Cavity Temp. (°F)	Internal Pressure (psiç)
1	150	100	Yes	138.35	4.27
2	200		<b>A</b>	142.70	4.72
2	300		T	151.20	5.68
4	400			159.47	6.75
5	500			167.51	7.96
6	600			175.39	9.30
7	700	<b>V</b>		183.06	10.77
8	800	100	l V	190.55	12.36
9	800	130	Yes	214.43	19.00
10	800	-40	Ю	81.47	0.46
ļ					



Revision O

### 3.4.5 <u>Maximum Thermal Stresses</u>

Table 3.4.2-1 demonstrates that the cask body possesses less than a two degree gradient through the walls, under a maximum decay heat payload of 300 watts; thus, stresses due to differential radial thermal expansions are negligible. Comprehensive evaluation of pressure and axial differential thermal expansions are completely evaluated in Section 2.6.1 utilizing two axisymmetric finite element models for the cask body and lid. All stresses are found to be well within established limits for the materials of construction.

# 3.4.6 <u>Evaluation of Package Performance for Normal Conditions of</u> <u>Transport</u>

The thermal behavior of the 1-13C II package is completely consistent with the allowables for all materials of construction. In particular, the maximum predicted temperature of the payload cavity,  $214.43^{\circ}F$ , is well below the established service limit of  $400^{\circ}F$  for silicon seals.

Key findings used elsewhere in this report for detail stress analyses are summarized in Table 3.4.2-1.





### 3.5 <u>Hypothetical Thermal Accident Evaluation</u>

The thermal evaluation of the 1-13C II package for the hypothetical thermal accident defined in 10 GTR 71, has been performed analytically using essentially the same model as described in Section 3.4 for normal transport conditions. The eight payload decay heats, discussed in Section 3.4.2, have been thoroughly evaluated versus the requirements of 10 GTR 71.

#### 3.5.1 Thermal Model

The common thermal model, described in Section 3.4.1, has been used for the transient analyses associated with the Hypothetical Thermal Accident. Salient differences are as follows:

The ambient environment, node 1, is described by a tabular definition of temperature versus time. Up to a time of 0.5 hours this environment is  $1475^{\circ}$  F. Subsequent to this time, the ambient environment drops to  $70^{\circ}$ F.

Initial conditions correspond to the eight decay heats examined in Section 3.4.2 and defined in Table 3.4.2-1.

The external convection heat transfer modes became effective only at the termination of the 0.5 hour accident event. Three hours following the termination of the thermal accident, artificial cooling is introduced by increasing the free convection film coefficient by one hundred fold.

#### 3.5.2 Package Conditions and Environment

Free drop and puncture test damage do not measurably alter the thermal behavior of the 1-13C II package. Specifically, free drop damage only affects the geometry of the overpack. This fact has been demonstrated conclusively by test, see Section 2.11. Since the net affect of the







### 3.5.3 Package Temperatures

Particularly important component temperatures are summarized in Section 3.1. Table 3.5.3-1 summarizes extreme temperatures encountered for all decay heats examined. Figures 3.5.3-1 to -16 plot the temperature versus time response of key locations in the package.











,

# Thuse 3.5.3-1 Selected Maximum Temperatures for

Hypothetical Accident Conditions, 1-13 C 11

			MAXIMUM FIRE TEMPERATURES ( <sup>O</sup> F)					
Case I	Decay lleat (Hatts)	Fire Shield (Node 9)	Lift Lua (Node 2)	Side Outer Shell (Node 10)	Top Outer Shell (Node 3)	Cavity (Node 22)	Transient Results Figure }	
1	150	1170	463.60	384.23	313.48	320.42	3.5.3-1 to -2	
2	200	1170	467.37	388.44	317.63	324.67	-3 to -4	
3	300	1171	474.73	396.66	325.76	332.98	-5 t.o -6	
4	400	1171	481.89	404.64	333.66	341.06	-7 to -8	
5	500	1172	488.85	412.40	341.35	348.93	-9 to -10	
6	600	1173	495.65	419.99	348.88	356.62	-11 to -12	
7	700	1173	502.27	427.37	356.20	364.11	-13 to -14	
R	800	1174	508.73	434.56	363.35	371.43	-15 LO -16	
					]			

.

.

Sevision 0

۱

Ł















## FIGURE 3.5.3-3

•• • • • • • •

HYPOTHETICAL FIRE ACCIDENT









FIGURE 3.5.3-4

HYPOTHETICAL FIRE ACCIDENT





:



3-43

·

## FIGURE 3.5.3-5

# HYPOTHETICAL FIRE ACCIDENT







.
FIGURE 3.5.3-6









,

1

FIGURE 3.5.3-7



**@** {

FIGURE 3.5.3-8

HYPOTHETICAL FIRE ACCIDENT





# FIGURE 3.5.3-9

HYPOTHETICAL FIRE ACCIDENT





FIGURE 3.5.3-10

## HYPOTHETICAL FIRE ACCIDENT







#### FIGURE 3.5.3-11

#### HYPOTHETICAL FIRE ACCIDENT









Revision O

# FIGURE 3.5.3-12

### HYPOTHETICAL FIRE ACCIDENT





FIGURE 3.5.3-13

#### HYPOTHETICAL FIRE ACCIDENT







FIGURE 3.5.3-14

HYPOTHETICAL FIRE ACCIDENT







# FIGURE 3.5.3-15









Revision 0

# FIGURE 3.5.3-16

# HYPOTHETICAL FIRE ACCIDENT







# 3.5.4 <u>Maximum Internal Pressures</u>

The maximum internal pressures under hypothetical accident conditions are found using the methodology and assumptions discussed in Section 3.4.4. Accident pressure results are summarized for the 1-13C II package as follows:

6264	Decay	Cavity Tepp	Internal	
Number	(Watts)	(°F)	(psig)	
1	150	320.42	96.64	
2	200	324.67	102.45	
3	300	332.98	114.27	
4	400	341.06	126.75	
5	500	348.93	140.04	
6	600	556.62	154.21	
7	700	364.11	168.99	
8	800	371.43	184.37	

## 3.5.5 <u>Maximum Thermal Stresses</u>

Comprehensive evaluation of pressure and axial differential thermal expansions are completely evaluated in Section 2.7.3 utilizing two axisymmetric finite element models for the cask body and lid. All stresses are found to be well within established limits for the materials of construction.

To illustrate the relatively low magnitude of resultant stresses, a simple ANSYS axisymmetric finite element solution was performed for the eight decay heats of the payload. A quasi-plane-strain solution was employed maintaining a common axial deformation. By "quasi", we mean







axial distortion of the body was allowed but the "plane-section remained plane" assumption was invoked at every cut. The analysis conservatively utilizes maximum pressures as derived in Section 3.5.4 and maximum thermal gradients (about  $112-114^\circ$ ) as shown in Figures 3.5.3-1 to -15, although these do not occur at the same point in time. The model used was as follows:





Notes: -Nodes 1-8 and 11-18 are coupled in the vertical direction -Node 1 is supported vertically -Therfore, vertical displacement of nodes 1-8 is always zero, and verticle displacements of nodes 11-18 are always equal to

each

other.

Tables 3.5.3-1 to -8 present stress results for the eight decay heat cases. Note that the highest stress anywhere, 14,756 psi tension, occurs in the inner shell in the axial direction for the 800 watt load case.



.

.



.

.

XEU 111LE=       FIRE 11         ALAPPA ELEMENT SIRESES         EL=       1         IC,YC>       13.50         IL=       2         MONES=       2         IC,YC>       13.50         IL=       2         NONES=       2         IC,YC>       14.25         AL,YC=       14.25         IL=       NUNES=         XL,YC=       14.25         IL,YC=       14.25         IL,YC=       14.25         IL,YC=       14.25         IL,YC=       14.25         IL,YC=       18.25         IL,YC=       17.00         IL,YC=       17.00	Km#SIENI 150 VATIS         ***** TIME = 0.         11       12       2         12       14       141-         16MP= 276.9       3       14         18MP= 315.5       4       15       5         15       5       MAT=       1         16MP= 312.8       5       16       6         5       16       6       nAT=       1         16MP= 332.8       5       16       6       nAT=       2         17       7       MAT=       2       16       17       2         18       8       4       18       14       1       1         18       8       4       18       1       1       1	$i_{UMI} SIIF = 1$ $I_{ISTM} = i_{MONL} = v$ $S_{x,51,1} i_{x,52} = 137.45$ $I_{STM} = 1_{MONL} = 0$ $S_{x,51,1A1,52} = 330.00$ $I_{STM} = 1_{MONL} = 0$ $S_{x,51,1A1,52} = 320.10$ $I_{STM} = 1_{MONL} = 0$ $S_{x,51,1A1,52} = 320.10$ $I_{STM} = 1_{MONL} = 0$ $S_{x,51,1A1,52} = 271.57$ $I_{STM} = 1_{MONL} = 0$ $S_{x,51,1A1,52} = 210.60$ $I_{STM} = 1_{MONL} = 0$ $S_{x,51,1A1,52} = 210.60$ $I_{STM} = 1_{MONL} = 0$ $S_{x,51,1A1,52} = 210.60$ $I_{STM} = 1_{MONL} = 0$ $S_{x,51,1A1,52} = 10.60$	ITERATION- 1 14259. 567.81 132.22 -276.53 -0e0.90 -1024.1 - 555.2	CUM. 111.K .71737E-08 .42112E-09 .42112E-09 .47.67E-09 .66710E-07 .76567E-10 14347E-02	1 12123. 434.72 5.5743 -307.38 -793.20 -1004.13 -6644.2	TABLE 3.5.5-1 Pressure and Thermal : Stresses, 150 Vatts, :
					·	Induced 1–13C II

•

•

.

•







.

NEW TILLE=	FIRE TRAN	51ENT 200 VALLS						
**** ELEXENT S	TRESSES	+++ 11HE = v.		LOAN SILF= 2	TTERATION# 1	CUN. IICN.=	2	
EL= 1 KODES=	1 11	12 2 <del>1</del> ÅT=	١	ISTR= 1 HONE= 0				
16,1C= 13.5V	3000	IENP= 283.0		SX, SY, IXY, SZ= 133.37	14298.	.77718E-08	12515.	
LL= 2 KOPES=	z 12	1J 3 XAT=	2	ISTA= 1 MODE= 0				
XC,YC= 14.25	5000	16hr- 301.1		54,58,181,52= 355.7V	559.43	. 42112E-09	410.13	0 17
EL= J NODES=	3 13	=1AH 4 HAI=	2	ISTA= 1 HODE= v				- 11
xL,fC= 15.25	2000	12nP= 317.7		58,51,1x1,5Z= 348.80	123.83	497691-09	_93569	ເ ເ
EL= 4 NOPES=	4 14	15 5 MAT=	2	Isins   Huits O				- Ch
16.25	5000	1EMP= 337.1		58,58,1X8,52= 317.J1	-103.72	.57425E-04	- 57 5.117	-
EL= 5 HUDES=	5 15	16 6 MAT=	2	ISTA= I HODE= V		•		ся 0
IC,1C= 17.23	2000	1EX4= 722.4		SX,S1,TX1,SL= 268.87	-061.11	.020856-08	-708.11	0
EL= & NODES=	6 16	17 7 HAT=	2	151/1= 1 nUPL= 0				20
XC,YC= 10.25	5000	1281= 308.8		54,3+,127,52= 207.87	-1032.5		-11751 - 1	
EL* / NODES*	× 17	18 V AHI=	1	15t+++ + MOPE= 0				и -
XÜ,NL= 17.00	5000	TEMP= str.+		51,51,111,51= 11.417	-7512.8	1434/E-v/	-0.13.1	مر

•

.

TABLE 3.5.5-2 Pressure and Thermal Induced Stresses, 200 watts, 1-13C II

•

Revision 0

:

:

3-59

٠



E TRANSTENT 300 VALIS					
SES + + + + + 11ME = 0.		LOAP SILL -	TTERATION= - L	COM. LIEK.#	3
11 12 ∠ nAT=	1	151n- 1 NODE= 0			
1000 IEAP= cri.c		5x,51,1x1,5c= 128.49	14320.	.///IoE-UU	12647.
2 12 13 3 MA1=	2	ISYN= I NUVE= 0			
DUUD TEMP= JUT.4		5x,5Y,TXY,3x= 353.04	543.40	.385846-09	414,04
5 i 5 14 4 KAT=	2	isin= 1 HOPE= v			
1247-9 1EMP= 327.9		SX, Sr, 1X7, SZ= 345.05	107.80	. 192945-09	-10.544
=1Ah Z ti -F	2	ISTH= 1 MODE= 0			
1000 IEMP= 345.3		5%,31,1%Y,52= 313.74	-300.76	.:	-31141
5 15 14 6 n∺t÷	4	151n- i MDPE- 0			
0000 IEMP= 361.6		52, 31, 127, 52- 284.64	-085.14	.050022-07	-/18./3
5 16 17 / MAT=	2	ISIN= 1 NUVL= 0			
5000 TEMP= 322.0		5x,5Y,1XY,52= 203.34	-1048.5	0.	-1917_4
/ 1/ 18 0 ħAl≖	1	151N= 1 400E= 0			
1000 1EMP= 390.6		52,51.121,52= 89.053	-7420.2	i/934E-07	-1.161.4
	E TRANSTENT 300 VALTS SES ***** TIME = 0. 11 12 c nAT= 000 IEAP= cri.c 12 13 3 MAT= 000 TEMP= 307.4 13 14 4 MAT= 000 TEMP= 327.9 14 15 5 MAT= 000 TEMP= 345.3 15 14 6 nmT= 000 TEMP= 361.6 16 17 / MAT= 000 TEMP= 370.0	E TRANSTENT 300 VALTS SES ***** TIME = 0. 11 12 c nAT= 1 000 IEMP= cri.c 12 13 3 MAT= 2 000 TEMP= 307.4 13 14 4 MAT= 2 000 TEMP= 327.9 14 13 5 MAT= 2 000 TEMP= 345.3 15 14 6 nmt= c 000 TEMP= 361.6 16 17 / MAT= 2 000 TEMP= 377.0 17 TB B AAT= 1 000 TEMP= 370.6	E       TRANSTERT 300 VATTS         SES       ***** TIME = 0.       LOAP SILE = J         11       12 $c$ nAT = 1       ISTA = 1       MOVE = 0         000       IEAP = $crt.c$ ST, ST, ITT, Sc = 128.47         12       13       RAT = 2       ISTA = 1       MUVE = 0         000       TEMP = $30r.4$ ST, ST, ITT, Sc = 35J.04         13       14       RAT = 2       ISTA = 1       MUVE = 0         000       TEMP = $327.9$ ST, Sr, ITT, Sc = $343.35$ 14       15       AAT = 2       ISTA = 1       MOVE = 0         000       TEMP = $345.3$ ST, $31, 137, 52 = 343.35$ 15       14 $nm1 = 2$ ISTA = 1       MOVE = 0         000       TEMP = $345.3$ ST, $31, 137, 52 = 343.35$ 15       14 $nm1 = 2$ ISTA = 1       MOVE = 0         000       TEMP = $361.6$ ST, $31, 137, 52 = 264.67$ 16       17       MAT = 2       ISTA = 1       number = 0         000       TEMP = $377.0$ ST, $57, 137, 52 = 203.34$ 17       IB       RAT = 1       ISTM = 1       IDVE = 0         000       TEMP = $370.6$ ST, $57, 137, 52 = 87.053$ 57, 57, 137, 52 = 87.053       57, 57, 57, 57 = 57.57.57	E       TRANSTERT 300 VATTS         SES       ***** TIME = 0.       1.0AP SILE = J       TIFEATTONE         11       12 $c$ nAT = 1       1.5tn - 1       MODE = 0         000       TEMP = $cri.c$ $Sx, St, TXt, Sc = 128.47$ 14320.         12       13       3       MAT = 2       1.5tn = 1       MODE = 0         000       TEMP = $sur.4$ $Sx, St, TXt, sc = 353.04$ 543.40         13       14       4       MAT = 2       1.5tn = 1       MODE = 0         000       TEMP = $sur.4$ $Sx, St, Txt, Sz = 343.35$ 107.40         14       14       4       MAT = 2       1.5tn = 1       MODE = 0         000       TEMP = $345.3$ $Sx, st, st, st, st = 343.35$ 107.40         14       15       14.4       mat = 2       1.5tn = 1       MODE = 0         000       TEMP = $345.3$ $Sx, st, st, st = 343.45$ -300.78         15       14       6       mm = 2       1.5tn = 1       MODE = 0         000       TEMP = $361.6$ $Sx, st, st, st = 203.34$ -1048.5         16       17       MAT = 2       1.5tn = 1       mode = 0         000       TEMF = $3$	ETRANSTENT 300 UATTSSES*****TIME = 0.1112 $z$ 12131121311414151151161417141814191419141114111512131314141415161617171418151914191419141115111412151314141515141415151416171714181519141015121415161716181519141915111512161314141515141617171618151914191519161115121613161415151617181815171817<

•





. .

.

REV TITLE* ***** FLEMENT S	FIRE TRANS STRESSES ++	51EXI 400 UATIS ++ 11HE = 0.		LUME SILLS	11[HA[]])#= 1	[1]X. [1].k.=	4
LL= 1 HODES= 1C,fL= 1J.50	1 11	12 3 nA1= 1EMP= 299.3	1	51n= 1 ngl = 0 54,51,127,52= 116.65	14453.	.036941-08	12702.
EL= 2 KUHES- AC,YC= 14.25	2 12 5000	1] j dul* 1ENP* 317.5	:	1570= 1 MULE= V 52.57,127,52= 340.29	522.05	.382846-09	494.15
il= 3 XUHEs= xL,1C= 15.25	5 13 5000	14 4 Ani= 1EdF- JJJ.7	2	151H= + MUDE= 0 51,51,111,52= 330.86	y1.442	.421126-09	-16.9-1
L= 4 KOUES= KC.YC= 18.25	4 14 5000	15 5 dH1- IEXP= 353.5	٤	15182 1 NUILE V S1,S1,111,S22 307.13	-317.11	.612546-09	- 370 . 51
L= 5 HONES= L.1C= 1/.25	5 15 5000	16 6 MAI- 1EMP= 387.6	2	1518= 1 HONE= V 54,31,117,5/= 230.14	-201.69	.689106-09	-2241
L XODES=	1 Ib - 5000	17 / nAl= 1EXP= 385.0	2	151M= 1 HUDE= 0 51,51,121,52= 196.84	-1064.9	0.	-1025.1
LL= 7 KOUES= 11,1C= 19.00	7 17 5000	18 B MAI= 12NP= 3+0.8	1	\$1N=   NUT[= 0 56,54,177,52- 85.79]	- { }	1 + + 3 + E - 47	-8118.1

.

.

.....

Revision O

3-61

.





ALV ILILL?	FIRE TRANS	IERI SOO VALIS					
••••• £LENEX1 5	1×62262 +++	int v.		LOAF STEP+ 5	KATIU#=I	EUN. LIEK.⊨	
[L= 1 NJUES=	1 11	12 2 MAI=	1	ISTN= i MODE= 0			
#C,TC= 13.50	3000	IERT- JUZZO		51,51,111,51- 102.51	8در ۲۰.	. 1	13121.
{L= 2 X01/ES=	2 12	13 3 NAT=	2	15YA= + MDNE= 0			
10,10= 14.25	5000	ILAF = Jears		51,57,127,52= 341.31	512.02	. 493695-09	401.05
{L= 3 NODES=	3 13	14 4 dk1=	2	151K= 1 NDNE= 0			
XC,YC= 15.25	5000	TEMP= 343.7		51,51,121,512 333.25	26.157	. 43-408-09	-11.900
[L= 4 durts=	4 14	15 5 HAT=	2	ISTA= I ADHE= 0			
16.25		IEMP= 361.1		51,51,111,51= JUL.71	-332.69	.52425E-09	-312.61
EL= 5 NUNES=	5 15	16 6 MAI=	2	ISIN= I NOPE= 0			
XC, YE= 17.25	500v	1EXP= 377.4		5×,5×,1××,5x= 252.80	-717.Ji	.803, ji -09	-/ 31 . 13
Lt= + HU+ES=	δ 15	17 7 лні-	2	15YN= 1 NDNE= 0			
10,10= 18.25	5000	iEdr- sy2.8		5x,51,1x1,52= 191.39	-1041.0	.7050/2-10	-1032.5
{L= 7 XONES=	7 12	18 8 nA1=	1	ista= 1 node= 0			
10,10+ 19.00	5000	IENF= 496.4		34,51,117,52= 83.02b	-2272.4	12934E-07	-1211.5

•

•

TABLE 3.5.5-5 Pressure and Thermal Induced Stresses, 500 Vatts, 1-13C II

•

.

3-62

.



									TABL
XEN TITLE*	FINE TRAN	SIENT 600 VATTS							دی س
ANNAL LLENER	STRESSES ++	*** line = J.		LUAD STATE o	LIERATION:	1 (Un. 11).*.*			Š
	1 11	12 2 MA1*	1	ISTN= I HUDL= O					5
	5000	TEMP= 314.5	-	54, 31, 121, 52= 91.451	14とりる。	81644[-08	1133	10	5
	4 17	15 5 NAIT 1580-1177	2	ISIN- I KOPE= U				Ï	10
IL- J AUDES-	1 11		7	1914- 1 MDF- A	476.01	.34455E-09	124.43	n u	22
40,10+ 15.25	5000	1EKF= 351.7	-	58.57.1.e s/z 577.70		1071		5	4
EL= 4 NUPES=	4 14	15 5 841-	2		00.314	. 202045-44		ູ້	n
IC,1C= 10.25	3000	IENP= Job.o	•	54.31.127.52= 295.92	- 118.17	- 01/541 -119	-411 -3	6	8.7.0
1L+ 5 XOPES=	5 15	=IAir à âl	Z	151A= 1 AUUE = 0				8	- 1 I
xL,1C= 17.25	5000	1E46= 384.8		51,51,111,52= 246.80	-/32.25	. 60410[-09	-/36.03	5	달
il = & NONES-	6 15	17 7 MAL=	2	ISFA= 1 nuliz- ú		-		- F	Ï
10,10= 18.25	2000	IEMF = 400.3		5x,5t,1xt,52= 185.37	-1015.9	785870-10	-1038.2	i.	ца Ц
IL- / HUNEST	2 12	18 8 MAT=	1	ISTN= I AUDER V				⊢	н
10.10 19.00	2000	1EMP= 413.9		SX, S1, 141, 51* 80.011	-7184.1	167396-0/	-208.1.4	1	Ц
								С С	ц 0
								-	n n
							•		

•

.

Revision O

•

.

1000	
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	
VICE 7	



.



.

MEW TITLE: ***** ELEMENT S EL- i MOULS: IC,TC: 13.50 Li: 2 NOPES: IC,1L: 14.25 LL: 3 NOPES:	FIRE IKANS IMESSES ++++ I II 5000 2 I2 5444 3 I3	ERT 200 UATIS • TIME = 0. 12 2 MAI= 1 TEMP= 321.9 13 3 MAI= 2 TEMP= 340.1	LUAN SILF= 7 ISIN= 1 NUUL= 0 SA,SI,IXI,SZ= 07.078 ISIN= 1 NOUL= 0 SX,SY,IXY,SZ= 327.47	116441104± 1 14691. 480.95	CUM. 11±K.= .95650E-08 .497676-09	.: 13565. 11. 500.
IC,TC= 15.25 EL= 4 NODES=	5000 4 14	TEMP= 358.6 13 5 HAT= 2	52,57,121,52= 321.90 15+n- ; dulf= a	45.337	.42112E-09	-35.1.4
IL, IL- 18.25 IL= 5 X0HLS-	5 12	16 6 nAT= 2	SA, JI, FAI, SZ= 290.04 ISIN= 1 NUHL= V	-363.23	.302046-09	-104.47
	5000	TEMP= 377.3 17 7 MAT= 2	5x,5t,1xt,52= 340.94 15th= 1 nuit= 0	-242.01	. 489102-09	-2434
	- 1000	TERE= 402.2 18 8 NAT= 1	34,51,1X7,52= 177.JD 157M= i HUDE+ 0	-ltri.v	0.	-1013.8
		12AP= 421.3	SX,SY,1XY,SZ= 22.075	-7113.1	191308-07	-5462.1

TABLE 3.5.5-7 Pressure and Thermal Induced Stresses, 700 vatts, 1-13C II

3-64

Revision O

•

REV TITLE=	FIRE TRANS	TENT 600 VATTS				
***** ELENENI S	TRESSES +++	the slide + 0.	1.041 5161- 03	ITERATION= :	CUA. LIEK	¥
EL= 1 MUHES=	1 11	12 2 MAI- 1	151H= 1 AUDE= 0			
10,11= 13.50	5000	12AF= 329.1	51,51,111,52= 26.253	14756.	.83844E-08	131109.
EL= 2 HODES=	2 12	13 J HAT= 2	15TH= 1 MODE= 0			
10,10= 14.25	5000	1EMP= 347.3	58,51,121,52= 123.31	406.20	.421126-09	381.07
et- 3 HODES=	3 13	14 4 HAI+ 2	ISYN= 1 HOIE = 0			
10,10= 15.25	5000	IFWL= 702.A	52,51,121,51- 315.81	30.500	.53597E-0Y	-40.0.17
EL= 4 HODES=	4 14	15 5 HAT= 2	151X= 1 hUvE= 0			
10,10= 14.25	5000	1646= 387°5	SX, ST, T (T, 52= 284.v.)	-377.90	.042246-09	-414.41
1L= 5 MUHLS=	5 15	5 =1AK 2 - 31	151N= 1 NDFC= 0			
16,16= 17.25	jvv6	1541 = 177.5	SX, ST, IXT, Si= 234.98	-162.30	.64224L-09	-748.24
EL= & NODES=	6 16	17 7 HAT= 2	ISTN= 1 MURL= 0			
IL,YC= 18.25	5000	TEXP= 414.9	SX, S1, 1x1, 523 113.64	-1125.7	.7656/2-10	-1648.1
LLS 7 AUVES=	7 17	18 8 MAL- 1	ISTAT I AURET O			
40,TC= 14.00	5000	1EMP= 428.5	3x, 31, 1X7, 563 /4.112	-2040.2	lo/j+t-0/	-5247.5

•

TABLE 3.5.5-2 Pressure and Thermal Induced Stresses, 200 warts, 1-13C II

•

•

Revision 0

•

3-55

•

.

.

•



Both values are well below accepted allowables for the stainless steel inner and outer shells.

# 3.5.6 <u>Evaluation of Package Performance for Hypothetical Accident</u> <u>Thermal Conditions</u>

The thermal behavior of the 1-13C II package is completely consistent with the allowables for all materials of construction. In particular, the maximum predicted temperature of the payload cavity,  $363.4^{\circ}F$ , is well below the established service limit of  $400^{\circ}F$  for silicon seals.

Key findings used elsewhere in this report for detailed stress analysis are summarized in Tables 3.5.3-1 and in Section 3.5.4.





# 3.5 Accendix

The thermal analysis presented in Section 3.0 shows the adequacy of the cask to meet a heat load of SCO W. This will not be exceeded by using the general debris containers or the shielded debris containers. Since the shielded debris containers can contain a greater quantity of fissile material than the general debris containers, the shielded debris containers will be used in the calculation of an average heat load. The following calculation snow that expected heat loads are far less than this value.

o Total TMI - II core decay heat load ≤ 15.000 H

- o Total fuel loading 177 assembly x <u>463 Kg</u> = 81.951 KgU assembly
- Maximum container fuel content 60KgU Average heat load = 15,000 H x <u>50 KgU</u> = 10.98 H << BOOW 81,951





#### A.O. CONTAINMENT

This chapter delineates the containment configuration of the 1-13C II package for normal transport and hypothetical accident conditions.

#### 4.1 CONTAINMENT BOUNDARY

### 4.1.1 <u>Containment Vessel</u>

The package containment vessel is defined as the inner shell of the shielded transport cask, together with the associated lid seals and lid closure bolts. The inner shell of the cask, or containment vessel, consists of a right circular cylinder of 26-1/2 inches inner diameter and 54 inches inside height. The shell is fabricated of 1/2 inch thick stainless steel plate, Type 304 (ASTM A-240). At the base, the cylindrical shell is attached to a circular end plate with full penetration welds. At the top, similar construction techniques are used for the removable lid. The lid is attached to the cask body with twelve equally spaced 1-1/4 inch bolts in a 35.38 inches diameter bolt circle. See Section 4.1.4 for closure details.

### 4.1.2 <u>Containment Penetration</u>

There are two penetrations of the containment vessel, a drain line and a vent line. At the base, the drain line consists of a 1/2 inch 0.D. by 0.065 inch wall stainless steel tube gravity line from the center of cavity bottom to the side of the outer shell near the cask bottom. At the top, a comparable vent/test line exists to the base of the recessed lid step. Both penetrations are sealed with silicone, one piece molded in place seals with the rubber scaling element mechanically locked to a stainless steel retainer. Both seals are protected from fire exposure by the "adiabatic" end overpacks. The fasteners installed in both lines are provided with pressure relief features.







# 4.1.3 <u>Velds and Seals</u>

The containment vessel is fabricated using full penetration groove wells. All weld configurations are designed and fabricated to the intent of Section III of the ASME Boiler and Pressure Vessel Code. Seals are described in Sections 4.1.2 and 4.1.4.

#### 4.1.4 <u>Closure</u>

The top closure consists of a shielded lid, with tapered sidewalls to assist in positioning and sealing. The lid is supported on the outermost top circular plate by a circumferential step. This step confines a flat solid high temperature silicone seal. A solid silicone O-ring also exists near the edge of the bottom of the lid. The step prevents overcompression of the flat gasket and O-ring by the closure bolt preload forces and hypothetical accident impact forces. The lid is attached to the cask body by twelve (12) equally spaced 1-1/4 inch - 7 UNC x 2-1/4 inch long bolts. These bolts are fabricated of ASTM A-354, Grade BD material. These bolts are torqued to 270 ft-lbs.  $\pm$  10% (lubricated) or 360 ft-lbs.  $\pm$  10% (dry). The lid bolt torque calculations are found in Appendix 4.4.3, p. 4-25.

The vent and drain penetrations are sealed with Parker Stat-O-Seals (applicable catalogue excerpts may be found in Appendix 4.4.4, p. 4-30) which are used beneath the heads of th S.S. 3/8 - 16 UNG Hex H.D. cap screws at both locations. Overcompression of the silicone is prevented because the silicone rubber sealing element is molded and mechanically locked into a stainless steel retainer. The vent and drain screws are torqued to 12 ft-lbs. lubricated (i.e. 144 in.-lbs.) which is well above the 80 in-lbs. minimum torque requirements for sealing and well below the 220 in-lbs. maximum torque for crushing the retainer.





# 4.2 Requirements for Normal Conditions of Transport

#### 4.2.1 <u>Containment of Radioactive Material</u>

The 1-13C II cask is designed to assure no release of radioactive material in excess of limits prescribed in N.R.C. Regulatory Guide 7.4, "Leakage Tests on Packages for the Shipment of Radioactive Materials", under normal conditions of transport.

The 1-13C II package is designed to accommodate a variety of payloads with differing contents possessing a range of normal condition temperatures and pressures, each associated with a differing decay heat value.

Figure 4.2.1-1 defines limitations on dissolved or suspended radioactive materials in residual fluids which may be present in the package containment cavity for devatered resins. These limitations are defined as a function of payload decay heat. Appendix 4.5 summarizes the derivation and application of this limit relation for normal conditions. Provided these limits are not exceeded, compliance of the 1-13C II package with the requirements of N.R.C. Regulatory Guide 7.4 is assured. In accordance with the Regulatory Position, Paragraph C of this guide, compliance of the 1-13C II package with the requirements of Section 71.51 of 10 GFR 71, for "no loss or dispersal of radioactive contents, as demonstrated to a sensitivity of  $10^{-6}\lambda_2$  per hour", Paragraph 71.51 (a)(1), is demonstrated.

#### 4.2.2 Pressurization of Containment Vessel

Section 3.4.4 summarizes normal condition temperatures and pressures within the containment vessel. The predicted pressure values conservatively assume a sufficient quantity of free water is present; thus all pressure predictions are based upon the properties of saturated water and steam. These conservative predictions of pressure and associated temperatures are used to evaluate integrity of the 1-13C II package. None of these conditions reduce the effectiveness of the package containment.





	FIGURE 4 P1	
	NORMAL TOKOTTION JIS ON DISSULVED DAT	
	HAP UACI I YE HATER FALS HOUL CHSJ (-13)	
		<b>)</b> <sup>2</sup> /1, <u>1</u> , <u>1</u>
	ILLAHME / OR	
	WURAAL COMDITIONS	
<u>n                                      </u>		
10 <sup>-4</sup> (cn <sup>3</sup> )		$\vec{\mathbf{A}}_{-} = [MAX H MAX ACT Y TY OF A RAD OACT Y TY OF A RAD OACT Y TY OE A TY TY OE A TY TY TY OE A TY TY OE TY TY OE A TY TY$
<u>                                      </u>		MIXTURE PERMITTED IN A TYPE A T
		++++
<u>┥┥┥</u>		┽╁┊┟╍┽┥┊╷┊┑┊┥╎┝╎╷┥╿┑╸┢╵┼┥╽║║

.

# 4.2.3 <u>Containment Criterion</u>

The containment criteria of section 4.2.1 are satisfied through the use of leak test procedures for fabrication verification, periodic, annual and seal replacement for 1-13C II Transport Cask. Procedures are based on ANSI N14.5-1977, American National Standard for leak tests on packages for shipment of radioactive materials.

#### 4.3 CONTAINMENT REQUIREMENTS FOR THE HYPOTHETICAL ACCIDENT CONDITIONS

The following is an assessment of the packaging containment under hypothetical accident conditions as a result of the analyses performed in Chapters 2.0 and 3.0 above. In summary, the containment vessel was not affected by these tests (refer to Section 2.7).

#### 4.3.1 <u>Fission Gas Products</u>

There are no fission gas products present. Any fission products contained in the fissile debris has previously been released.

#### 4.3.2 <u>Containment of Radioactive Materials</u>

The 1-13C II cask is designed to assure no release of radioactive material in excess of limits prescribed in N.R.C. Regulatory Guide 7.4, "Leakage Tests on Packages for Shipment of Radioactive Materials", under hypothetical accident conditions.

The 1-13G II package is designed to accommodate a variety of payloads with differing contents possessing a range of accident condition temperatures and pressures, each associated with a differing decay heat value.







Figure 4.3.2-1 defines limitations on suspended radioactive materials in residual steam which may be present in the package containment cavity. These limitations are defined as a function of payload decay heat. Appendix 4.5 summarizes the derivation and application of this limit relation for accident conditions. Provided these limits are not exceeded, compliance of the 1-13C II package with the requirements of N.R.C. Regulatory Guide 7.4 is assured. In accordance with the Regulatory Position, Paragraph C of this guide, compliance of the 1-13C II package with the requirements of Section 71.51 (a)(2) of 10 CFR 71, for no release of radioactive material from the containment vessel exceeding specified limits is demonstrated.









# 4.3.3 <u>Containment Criterion</u>

A design verification leak test was performed according to Section 7.4.1 No leakage was detected at a temperature of  $59^{\circ}F$  and cask charge pressure of 12.0 psig, using a leak detector calibrated for detecting leaks in excess of 0.21 oz/yr of dichlorodiflouromethane (Freon-12) at 25<sup>°</sup>C and 1 ATM pressure.

The result of the design verification leak test serves two purposes;

 it provides a means to determine permissible activity concentrations of the leakable medias for normal and accident conditions of transport.

(i.e., 
$$\frac{C_{N}}{A_{2}}$$
 and  $\frac{C_{\lambda}}{A_{2}}$ ; Re: Section 4.4)

(2) it established acceptance criteria for verifying leaktightness upon fabrication of new packages, after the third use of the package, annually thereafter, and any time that the flat gasket 0-ring or seals may need to be replaced.

## 4.3.3.1 Problem Description

Typically halogen detectors are designed to identify the presence of leaks in excess of some threshold value. These thresholds are generally calibrated for detecting leakage of dichlorodiflouromethane (Freon-12) gas in terms of oz/yr.

For the 1-13C II cask, permissible leak rate limits are specified for normal and accident conditions in  $atm-cm^3/sec$ , standard conditions per ANSI 14.5. These conditions imply the presence of a 1 atmosphere pressure differential at a temperature of  $25^{\circ}C$ .

To test the adequacy of containment, the threshold values of the sensor must be equated to the prescribed leak rate limits for normal and accident conditions. This may be achieved by proper selection of the Freon (R-12) charge pressure applied to the cask cavity.





# Solution Method

The threshold sensitivity of the sensor is given in (cz/yr), a weight per unit time. Thi, must be converted to a volumetric flow (equivalent to the sensor threshold) becomes a variable expressed in terms of both pressure and temperature.

The prescribed leakage limits for the cask are stated in terms of standard conditions. This, in effect, defines the permissible size of the leak orifice. If pressure and temperature vary, during test, from standard conditions, the acceptable test leak limit differs from the standard condition value. Thus the permissible leak rate is also a variable of both test pressure and temperature.

If sensor threshold, expressed in volumetric terms, is equated to test leak rate limits, an equation is formed in terms of pressure and temperature. For each value of temperature, a charge pressure value can be found that sets this detector threshold at the prescribed leakage limits.

## Detector Threshold

In fps units, the detector threshold may be expressed as:  $I_H = C_1 i_H$ Where:  $I_H$  = detector threshold in lbs/sec  $i_H$  = detector threshold in oz/year  $C_1 = \frac{1 \ 1b_1}{1 \ 2ear} = 1.9819 \ x \ 10^{-9} lb_yr/sec_{z}$ 16 oz. (3600)(24)(365)sec



A 2 :

The volumetric equivalent of this threshold is:

$$\frac{L_{V}}{M_{H}} = \frac{1_{H}}{(M_{H}/V)}$$

Where:  $L_v = detector$  threshold in ft<sup>3</sup>/sec ( $M_R^{/V}$ ) = specific concentration of R-12 per unit cavity volume in lbs/ft<sup>3</sup>

The specific concentration of trace gas,  $(M_R/V)$  is obtained by use of the perfect gas law considering the partial pressures of trace gas and air within the cask cavity. The total pressure in the cask cavity is:

$$P_{T} = P_{R} + P_{A} \qquad (1bs/ft^{2})$$
Where:  $P_{T} = \text{total pressure, } 1bs/ft^{2} \text{ (absolute)}$ 

$$P_{R} = \text{partial pressure of } R-12$$

$$P_{A} = \text{partial pressure of air}$$

$$= (14.7) (144) = 2116.8 \ 1b/ft^{2}$$
From the perfect gas law<sup>1</sup>:
$$P_{R} = \binom{M_{R}}{\sqrt{V}} \cdot R_{R} T_{R}$$
Where:  $T_{R} = \text{temperature, } ^{O}R$ 

$$R_{R} = \text{gas constant} = 1545.32^{(1)} = 12.78$$

$$M_{R} = 1b \text{ molecular vt. of } R-12 = 120.93^{(2)}$$
Thus:  $P_{T} = \binom{M_{R}}{\sqrt{V}} R_{R}^{T} + P_{A}$ 
and, the specific concentration of trace gas is found
$$\frac{M_{R}}{V} = \frac{P_{A} (p_{T} - 1)}{R_{P} T_{P}}$$

Where:  $p_T = \frac{P_T}{\frac{P_T}{P_A}}$  the total pressure expressed in atm.



Now, in fps units the volumetric equivalent of detector sensitivity is:

$$L_{v} = \frac{1_{H}}{(M_{R}/V)} \frac{C_{1}i_{H}R_{R}T_{R}}{P_{A}(P_{T}-1)} ft^{3}/sec$$

This can be expressed in S. I. units as:

$$L_{\rm H} = C_2 \cdot L_V$$
 cm<sup>3</sup>/sec  
Where:  $C_2 = (1728 \text{ in } ^3/\text{ft}^3) (2.54 \text{ cm/in})^3 = 28316.85 \text{ cm}^3/\text{ft}^3$ 

Finally, the detector sensitivity may be expressed as:

$$L_{H} = \frac{C_{3} \cdot I_{H}T_{K}}{(p_{T}-1)} \qquad \text{cm}^{3}/\text{sec} \qquad \text{Eq. (A)}$$
Where:  

$$C_{3} = \frac{C_{1} \cdot C_{2} \cdot R_{R} \cdot (9/5)}{P_{A}}$$

$$= \frac{(1.9819 \times 10^{-9}) (28316.85) (12.78) (9/5)}{2116.8}$$

$$= 6.0988 \times 10^{-7}$$

$$T_{K} = \text{temperature, } {}^{0}X$$

$$(T_{R} = 9/5 \cdot T_{K})$$

As a check, standard conditions are inserted:

$$(p_T -1) = 1$$
 atm  
 $T_K = 25^{\circ}C + 273 = 298^{\circ}K$   
Then,  $L_H = (6.0988 \times 10^{-7})(298) \times i_H = 1.8 \times 10^{-4} i_H \text{ cm}^3/\text{sec}$ 





This compares precisely with the value shown on page 40 of the G.E. Leak Detector Manual, ID-48163

References:

C.L. Brown, "Basic Thermodynamics", McGraw Hill, 1951, p. 80
 (2) ASHRAE Handbook of Fundamentals, Table 2, Chapter 15, 1967

# Permissible Leak Rates at Test Pressures

Equation B5 of ANSI 14.5 provides a means to translate test conditions to Standard conditions:

$$L_{t} = L_{s}n_{s}(P_{u}^{2} - P_{d}^{2})t = n_{t}(P_{u}^{2} - P_{d}^{2})s$$

Where: subscripts s and t refer to "standard" and "test", respectively. subscripts u and d refer to "upstream" and "downstream" respectively. n = viscosity of gas at temperature, in centipoises P = pressure in atmospheres

> $L_t$  = leak rates in cm<sup>3</sup>/sec.  $\theta$  test conditions ( $L_{c_1}$  = standard permissible leak rate

For standard conditions:

 $n_{g} \approx .0185 \text{ cP} (\text{air at } 25^{\circ}\text{C})$   $P_{us} \approx 1 \text{ atm}$  $P_{ds} \approx 1 \times 10^{-2} \text{ atm}$ 

For test conditions:

$$n_t = .0126 cP^{(3)}$$
 (R-12 at 80°F)  
 $P_{ut} = n_r$ , atm.  
 $P_{dr} = 1$ , atm







Eq. (3)

Substituting, the permissible leak rate at test conditions is found as:

$$L_{z} = C_{4}L_{5} (p^{2}-1)$$
where:  $C_{4} = \frac{.0185}{0.0126 (1-1x10^{-4})} = 1.4634$ 

Reference:

(3) ASHRAE Handbook of Fundamentals, Table 4, Chapter 15, 1967.

Equations (A) and (B) represent the detector threshold flow rate and permissible cask test leak rate, respectively. Equating these two gives:

 $L_{H} = L_{T}$ C<sub>3</sub>i<sub>H</sub>T<sub>K</sub> =  $C_4 L_2 (P_T^2 - 1)$ (P<sub>1</sub>-1)

The expression for charge pressure is found as:

$$P_T^3 - P_T^2 - P_T + 1 = \frac{C_3}{C_4} \frac{i_H T_K}{L_s} = 0$$
 Eq. (C)

A leak test was performed for design leaktightness verification. No leakage was detected at a temperature of  $59^{\circ}F$  (288°K) and a charge pressure of 12 psig (1.816 atm) using a leak detector calibrated for detecting leaks in excess of 0.21 oz/yr. Solving equation (c) for the equivalent leak rate of air at Standard conditions:

$$L_{S} = \begin{pmatrix} C_{3} \\ \hline C_{4} \end{pmatrix} \begin{pmatrix} i_{H} & T_{I} \\ \hline p_{T} & -p_{T} & -p_{T}+1 \end{pmatrix}$$


Where:

.

 $\frac{C_3}{C_4} = 4.1534 \times 10^{-7}$   $i_H = 0.21 \text{ oz/yr}$   $T_K = 288^{\circ}K$   $P_T = (1 + 12/14.7) = 1.816 \text{ ATM}$   $L_S = \frac{(4.1534 \times 10^{-7}) \cdot (0.21) \cdot (288)}{(1.816^3 - 1.816^2 - 1.816 + 1)}$   $L_S = 1.34 \times 10^{-5} \text{ ATM} - \text{cc/sec.} \text{ (Standar)}$ 

.







.•

# 4.4 APPENDIX A - NORMAL AND ACCIDENT RADIOACTIVE MATERIAL LIMITS FOR THE 1-13C II CASK

Figures 4.2.1-1 and 4.3.2-1 define limits for activity of the medium within the containment vessel as defined in Paragraph 5.3.1 of ANSI Standard N14.5-1977. For the 1-13C II package, the medium is assumed to be a small quantity of residual water or steam, as appropriate.

Paragraph 4.4.1 outlines the method employed to establish these activity limit relations. Paragraph 4.4.2 outlines an example application of the method to a particular payload.

### 4.4.1 Derivation of Limit Relations

A leak test for design leaktightness verification (Re: P. 4-13) demonstrated that a new 1-13C II package did not leak at a rate of:



 $L_{1} = 1.34 \times 10^{-5} \text{ atm-cm}^{3}/\text{sec}$ 

This rate is at Standard Conditions per Paragraph 3.3 of ANSI N14.5 (. . . dry air at  $25^{\circ}$ C for a pressure differential at 1 atm. against a vacuum of  $10^{-2}$  atm. or less.)

Because of the procedure sensitivity requirements of ANSI N14.5, the cask may only be considered not to leak at a rate of two times the demonstrated leaktightness value or:

> $L = 2L_s = 2(1.34 \times 10^{-5}) \text{ atm-cm}^3/\text{sec}$ (Standard Conditions)



Revision D

Additional testing, for information only, showed that the package did not leak at a rate considerably less than  $1.34 \times 10^{-5}$  Atm-cm<sup>3</sup>/sec. For discussion purposes and because of the procedure sensitivity requirements of ANSI N14.5 it will be conservatively assumed that the leak rate after normal and accident conditions was: (See Discussion, Section 2.11, Paragraph 2.11.4.1, P. 2-233 showing that the pre-accident cask condition is actually representative of the post-accident cask condition).

 $L = 2.53 \times 10^{-5} \text{ atm-cm}^3/\text{sec}$  (Standard Conditions)

respectively. These rates are at standard conditions per Paragraph 3.3 of ANSI Standard N14.5 (. . , dry air at  $25^{\circ}C$  for a pressure differential at 1 atm against a vacuum of  $10^{-2}$  atm or less.)

For normal conditions, the permissible leak rate is given as:

 $L_{N} = \frac{R_{N}}{C_{N}} cm^{3}/sec$ Where:  $R_{N} = 2.73 \times 10^{-10} A_{2}/sec \text{ (Table 1, ANSI N14.5)}$   $C_{N} = \text{Permissible activity of medium, Ci/cm}^{3}$   $\overline{A}_{2} = \text{Maximum Activity for Type A Mixtures}$   $= \frac{\sum Ci}{\sum Ci/A_{2}}$   $A_{2} = \text{Maximum Activity for Type A (para. 3.2, ANSI N14.5)}$ 

or:

$$L_{N} = 2.78 \times 10^{-10} \qquad \frac{\Lambda_{2}}{C_{N}}$$
$$\frac{C_{N}}{\Lambda_{2}} = \frac{2.78 \times 10^{-10}}{L_{N}}$$

4-16

The quantity,  $L_{\chi}$ , represents the permissible leak rate of the medium (water) at temperatures and pressures associated with normal transport conditions. Correlation of this quantity with the demonstrated leak rate at standard conditions,  $L_{\chi}$ , is achieved via use of Equation (3.9) of ANSI N14.5:

	$\frac{2L_n n_n(P_u - P_d)N}{2L_n n_n(P_u - P_d)N}$	
	$n_{N}(P_{U}^{2}-P_{d}^{2})n$	
Yhere:	$L_n = 2.58 \times 10^{-5} \text{ atm-cm}^3/\text{sec}$	
	$n_{\rm H}$ = .0185cP, air viscosity	("Standard
	$P_{un} = 1 \text{ atm}$	<pre>Conditions",</pre>
	$P_{dn} = 1 \times 10^{-2} atm$	ANSI N14.5
	r <sub>χ</sub> = water viscosity, cP	(at "normal"
	P <sub>uN</sub> = vessel pressure, atm	) temperature
	= 1+ (psig/14.7)	and pressure
	P <sub>dN</sub> = 1 atm	(conditions.



The calculations for normal conditions are found in the table below:

Case No.	Decay Heat (Watts)	Cavity Temp. (°F)	Internal k Pressure Yis (psig)	later cosity (cP)	(C <sub>N</sub> /I <sub>2</sub> ) (cm <sup>3</sup> )
1	150	138.35	4.27 .	4759	4.592x10-4
3	300	142.70	4.72     .       5.68     .	4584 4290	4.002x10 <sup>-4</sup> 3.112x10 <sup>-4</sup>
4 5	400 500	159.47 167.51	6.75 . 7.96 .	401 <u>0</u> 3790	2.448x10 <sup>-4</sup> 1.962x10 <sup>-4</sup>
6 7	600 700	175.39 183.06	9.30 . 10.77 .	3590 3395	1.591x10 <sup>-4</sup> 1.299x10 <sup>-4</sup>
8	800	190.55	12.36	3239	1.080x10 <sup>-4</sup>

NOTE:

 Handbook of Chemistry & Physics, Chemical Rubber Company, New York Page 2181: Viscosity & Fluidity of Water, D<sup>0</sup>-100<sup>o</sup>C.



. •



For accident conditions, the permissible leak rate is given as:

$$L_{A} = \frac{R_{A}}{C_{A}} \text{ cm}^{3}/\text{sec}$$
Where:  $R_{A} = 1.65 \times 10^{-9} \text{ X}_{2}/\text{sec}$  (Table 1, ANS:  
N14.5)  
 $C_{A} = \text{Permissible Activity of Medium, Ci/cm}^{3}$   
 $X_{2} = \text{Maximum Activity for Type A Mixtures}$   
 $= \frac{\Gamma C1}{2(C1/A_{2})}$   
 $A_{2} = \text{Maximum Activity for Type A}$   
(Para. 3.2, ANSI N14.5)  
 $\text{Or:} \quad L_{A} = 1.65 \times 10^{-9} = \frac{X_{2}}{C_{A}}$   
 $\frac{C_{A}}{A_{2}} = \frac{1.65 \times 10^{-9}}{L_{A}}, 1/\text{cm}^{3}$ 

The quantity,  $L_A$ , represents the permissible leak rate of the medium (steam) at temperatures and pressures associated with accident conditions. Correlation of this quantity with the demonstrated leak rates at standard conditions,  $L_A$ , is achieved via use of Equation (B.5) of ANSI N14.5:

$$L_{A} = \frac{L_{a} n_{a} (P_{u}^{2} - P_{d}^{2})_{A}}{\eta_{A} (P_{u}^{2} - P_{d}^{2})_{a}}$$
  
Where:  $L_{a} = 2.68 \times 10^{-5} \text{ atm-cm}^{3}/\text{sec}$ 
  
 $n_{a} = .0185 \text{ cP}, \text{ air viscosity} \begin{cases} \text{"Standard"} \\ \text{Conditions}, \\ \text{P}_{da} = 1 \text{ atm} \\ \text{R}_{da} = 1 \times 10^{-2} \text{ atm} \\ \text{Steam viscosity}, \text{ cP} \\ \text{P}_{uA} = \text{vessel pressure, atm} \\ = 1 + (\text{psig}/14.7) \\ \text{P}_{dA} = 1 \text{ atm} \\ \end{cases}$ 







The calculations for accident conditions are found in the table below:

 Doolittle, Jesse S., "Thermodynamics for Engineers", International Textbook Company, 1964, Figure A-1: Viscosity of Gases, pp. 632.

At the temperatures and pressures associated with accident conditions for different payload decay heats the actual activity concentrations of the vapor inside the cask will vary. In order to determine the activity concentration of the vapor concentration (ml vater/cm<sup>3</sup> vapor) must first be determined as related to payload decay heats.





Yapor Concentration	رم را	1 Specific Vol.		x <u>(Density Wa</u> )		ier)	Ħ
-	Specific ft3	Yol. x x 2.832	<u>15-</u> fz.3 x 104=	- x <u>-1</u> x	<u>in</u> 3 16.39cm <sup>3</sup> <u>ft3</u>	×	

The results of the calculations are given in the following table and depicted graphically on Fig. 4.4.1-1.

150         320.42         90.25         4.886         3.280 x           200         324.67         95.92         4.614         3.473 x           300         332.98         107.52         4.146         3.865 x           400         341.06         119.77         3.737         4.286 x           500         348.93         132.85         3.390         4.727 x           602         256.92         147.37         3.070         5.220 x	ATTS	CAVITY TEMP. °F	PARTIAL YAPOR PRES. psia+	SPEC. YOL. OF YAPOR* Ft /1bm	YAP. CONC. ml WATER/ cm YAPOR
300         332.98         107.52         4.146         3.865           400         341.06         119.77         3.737         4.286           500         348.93         132.85         3.390         4.727           600         356.92         147.37         3.070         5.220	150	320.42	90.25	4.886	$3.280 \times 10^{-3}$
	200	324.67	95.92	4.614	$3.473 \times 10^{-3}$
600 356.92 147.37 3.070 5.220 x	300	332.98	107.52	4.146	$3.865 \times 10^{-3}$
	400	341.06	119.77	3.737	$4.288 \times 10^{-3}$
	500	348.93	132.85	3.390	$4.727 \times 10^{-3}$
700364.11161.402.8145.695 x8C0371.43176.572.5826.207 x	600	356.92	147.37	3.070	$5.220 \times 10^{-3}$
	700	364.11	161.40	2.814	$5.695 \times 10^{-3}$
	8C0	371.43	176.57	2.582	$6.207 \times 10^{-3}$



"Fundamentals of Classical Thermodynamics, 2nd Edition; Yan Wylen and Sonntag, 1973.







4-21







### 1.1.1 Example Use of Activity Limits

The activity limits derived in Section 4.4.1 are strictly applicable only to those payloads possessing some potentially leakable portion, such as residual water existing in dewatered resins. Clearly, solid-materials, such as neutron activated reactor components, possess no leakable fraction. As a result, these solid materials comply with containment requirements so long as there is "no ejection of contents."

To assess the containment requirements for materials such as devatered resins, the specific activities of isotopes within the leakable fraction results from residual fluid present in devatered resins.

As an example, consider a vessel containing approximately 8 cu. ft. of ion exchange resins where the only leakable portion of this payload is a small quantity of residual process fluid. There are no gaseous radioactive decay products present.

### Example: Normal Conditions

Assume the radionuclide distribution and maximum concentration for the residual water in the resin are as given in Table 4.4.2-1.







~ • • • •	•	2 1
******	· •• .	 

T				
	ACTIVITY		_	i
5	OF WATER	A2 VALUE	(Ci/m1)/Å2	
ISOTOPE	Ci/ml	ļ	<u> </u>	. <u></u>
Cestum 134	2.7702-005	10.00	2.7702-006	
Cesium 137	1.720E-004	20.00	8.6005-006	
Strontium 90	1.9005-005	0.40	4.7502-005	
Tritium	1.0005-006	1000.00	1.0005-009	•
Niobium 95	1.0002-009	20.00	5.000E-011	
Zirconium 95	1.1002-009	20.00	5.500E-011	
Ruthenium 106	7.3002-008	7.00	1.043E-008	
Antimony 125	1.5002-008	30.00	5.00CE-010	
lodine 129	5.7002-012	2.00	2.8502-012	
Tellurium 125M	1.2002-009	100.00	1.200E-011	
Tellurium 127M	2.2002-008	40.00	5.500E-010	
Tellurium 129H	4.40CE-011	30.00	1.467E-012	
Cobalt 60	1.4005-009	7.00	2.0002-010	
Manganese 54	1.2002-009	20.00	6.000E-011	
Cerium 144	1.0002-007	7.00	1.429E-008	
Uranium 234	1 5602-012	0.10	1.560E-011	
Uranium 235	1.6702-010	0.20	8.3502-010	
Uranium 236	4.6205-012	0.20	2.310E-011	,
Uranium 238	6.930E-009	unlimited		
Plutonium 239	1.290E-011	0.002	6.4002-009	
Plutonium 240	1.050E-012	0.002	5.2502-010	
Plutonium 241	1.540E-013	0.10	1.540E-012	
Americium 241	8.110E-014	0.01	1.014E-011	
		}		

I Ci/m1 = 2.199E-004  $I (Ci/m1)/A_2 = 5.890E-005$ Composite  $\overline{A_2} = \frac{I(Ci/m1)}{I(Ci/m1)/A_2} = \frac{2.199E-004}{5.890E-005} = 3.733 Ci$  $C_n = ICi/m1 = 2.199E-004$ 







Revision D

From Figure 4.2.1-1 the permissible activity of the water for A 320 watt payload decay heat is found to be:

 $\frac{C_N}{h_2} = 2.92 \times 10^{-4} \quad 1/cm^3$   $\frac{1}{2} = 3.733 \text{ Ci as determined previously}$ 

Therefore the permissible activity of the water is:

$$C_{x} = (2.92 \times 10^{-4}) \times 3.733 = 1.09 \times 10^{-3} \text{Ci/cm}^{-3}$$

The Containment Factor of Safety is:

F.S. =  $\frac{C_H}{C_D}$  = 1.09x10<sup>-3</sup>/2.199x10<sup>-4</sup> = a number larger than 1.0

If the factor of safety had been less than 1.0 the wattage was too high. A lower value would then need to be assumed and the calculations revised until a positive F.S. is obtained.

### Example: Accident Conditions

To determine the containment safety factor for accident conditions the vapor concentration is determined from Figure 4.4.1-1 given an actual payload decay heat. The actual activity concentration of the vapor (Ca) is then determined knowing the radionuclide distribution and residual fluid activity concentrations. The actual activity concentrations of the vapor are then compared to the permissible vapor activity (C<sub>A</sub>) found from Figure 4.3.2-1 for the given payload decay heat and the composite  $\lambda_2$  value determined previously.





Using the same radionuclide distribution and residual fluid activity concentrations as the example for normal conditions and an actual payload decay heat for accident conditions of 320 watts, the 'apor concentration is found to be  $0.00392 \text{ ml/cm}^3$ , from Figure 4.4.1-1.

The vapor activity is therefore determined from knowing the residual fluid activity (determined in the example for normal conditions) and the vapor concentration.

Ca = Yapor Activity =  $2.199 \times 10^{-4}$  Ci/ml x 0.00392 ml/cm<sup>3</sup> =  $8.62 \times 10^{-7}$  Ci/cm<sup>3</sup>

From Figure 4.3.2-1 the permissible activity of the vapor is computed for a 320 watt payload decay heat as follows:

$$\frac{C_A}{Z_2} = 6.25 \times 10^{-7} \ 1/cm^3$$

$$\lambda_2 = 3.733 \ Ci \ as \ determined \ previously.$$

Therefore the permissible vapor activity  $C_{A}$  is found to be:

$$C_{\pm} = (6.25 \times 10^{-7}) \times 3.733 = 2.333 \times 10^{-6} c i/cm^3$$

The containment factor of safety is:

F.S. = 
$$\frac{C_A}{C_B}$$
 =  $\frac{2.333 \times 10^{-6}}{8.62 \times 10^{-7}}$  = 2.71

If the factor of safety had been less than 1.0 the assumed maximum wattage was too high. A lower value would then need to be assumed and the calculations revised until a positive F.S. is obtained.

For radionuclides and actual residual fluid activity concentrations different from those assumed in this example the package operator may demonstrate compliance to the containment requirements by appropriate modifications of the methods outlined herein.





### .....3 Gasket Size and Bolt Torque Galculations:

Determination of Required Gasket thickness and Bolting Torque for the 1-13C II Cask. Gasket: Duro 50 .114" x 1.3", 32.90" I.D. Maximum internal pressure = 184 psi

Using the ASME Code & Section VIII, Division 1, Appendix 2, Part A-Flanges with ring type gaskets.

Two design conditions must be considered.

- Operation conditions determine bolt load resulting from the hydrostatic end force and the clamping force necessary to maintain a seal.
- 2) An initial seating load must be applied to the joint to assure a complete seating of the gasket.

1) From Paragraph 2-5(c) the operating bolt load is determined from:

 $W_{m1} = H + Hp = 0.785 G^2 p + (2b \pm 3.14 GmP)$ where

> H is the load due to hydrostatic pressure and Hp is the reaction' of the compressive load on the gasket required to maintain a tight joint.

Hp must be provided by the torquing of the bolts. From Formula (1) of the above paragraph:

Where

b = gasket seating width determined from Table 2-5.2

 $b = \sqrt{b_0}/2$ ,  $b_0 = N/2$ 





For the 1-13C II, X = 1.3"  $b = \sqrt{1.3/2}/2 = .403"$ G = gasket OD = 25 = 35.5 = 2(0.403) = 34.694 m = .5 for gasket with <75A Shore Durometer P = Internal pressure.

$$H_{P} = 2(.403) 3.14 (34.694)(.5) 184 = 8078 16.$$

2) Check also minimum initial seating load which is determined using Formula (2) of Paragraph 2-5(c)(2).

$$W_{m,2} = 3.14 bGy$$
 (2)

There

```
y = minimum design seating stress = 0
Table 2-5(1)
Y_{m2} = 0
```

Also from Formula (1) of the previous page:

 $H = 0.7253^2P = 0.725 (34.694)^2(124) = 173859$  lbs.

Total initial bolt load required:

 $K_{m1} = H + Hp = 173859 + 8141 = 182,000$  lbs. Load per bolt =  $F_b = 182,000/12 = 15167$  lbs.

Finally, the minimum bolt load for fully seating the lid standoff against the top of the cask body will be determined by following the method of Harris & Crede on page 35-15 of "Shock & Yibration Handbook". Eqn. 35.8 is:

$$E = \frac{F}{5} \left( \frac{h_1}{h - h_1} \right)$$



Revision O

Where.

E = elastic compression modulus  
(use 375 psi for Duro 50 rubber)  
Sesket area 
$$S = \frac{\pi}{4} (35.5^2 - 32.9^2) = 139.7 \text{ in}^2$$
  
F = compressive force in lbs.  
h = unstrained gasket thickness  
h\_1= standoff height (i.e. gasket thickness when fully compressed)  
For a 1/4<sup>2</sup> gasket of Duro 50 material  
E = 375 psi  
h\_1= .216  
h = .250  
F =  $\frac{E(h - h_1) S}{h_1}$   
=  $\frac{375 (.250 - .216) 139.7}{.216}$  = 8246 lbs.



The minimum load per bolt required to fully compress the gasket initially is therefore:

 $F_{b}' = 8246/12 = 687$  lbs/bolt

The bolt load required for maintaining the gasket sealing capabilities at the maximum expected pressures for accident conditions was shown to be:

 $F_{\rm b}$  = 15167 lbs.

Since  $F_b > F_b$ ' the lid will be fully seated against the cask body when torqued initially.

Torque required:

T ⊨ ★F<sub>b</sub>d

Mhere



k = .20 (dry)
= .15 (lubricated)

.

4-28

d = nominal diameter of bolt = 1.25"
T<sub>dry</sub> = .20 (15167) (1.25) = 3792 in-lbs.
= <u>216 ft-lbs.</u>
T<sub>lubr.</sub> = .15 (15167) (1.25) = 2844 in-lbs.
= <u>237 ft-lbs.</u>

For a 10% tolerance in a torque wrench the lid bolt torque requirements for the 1/4 gasket are as follows:

Lubricated:  $T_{L} = \frac{237}{0.9} = 263 \text{ ft-lbs.}$ Dry:  $T_{D} = \frac{316}{0.9} = 351$ Use: 270 ft-lbs  $\pm 10\%$  lubricated 360 ft-lbs.  $\pm 10\%$  dry.





Revision 0.

- - -



# 4.5 APPENDIN B

PARKER STAT-O-SEAL INFORMATION

--Applicable Catalogue Excerpts--



•



•

Stat-O-Seal'features

1

Revision 0

Parker STAT-O-SEALS are one-piece, molded-in-place seals with the nubber sealing element mechanically locked to the metal retainer, STAT-O-SEALS are designed to seal beneath the head of the fastener as shown for both INTERNAL and EXTERNAL pressure systems.



7\*\* ---inciple of controlled continetilized in the Stat-C-Seal, As be ....ener is tightened the rubber

seal is compressed, forcing the sealing surfaces securely around the fastener shank — but without squeezing the rubber beyond its elastic limit, or destroying its Inherent "memory." With the Stat-O-Seal there is full metal-to-metal contact of faying surfaces. The unique Stat-O-Seal design provides:

- . SET EDITERING OF SEAL
- AACUUM SEALING TO POSITIVE HIGH
   PRESSURE SEALING (\*SEE PACE 2.)
- RIGH RE-USEABILITY FACTOR
- + LONG RELABLE SOMOC
- + POOL PROOF INSTALLATION
- . MODERATE TOROUTHE
- HO RETORQUING
- DUICK YTSUAL INSPECTION

Stat-O-Seals are available for Immediate delivery in standard sizes for bolts and screws, from the number 6 size thru one inch, and will seal al-

is from --80° to 450°F. The archimed information on the following pages will assist you in ordering.





. . . .

4-31

....





BOLT TORQUE High pressure fluids to prevent stretching. acting under the head of a bolt tend CLEARANCE FOR EXTRA RUBBER to stretch it, litting the bolt head off Most Stat-O-Seals have some excess Its seat. When this happens, the space nubber, and, as noted on page 5, sposed under the bolt head can per- clearance must be provided to reit the Stat-O-Seal elastomer to ex- ceive h. In low pressure applications, trude and fail, it is, therefore, impor- this clearance may be either in the tant in high pressure applications to mounting surface or in a washer betorque the bolt well at assembly, pro- tween the bolt head and the Stat-Owiding enough prestress in the bolt. Seal, in high pressure applications,

4-32

however, the normal clearance must be on the high pressure side of the Stat-O-Seal, with little or no clear. ance on the low pressure side. Otherwise, the fluid pressure will extrude too much nubber into the clearance gap, resulting in a leak.



1

Sint-O-Seal 600 series





111081

<u>م</u> _	JITE DAS= NUMBER		1 D 016, 2	BUTIANA DIAMELEN E	9 D X -110	RETAINER THICKNELLS	THICKNESS	DIAN (TRAL ELLARANCE (RL/ J
	-6	.136	.130	275	385	بکھ = تەھ	دتن. = ددي	L MAL
	-8	.164	.156	.255	385	۸۵۵ = CAG	.050 = .003	X. WAX.
	-10	.150	.180	117	<i>ب</i> بر	200 = 2005	£72 ± £05	KNX
	•10 0.5.	.190	.186	ನಿಜ	<b>A68</b>	۱ .	4	X TO 1
	-4	.250	240	.:85	<b>.5</b> 05			X. MAX.
	-¥ 0.5.	250	.245	A22	531			• ۲. TO 1
2	-K.	12	.301	354	د0ع.			
` بن	*	375	264	<u>عىد</u>	.665		·····	
<i>,</i>	-K.	352	A27	.618	.760			<u> </u>
	*	<u>يحد</u>	A90	<b>.69</b> 6	C8&			
•	к.	.562	.517	.759	1.067	1	•	
•	-%	£25	.615	<b>"1</b> 18	1.193	200 ± 500	D72 = .005	
	*	.750	.740	.982	1.322	264 ± 2005	.096 ± .005	
•	*	\$75	364	1.105	1.510	_064 ± _005	_D96 ± _D05	1
	-1	1.000	839.	1.234	1.760	.054 ± .005	.096 ± .005	Y. TO Y
4		ففاكر والمرجوعة والمحاف						

#### NOTES:

- Color code provided only when specified by customer. A washer may be required under the boll head to distrib-[]]
- for que pressure (see page 4). fastener size dash number lo complete the call-out. Imple. 600-001-4. [4] For uses not listed here, see Parker Seal Gask-0-Seal Handbook, or contact your local Parker Seal representative.
- [5] Chrome Molybdenum retainers are heat treated to 125,-
  - DOG psi minimum tensile strangth.

- [6] Stainless steel retainers are annealed to half hard.
   [7] A slight parting line projection, similar to that found on O-Rings, will often be observable around the inside of the sealing element.
- [8] Trademarks: Yiton-duPont, Fluoret-Minnesota Mining [7] This material combination is not available in sizes -6 4
- -8. See 800 Series Lock-O-Seals,

[10] Other material combinations available on special order.



DOM: NO

Revision C

1	<b>``</b>					
JHER	GPTEIL KATION)	APLEIFICATION,		Sec. 2 (1)		
T DEL TIL						
600-0101	Stee. (As Specification)	Capmium Plate (No Specification)	Nitrile (BunsiN) (No Specification) N405-60	Industriat for weather and petroleum fluids — 65°F to — 225°F (—54°C to — 107°C). Water and many other fluids suitable with lower maximum temperature		
£XX-XXX ;	1072/1030 S.MI (CO S-693:	Cathorn Plate (CC P-16 Ci 2 Type I)	Nitrile (Euna N) (M'L-F-6855 Class 1 4 2 Graze 60)	Air, petroleum fluids, (fue's lois, reses, silicone lubricants and direster base lubricants —EE' to # 2251F		
600-015	707576 Aluminum (00-A-250/12)	Anodize (Mili-A-8525, Type II)	N403-60	(-54° ± + )07°0).		
	STANDARD PARTS (MAD)	TO ORDER [18]				
602042	4130 Chrome Maly Steet [5] (MIL-S-18729)	Cas Plate Dyes Black (00 P-416 Cl. 2, Type II)	Nitrile (Buna-N) (Mit-R-6255 Class 1 2 2, Grade 60) N406-60	Air petroleum fluids, (fue's oils, gases) silicore fubricar's and directer base fubricants $-65^{\circ}$ to $\pm 225^{\circ}F$ ( $-54^{\circ}$ to $\pm 107^{\circ}C$ )		
600-430	302/304 Staintess Size [6] (H5 5059)	Passivate (QC-P-35)	Nitrile (Buna N) (M.L.F.7362 Type II) 47-071	Mit-1-7808 Snythetic engine oil -65' to + 275'F (-54' to + 135'C).		
600-4:2	4:30 Chrome Moly Stret [5] (M5-18729)	Cat Plate Dret Biath 100 P-416 Cl. 2 Type III "				
10.701	1020/1030 Site! (00 5-695,	Calmium Plate (05 F-16 Cl. 2 Type I)	Nesprene (No Specification) CADS-70	Silicate Ester: - 65' to - 275'F (-54' to + 135'C).		
600-742	4120 Chrome Moly Stee' [5] (M. 5 18729.	Cas Plate Dyes Blast 103-P-416 Ci. 2 Type II)				
600-3130	302/304 Stainless Steet (6), 19) (Mil-S-5059)	Passivate (CC-P-35)	Fivorosaroo- 18: 19; (MIL-R-83248, Type ( Class 1)	Air, Petroleum fluids, silicone fluids, many acids and phosomate esters -20° to +400°F		
600 3142	4130 Chrome Moly Steel (5) (MIL-5-18729)	Cadmium Plate Dyed Black (DQ-P-416 Cl. 2 Type II)	¥185475	(—29° ю +204°С).		
800-6015	7075-T6 Aluminum (QC-A-250/12)	Anod ze (MIL-A-8625 Type II)	Fluorosilicone (MIL-R-25988)	Petroleum fluids, silicone fluids, silicati esters -65° to +350°F		
<b>6</b> 00-6030	302/304 Stainless Ster' 16; (MIL-S-5059)	Passivate (DGA-35)	57 60 67 60 11830-65			
600-6042	4130 Chrome Moly Steel [5] (MIL-S-18729)	Cas Plate Dyes Bisch (Q3-P-416 Cl. 2 Type II)		_		
600-6230	302/304 Staintess Stee (C) (MIL-5 5059)	Passhrate (CS-A 35)	Ellicone (AMS 3304) \$604-70	Air and gases -83' to +453'F (-62' to +232'C)		
600-6242	4130 Chrome Holy Stret[5] (MIL-S-18729)	Cadmium Plate Dyed Black (QQ P-416 Cl. 2 Type II)				
-2230	302/304 Stainkss Steel 16) (MIL-S-5059)	Passivate (OC-A-35)	Butyl (AMS 3238) 8318-70	Stydrol, phosphate esters, water, ateam and air -65° to +225°F		





.

.

5

# recommended torque

### RECOMMENDED BOLT TORQUE VALUES FOR STAT-O-SEALS-POUND INCHES

	Betainer Motal			منط	~~~~	jina mia	na 31.0+1	C	
	Part Jaman	805-0101 64.2 X ( 64.7 X ( 60.7 X ( 80.7 X ( 80.		600-015 600-4015		407-430 40-3535 40-4235 407-4235		800-047 6/2-447 6/2-47 8/2-747 802-6247	
	\$17E	Min.	Maz.	24 m	<b>M</b> e 1.	M m	Mas.	Hn,	àd a d.
-	6	5	33	5	25	-		5	45
	2	10	ట	10	50	-		10	75
	10	13	60	13	70	13	60	13	90
	10 05.	33	ట	13	<b>8</b> 0 .	13	60	13	95
	<u> </u>	40	100	40	110	40	100	40	145
	¥ CS.	40	100	45	110	40	100	75	145
<b>*</b>	K.	හ	180	70	₿5	60	147	75	180
- )	*	C4	220	83	160	(AU)	2225	80	230
•	χ.	110	282	110	260	110	300	110	400
	<del>14</del>	130	540	130	<u></u> 3හ	130	420	130	600
	X.	450	800	400	500	250	1000	400	1000
	*	650	1100	650	1040	350	1100	650	1700 .
	*	550	2920 .	550	1550	500	29900	650	3820
	K	650	3920	650	1900 ·	570	3900	650	5400
	1	720	5900 <sup>'</sup>	ತ್ರಾ	2840	700	59-00	720	3000

The values in this table exceed permissible torques for many bolt materials.

Data is for fine thread, lubricated, SAE Grade, B Bolts.

Torque values may be increased 20% if the bearing surface under the head of the bolt completely covers the Stat-O-Seat

# fastening torques

Whenever fastening torques are discussed by engineers, there is always controversy. There are many variables such as wrenching methods and thread friction influenced by lubrication, plating, surface finishes, length of grip, class of thread, etc. Often torque wrenches are not even used. Machines do a fairly consistent Job, but if the seating surface is not flat, using pure torque values can cause difficulty. Parker STAT-O-SEALS provide a certain amount of latitude in torque requirements. Actually, STAT-O-SEALS are often capable of sealing when only "finger tight" so that extra high fastening torques may not be necessary. If extra firm seating is required, STAT-O-SEALS should be torqued within the limits shown to avoid crushing the metal retainer itself.



4-35



Revision O

•

	3011 J.1	LCM CLARDH STLL	វាមា ភា. ភា.	<b>BUTZ</b>	SIL ICON BRONLL	لال 14 تاريخ <u>الله</u> 14 تاريخ	ារ ភា. ភ.	100 CL
Ì		* h	An	<b>s</b> .h.	<b>R</b> h	<b>h</b> h	<b>h</b> n	Ba pr.
	6-40	87	9.6	7.9	8.9	5.3	101	9 5
	6-40	10.9	12.1	9.9	11.2	6.6	12.7	12.3
	3 32	17.8	15 8	16.2	18 4	10.8	20 7	20 2
	3 36	19.8	22.0	18 0	20.4	12.0	23.0	22 4
	10-24	20 B	22 B	1E 6	21.2	13 8	23 E	25 <del>3</del>
	10-32	29 7	31.7	25 9	29.3	19-2	33.1	34 9
	420	65.0	75.2	61.5	68 B	45 6	78 8	85 3
	428	90.0	94.0	77.0	87,0	57.0	99 0	106 C
	K.*-18	129	132	107	123	30	138	149
	K.*-24	139	142	116	131	86	347	165
Ľ	***•16 ***•24	212 232	235 259	192 212	219 240	143 157	217	264. 254
	X*-14	338	376	317	349	228	393	427
	X*-20	361	400	, 327	371	242	418	451
	H*·13	485	517	422	480	313	542	584
	H*·20	487	641	443	502	328	565	613
	Х.°-12 Х.°-18	613 668	682 752	558 615	632 697	413 455	713 787	774 855
	%-11 %-18	1000 1140	1110 1244	907 1016	1030 1154	715 798	1160 1301	1330 1462
	44*•10 ₩*•16	1259 1230	1530 1490	1249 1220	1416 1382	980 958	1582 1558	1832 1790 ·
	71*-9	1919	2328	1905	2140	1495	2430	2775
	71*-14	1911	2313	1895	2130	1490	2420	2755
	2*- <b>8</b>	2132	3440	2815	3185	2205	· 3595	4130
	1*-14	2562	3110	2545	2885	1995	3250	3730

SUGGESTED MAXIMUM TORQUE VALUES FOR FASTENERS OF DIFFERENT MATERIALS

NOTE: Table & Intended as a guide, obtained from Machine Design's Reference Issue, "Fastening and Joining," June 15, 1967



,

• .

### 1.1.3 Combustible Gas Generation Safety Assurance

Assurance of safe shipment of vessels which may generate combustible gas is based on meeting the following criteria over the shipment period.

- (i) The quantity of hydrogen generated must be limited to a molar quantity that would be no more than 5% by volume at STP (or equivalent limits for other inflammable gases) of the secondary container gas void (i.e., no more than 0.063 g-moles/ft<sup>3</sup>); or
- (ii) The secondary container and the cask cavity (if required) must be inerted with a diluent to assure the oxygen, including that radiolytically generated, shall be limited to 5% by volume in those portions of the package which could have hydrogen greater than 5%.

The following discussion establishes the safety assurance of the above criteria and describes how vessels will be prepared such that the above criteria will be met while the vessels are in shipment.

Criterion (i) essentially stipulates that the quantity of hydrogen shall be limited to 5% of the secondary container gas void at STP. This 5% hydrogen gas volume at standard conditions is equivalent to a, hydrogen partial pressure of 0.735 psi or 0.063 gram moles/cubic foot. By actual experiment\*, the ignition of a 5% (H<sub>2</sub>) gas volume has been demonstrated to produce an approximate 2.3 psi incremental pressure increase above an initial pressure of approximately atmospheric. The reason this is so is that the 0.063 gram moles of hydrogen per cubic foot provides <u>such</u> a small source that the peak pressure rise resulting from ignition of this source is slight. It is felt that 1-13C II is able to sustain an incremental 3 psi internal pressure rise from atmospheric pressure without failure. This pressure increase is not considered in Section 2.6 and Chapter 3 and 4 calculations as it is considered to be insignificant.









Criteria (ii) is invoked to ensure that when a secondary container's hydrogen concentration potentially exceeds 5% volume, release of that hydrogen to the then existing total volume (secondary container void plus cask void) will not result in a total mixture of greater than 5% volume hydrogen in a greater than 5% oxygen atmosphere. Maintaining the oxygen lower than five (5) volume % assures a non-flammable mixture.\*\*

- \* Carlson, L.W., et al., "Flame and Detonation Initiation area Propagation in Various Hydrogen - Air Mixtures With and Without Water Spray", Atomic International Division of Rockwell International, Canoga Park, California, May 11, 1973. (The incremental pressure rise is basically independent of the total volume under test--i.e.,--that the 0.063 gram moles per cubic foot relationship to 2.3 per rise is valid for one or many cubic foot of specimen volume.)
- \*\* Lewis, B. and von Elbe, G., "Combustion, Flames and Explosions of Gases", Academic Press, New York, 1961, Second Edition, Appendix B.





# 5.0 <u>SHIELDING</u>

### 5.1 Discussion and Results

This cask will be operated such that the radioactive inventory within the cask will not result in dose rates exceeding 200 mRem/hr on the cask surface, or 10 mRem/hr at six feet from the surface of the cask.

The package shielding must be sufficient to satisfy the conditions on 10 CRF 71, Paragraph 71.73 for the hypothetical accident conditions. It is shown that shielding loss resulting from either the 30-foot drop or the fire transient will not increase the external radiation dose rate to more than 1.000 mRems/hr at 3 feet from the external surface of the cask.

The fissile material to be loaded into the debris containers has experienced small burnup levels of only 3165 MWD/MTU. This results in a very small neutron dose rate relative to the fuel gamma levels. Hence, the effect on shielding is insignificant.

### 5.2 Source Specification

The equivalent point source, assuming  $Co^{60}$  energy, is determined for the normal geometry. This equivalent source is then used to evaluate the effects of the hypothetical accidents.

)

$$\phi_{\gamma} = \frac{BS_{0}e^{-b}1}{4\pi r^{2}} \qquad (1)$$

$$\phi_{\gamma} = Photon \ Flux, \ \frac{\gamma}{cm^{2}-s}$$

$$S_{0} = Equivalent \ Source = \frac{\gamma}{s}$$

$$b_{1} = \frac{Ev_{1}t_{1}}{i} \ for \ shielding$$









3 = Buildup Factor

r = Distance from source to dose point

$$D = x_{\gamma}$$
  
 $X = 2.3 \times 10^{-6} \frac{R/hr}{r_{\gamma}} \text{ for } Co^{60}$  (2)

Through the side of the cask, the following values are used: (3)

Lead:  $t = 5^{n} = 12.7 \text{ cm}, \nu/\rho = .0600 \frac{\text{cm}^{2}}{\text{gm}}$ Steel:  $t = 1\frac{1}{3} = 3.175 \text{ cm}, \nu/\rho = .0515 \frac{\text{cm}^{2}}{\text{gm}}$ 



For these values:

 $b_1 = 10.0$ B = 4.0 (4)

For the two dose conditions:

 $D_1 = 10$  mRem/hr. r = 6.31 + 36 = 42.31 inch  $D_2 = 200$  mRem/hr. r = 6.31 inches - Distance to surface of cavity.

$$S_{o} = \frac{4\pi r^{2} D}{BKe^{-b} l}$$

Substituting into the above expression:

(1)  $D = 10 \text{ mRem/hr} \in 3 \text{ feet}$   $S_0 = 3.475 \times 10^{12} \text{ y/s}$ (2) D = 200 mRem/hr on Surface $S_c = 1.546 \times 10^{12} \text{ y/s}$ 







The doserate on the surface governs and will be used in the accident analysis.

5.3 Model Specification

Hypothetical accident drop tests described in Section 2.11 demonstrate conclusively that the cask and its shield are not altered in any fashion by the hypothetical accident conditions. Nonetheless, shielding analyses for accident conditions, are carried out assuming a damaged geometry presuming no overpack exists. These very conservative analyses are taken directly from the Safety Analysis Report supporting the 1-13C II Packaging, USA/9081/B(). In all instances, both impact and thermal damage assumptions due to hypothetical accident events are more severe than found by test, for drop events, (see Section 2.11) and analyses, for thermal events (see Section 3.0).

Shield displacements are assumed to result from either the 30 ft drop or from the lead melt in the fire transient. The maximum displacement for the drop will occur using the minimum dynamic flow pressure for lead given. This value is:

k = 5000 psi

The shielding deformation resulting from the 30 ft drop will be developed for each impact mode using this value.

The lead volume displacement for the fire transient is assumed as:

 $V_{\rm F} = 1305 \ {\rm in}^3$ 

### 5.3.1 Damage Predictions - Corner

The shielding displacement for the 30 ft corner drop will be determined for the combined displacement of the lead shield and the steel shell and fire shield.





The volume of the ungula of a cylinder is:

 $V = R^3 tana \{f(\theta)\}$ 





 $tana{f(E)}$ 



The total displaced volume from kinetic energy considerations is:

 $V_T = \frac{hW}{k}$  Where:  $\overline{k}$  is the effective dynamic flow stress of the combined lead and steel.



5-4



Assume: 
$$\tilde{k} = \frac{k_s V_s + k_L V_L}{V_s + V_L}$$

Then: 
$$V_{T} = \frac{hW(V_{S} + V_{L})}{K_{S}V_{S} + K_{L}V_{L}} = \frac{kWV_{T}}{K_{S}V_{S} + K_{L}V_{L}}$$
  
 $hW = k_{S}V_{S} + K_{L}V_{L}$   
 $= k_{S}(R_{O}^{3} - R_{L}^{3}) \tan \left\{f(\hat{\tau})\right\} + K_{L}R_{L}^{3} \tan \left\{f(\hat{\tau})\right\}$   
 $= \left\{k_{S}R_{O}^{3} + R_{L}^{3}(k_{L} - k_{S})\right\} \tan \left\{f(\hat{\tau})\right\}$ 

$$f(3) = \frac{hW}{(k_s R_0^3 - R_L^3 (k_s - k_L)) \tan \alpha}$$
Let:  $R_L$  = Lead Outer Radius = 18.75 inch  
 $R_0$  = Cask Outer Radius = 19.50 inch (with  
fire shield)  
 $k_s$  = Steel Dynamic Flow Stress = 45,000 psi  
 $k_L$  = Lead Dynamic Flow Stress = 5,000 psi  
 $\alpha$  = 23.3°  
 $h$  = 360 inch  
 $W$  = 25,950 lbs.

 $F(2) = \frac{(360) \text{ in } (25950) \text{ 1bs-in}^2}{(.533) ((45000) (19.50)^2 - (40000) (1875)^3) \text{ 1bs-in}^3}$ =  $\frac{1.73 \times 10^7}{(3.337 - 2.637) 10^8}$ = .248 From Figure 5.3.1-1:  $\theta = 1.225$  Radians = 70.2°









The shielding geometry for lead displacement in the corner for the two cases is as shown:



R

5"

1 3/4"

1/2"

13.75

19.50

CASE 2

6.75

1

5.75"

- $\theta = 70.2^{\circ}$  (Figure 5.3.1-1) a = 28.3<sup>°</sup> R = 19.50"
- $c = (1 \cos 6)R = 12.29$  in.  $c = s \tan \alpha = 6.94$  in.  $t_s = 1.0 + .50 = 1.57$  in.  $cos \alpha$

R = 18.75" É = 3.45 in.







# 5.3.2 Damage Prediction - Side

The lead displacement for the geometries along the side of the cask are shown below. The displacement for the fire analysis assumes all the lead melt opposite the reinforcing ring.



$$9 = 59.9^{\circ} \text{ (Figure 5.3.1-1)}$$

$$R = 18.75^{\circ}$$

$$\delta = (1 - \cos \frac{\theta}{2}) R$$

$$\delta = 2.50 \text{ inch}$$

$$t = 5.00 \text{ inch}$$

$$X_{T} = t - \delta = 2.50 \text{ inch}$$



(b) Fire

v<sub>y</sub> = 1303 in<sup>3</sup>

R<sub>F</sub> = 17.75" - 2

1



 $\delta^2 - 17.755 + \frac{V_M}{2\pi H_R} = 0$   $\delta = 1.61$  inch  $X_F = 4.00 - 1.61$  $X_F = 2.39$  inch





## 5.3.3 Damage Prediction - Ends

Consider the amount of lead displacement for the 30 foot free drop on both the top and the bottom of the cask. The displacement will be determined as follows.

 $\Delta H = \frac{R W H}{\tau (R^2 - r^2) (t_s c_s + R c_{Pb})}$ 

Where: R - Outer Radius of Lead = 18.75 in. r - Inner radius of lead = 13.75 in. W - Cask Weight = 25,950 lbs. H - Drop Height = 360 in.  $c_s$  - Steel Dynamic Flow Stress = 45,000 psi  $c_{pb}$  - Lead Dynamic Flow Stress = 5,000 psi  $c_s$  - Steel Shell Thickness= 0.50 in.

The thickness of the steel shell has been taken as the thickness of the cask outer shell. The displacement of the 0.25 inch thick fire shield has been neglected for conservatism.







# (a) Bottom Displacement





r = 13.75 in. R = 18.75 in.

Assume only the lid does not deform and all deformation occurs in the cask body.

5-11


$$2H = \frac{(13.75) \text{ in } (25950) \text{ lbs } (360) \text{ in-in}^2}{-((13.75)^2 - (13.75)^2) \text{ in}^2 ((1.0) (45000) - (5.00) (5000) \text{ lbs-in}}$$
$$= \frac{(1.75 \times 10^8) \text{ in}}{\tau (162.5) (7.0 \times 10^7)}$$
$$= 4.90 \text{ in.}$$
$$X_{IT} = t_{Pb} - 2H$$
$$= 0.85 \text{ inch}$$

The deformation stops short of the cask cavity lid for this case. If the lid also deforms, the deformation model will be the same as for the bottom end drop.

:





△H = 2.96 in. X<sub>IT</sub> = t<sub>Pb</sub> - 2H = 2.79 inch





# 5.4 Snielcing Evaluation

To determine the lead displacement having the greatest effect on dose rate, compare the attenuation for each case as follows:

$$A = \frac{Be^{-b_1}}{4\pi^2} \qquad \text{Where: } b_1 = \frac{2}{2}u_1 u_1$$

r - Distance from inside surface of cask cavity
 to a dose point 3 feet from the outside surface
 of the cask.

A summary of the pertinent data is presented in the following table.





TABLE 5.4.1

•

i.

CY2E		t-steel (cm)	t-lead (cm)	51	e <sup>-b</sup> .	3 <sup>(4,5)</sup>	à
CORNER-IMPACT	199.17	3.93	4.75	4.39	7.531-3	2.5*	1.493x10 <sup>-'</sup>
CORNER-FIRE	112.73	3.13	10.54	3.52	1.995-4	3.6*	4.497x10 <sup>-9</sup>
SIDE-IMPACT	100.97	3.18	6.35	5.65	3.505-3	2.8*	7.660x10 <sup>-8</sup>
SIDE-FIRE	107.47	5.72	6.07	6.509	1.491-3	9.0**	9.246x10 <sup>-3</sup>
TOP-IMPACT	102.34	3.81	7.09	6.419	1.630-3	3.0*	3.715x10 <sup>-3</sup>
- SOTTOM- IMPACT	102.34	3.175	7.72	6.59	1.376-3	3.0•	3.136×10 <sup>-3</sup>

\*Use Buildup for Lead, Figure 5.4-1 \*\*Use Buildup for Iron, Figure 5.4-2



3ASIC DATA









.

DALIC DATA



:





The case of 30 foot impact on the corner will be limiting:

D = KAS  $K = 2.3 \times 10^{-6}$  <u>R/hr</u>  $\phi$ S<sub>0</sub> = 1.546 x 10<sup>12</sup> y/S D = 0.530 R/hr < 1.0 R/hr

Therefore, the dose rate following hypothetical accident events, assuming very conservative damage estimates, is less than the maximum permissible rates defined in 10 CFR 71. The Model 1-13C II package fully complies with all shielding requirements of 10 CFR 71.





.

ł

#### 5.5 Reierences

- <u>Reactor Shielding Design Manual</u>, Theodore Rockwell III, D. Van Nostrand Co., Inc., Princeton, N. J. 1956 - page 347
- 2. 131D, page 19
- 3. IBID, page 447
- 4. IBID, page 430
- 5. IBID, page 432





# 6.0 CRITICALITY EVALUATION

# 6.1 Discussion and Results

The general debris container configuration can be shown to be safe since the maximum quantity of material shipped is significantly less than the critical mass limited to (15.4 kg -  $UO_2/400$  gm &-235).

# 6.2 Package Fuel Loading

Fissile material is shipped in the general debris containers with a limit of 400 grams as U-235 in a solid form. the shipment meets the requirements of 10 CFT 71.22.

# 6.3 Model Specification

Not applicable

# 6.4 Criticality Calculation

A criticality evaluation on a shipment with the general debris containers considers the effect of the shipment of 400 grams of U-235 with the fuel debris. It is conservatively assumed that the maximum fuel enrichment in the TMi-2 core (prior to burnup) of 3% comprises the sample. <u>The configuration is shown to be safe since the maximum</u> <u>quantity of material shipped is significantly less than the critical</u> <u>mass</u>.

The original TMI-2  $UO_2$  fuel pellets has a maximum of 3% U-235 (uranium dioxide with the uranium content enriched to 3% U-235 by weight). It is impossible to achieve a critical configuration of pure  $UO_2$  enriched to 3% with a mass of less than 75 kg assuming optimal moderation and full neutron reflection. Hence no arrangement of the TMI-2 core debris samples contained in the shielded debris canisters can result in a critical configuration under any normal or accident







conditions with the total mass limited by 29 kg by the volume restriction. The actual mass of approximately 15 kg will provide a safety factor of five.

Isotopes of plutonium generated by neutron activiation can be ignored in the critically evaluation since the calculated total core inventory of 161 kg of Pu isotopes in the total core mass of approximately 125,000 kg represents a negligible percent compared to the U-235 content.

6.5 Critical Benchmark Experiments

Not applicable.



Revision U











SUBCRITICAL MASS LIMITS FOR INDIVIDUAL SPHERES OF WATER REFLECTED AND MODERATED U ( $\leq$  5)

Acrision 0





SANL-SHG -6-+2242



SUBCRITICAL VOLUME LIMITS FOR INDIVIDUAL SPHERES OF WATER REFLECTED AND MODERATED U ( $\leq$  5)



ACT 12 (20) 0







SUBCRITICAL DIAMETER LIMITS FOR INDIVIDUAL CYLINDERS OF WATER REFLECTED AND MODERATED U ( $\leq$  5)

REVISION U

# 5.5 <u>Appendix - References</u>



1. J. T. Thomas, Ed. <u>Nuclear Safety Guide</u>, Tid - 7015, Rev. 2, June 1978.

 L. "K. Poppe's letter Charles E. MacDonald, "Amendment to Certificate of Compliance USA/9152/8 to Handle Fissile Material", July 15, 1985.





### 7.0 OPEPATING PROCEEURES

This chaoter describes the general procedures for loading and unloading the Model 1-13C II Shipping Package.

7.1 Procedures for Loading Package

- 7.1.1 Loosen and release ratchet binders securing upper overpack to lower overpack.
- 7.1.2 Remove upper overpack by attaching suitable hocks to lifting lugs. Care should be taken during this operation so that the overpack is not damaged while setting it down.
- 7.1.3 Determine if cask must be removed from lower overpack in order to load. If so, attach two lifting ears to cask walls and torque lifting ear bolts to 200 ft.-lbs. ( $\pm$  10% lubricated or dry). Cask may then be lifted vertically from lower overpack and carefully placed in position for loading.
- 7.1.4 Using proper radiological precautions, loosen vent plugs to relieve potential internal cask pressure prior to loosening and removal of the 12 bolts which secure the cask lid to the cask body.
- 7.1.5 Remove the lid by attaching a suitable hook to the lid lifting lug. Care should be taken during this operation so that the lid gasket and O-ring are not damaged.
- 7.1.6 Inspect the cask interior to ensure that there are no loose articles.
- 7.1.7 Inspect and clean lid gasket, O-ring, drain seal, and vent seal and gasketed surfaces. Replace any gasket if it shows sign of wear or deterioration.



#### 7.1.3 Load cask

7.1.3.1 Payload weight shall not exceed 3.000 pounds. When loaded in a submerged fashion, both vent and grain plugs should be removed. Plugs should be reinstalled after cask cavity is drained. When positioning the lid onto the cask while loading in a submerged fashion, the lid shall be shimmed open to allow venting and drainage from the area between the O-ring and flat lid gasket.

#### 7.1.8.2 Special Options for Loading Fissile Material

For guidelines formulated for the detailed procedure for collecting nuclear core assembly debris from the reactor vessel in the shielded container of 2R Vessel see section 7.1.8.2.1. For guidelines formulated for a detailed procedure of collecting fissile material or surface contaminated material from the reactor vessel by special buckets see section 7.1.8.2.2.

#### 7.1.8.2.1 Shielded Container with 2R Vessel Option

#### 7.1.8.2.1.1 <u>Sample Collection</u>

- a. Weigh the debris shielded container with debris bucket or fuel pin sample cell prior to loading with core assembly debris. (See Section 1.3.1.2).
- b. Place an open debris shielded container on a platform with either the debris bucket or the fuel pin sample cell in the bottom of the debris shielded container. Lower the container into the reactor vessel.



Revision D

7.1.3.2.1.2	Samole	<u>Collection</u>	from	the	<u> </u>	
	<u>Vassal</u>					

- a. Samples will be collected from the reactor vessel.
- b. Samples will be individual chunks of debris and/or small gravel/fines whose volume will not exceed the volume of the cavity of the debris bucket in the debris shielded container. (See Section 1.3.1.2)
- c. Other samples will be short sections of fuel rod pins whose volume will not exceed the volume of the fuel pin sample cells. (See Section 1.3.1.2.)
- 7.1.8.2.1.3 <u>Placement of Samples in Debris Shielded</u> <u>Container</u>

Place the sample in the appropriate debris bucket or fuel pin sample cell within the shielded container.

7.1.8.2.1.4 <u>Closure of Container</u>

Close the lid on the debris shielded container.

- 7.1.8.2.1.5 Drainer Container and Sample
  - a. Raise the platform with the debris shielded container(s) above the surface of the water.



### Revision O

- b. Allow the debris shielded container(s) | to orain and prip-dry until no water is pischarged from the container(s).
- c. Tighten the lid locking bolt and jam nut on the container(s).

7.1.8.2.1.5 <u>Placement of Loaded Debris Shielded</u> <u>Containers in 2R Vessels</u>

- a. Weigh the loaded shielded debris container(s) after draining.
- b. After a debris shielded container is thoroughly drained and weighed, transfer the container to a 2R vessel.
  Shore up the debris container in the 2R vessel as shown. (See Section 1.3.1.3.)
- 7.1.3.2.1.7 <u>Close 2R vessel per prescribed</u> procedures

Loading Cask with 2R Vessels

Verify that the total net weight of the samples to be placed in the shipping cask is less than 15.4 kg.

- a. Place shoring for the first 2R
   vessel in the 1-13C II cask. (See
   Section 1.3.1.3.)
- b. Transfer the first 2R vessel to the 1-13C II cask, being sure that vessel is aligned per shoring drawing.



- c. Place additional shoring over the first vessel and repeat the transfer and shoring procedure for the second and third vessels.
- 7.1.9 Remove the lid shims and secure the lid to the cask body by torquing the 12 lid bolts to 270 ft-lbs  $\pm$  10% lubricated (360 ft-lbs  $\pm$  10% dry).
- 7.1.10 Verify the leaktightness of the assembled package prior to each shipment according to Section 8.1.3.
- 7.1.11 If cask has been removed from lower overpack for loading, use lifting ears to place it back in lower overpack on transport vehicle. Remove lifting ears.

NOTE: UPPER OVERPACK WILL NOT FIT ON CASK TOP AND RATCHET BINDERS CANNOT BE TIGHTENED IF LIFTING EARS ARE LEFT ON CASK.

- 7.1.12 Verify that package is oriented on transport vehicle per Figure 7.1.1-1. Verify that distance from cask centerline to transport vehicle tiedown lug is 76 inches.
- 7.1.13 Install the overpack, verifying that tiedown lugs are oriented to trailer tiedown assemblies and that ratchet binder lugs are aligned vertically.
- 7.1.14 Attach and tighten ratchet binders attaching upper overpack to lower overpack.
- 7.1.15 Inspect the package for proper labeling required to meet applicable regulations.



7.1.16 Secure package to transport vehicle using appropriate tiedowns.

## 7.2 Procedures for Unloading the Package

- 7.2.1 Move transport vehicle to unloading site.
- 7.2.2 Perform a visual inspection of unopened package exterior. Record any significant observations.
- 7.2.3 Repeat steps 7.1.1 to 7.1.3 above.
- 7.2.4 Loosen drain and vent plugs six (6) turns to vent pressure generated by decay heat.
- 7.2.5 Repeat steps 7.1.4 to 7.1.5.
- 7.2.6 Unload cast.

# 7.3 Preparation of Empty Packages for Transport

The 1-13C II shipping package requires no special transport preparation when empty. Standard loading and unloading procedures outlined above shall be followed.

# 7.4. <u>Procedures for Shipment of Packages Which</u> <u>Generate Combustible Gases</u>

Procedures for preparing packages for shipment which radiolytically generate combustible gases are outlined below. These procedures are divided into two categories:

- a. Combustible gas control by inerting, and
- b. Combustible gas suppression.









# 7.4.1 Combustible gas control by inerting

- a. Dewater the secondary container. The bulk of the free water is removed from the secondary container by displacing the water with nitrogen gas.
- b. Inert the secondary container (and, if necessary, the cask). The inerting operation is done at the dewatering station just before the cask is loaded. Inerting is performed if the hydrogen generated will be greater than 5% in any portion of the package. Inerting is intended to limit the oxygen that is radiolytically generated. If a leak path can develop between the secondary container and the cask, the cask will also be inerted. (The inerting of the cask shall be performed according to a special procedure.)
- c. Sample the gas in the package (and cask, if inerted).
- d. Load the secondary container.
- 7.4.2 Combustible gas suppression
- a. Dewater the secondary container.
- b. Install the combustible gas suppression system (e.g., a vapor pressure catalytic recombiner).
- c. Sample the gas in the secondary container and measure static pressure. This will assure that the combustible gas control method is working properly.
- d. Load the secondary container.





#### 3. ACCEPTANCE TESTS AND MAINTENANCE PROGRAM

#### 3.1 Acceptance Tests

The following discussion includes details of those tests which must be performed on the 1-13C II package.

#### 3.1.1 <u>Visual Inspection</u>

The entire package, both inside and out, shall be visually inspected prior to loading, noting any significant damage (cracks, punctures, broken welds, etc.). Exterior stencils and nameplates must be in place and legible. Seals and bolts must be in place and in good condition.

#### 8.1.2 <u>Structural and Pressure Tests</u>

No structural or pressure testing is required.

#### 8.1.3 Leak Tests

The package shall be leak tested before first use in accordance with the requirements of ANSI N 14.5, Paragraph 6.3.1 for Containment System Fabrication Verification. The package will be leak tested every twelve months thereafter in accordance with the requirements of ANSI 14.5, Paragraph 6.3.1 for periodic testing. After the third use of the package and whenever any gasket or seal is replaced, it will again be tested to the same levels as the fabrication verification test. All the above tests will be per the leak test procedure of Appendix 8.3.

The flat lid seal will be leak tested after each loading with radioactive material prior to shipment in accordance with the assembly verification test procedure (Appendix 8.3).







#### 3.1.4 <u>Component Tests</u>

There are no values or other penetrations into the containment boundary, excepting the vent/test ports noted in Section 3.1.2, and, therefore, no additional component testing will be performed. The various foam properties of the impact limiters will be checked and verified during fabrication.

#### 8.1.5 Test for Shield Integrity

A gamma scan or equivalent test shall be performed on the 1-13C II package prior to initial use to detect any shielding deficiencies from lead voids equal to or greater than 10% of shielding thickness.

#### 8.1.6 <u>Thermal Acceptance Testa</u>

No thermal acceptance test will be performed on the 1-13C II package. Please refer to Section 3.0 for thermal evaluation.

#### 8.2 <u>Maintenance Program</u>

The owner (US DOE) is committed to an ongoing preventative maintenance program for all shipping packages. The 1-13C II package will be subjected to routine and periodic inspections and tests as outlined in this section and DOE approved procedures.

#### 3.2.1 <u>Structural and Pressure Tests</u>

Routine visual examinations will be performed to detect damage or defects significant to package condition. Exterior stencils, nameplates, seals and bolts will be verified in place.

#### 8.2.2 Leak Test

Leak test procedures are discussed in Appendix 8.3.

# 8.2.3 <u>Subsystem Maintenance</u> The cask does not have any subsystems.



3.2.4 <u>Valves and Gaskets on Containment Vessel</u> Annual replacement will be made of all seals and gaskets.

3.2.5 <u>Shielding</u> No tests are required for shielding performance other than normal transportation compliance surveys.

3.2.6 <u>Thermal</u> No thermal tests are required.







.

.

.

.

APPENDIX 8.3



# APPENDIX 3A LIAK TEST PROCEDURE

#### 1.0 <u>SCOPE</u>

1.1 <u>Purpose</u>

This procedure will be used in performance of Halogen Gas Leak Testing of the 1-13C II Transport Cask.

1.2 Applicability

This procedure establishes the method for fabrication verification, third use, annual and/or seal replacement leak testing.

# NOTE: THIS PROCEDURE IS NOT TO BE USED FOR ASSEMBLY VERIFICATION OF A LOADED CASK.

#### 2.0 <u>REFERENCES</u>

- 2.1 ANSI N14.5 1977, American National Standard for leak test on packages for shipment of radioactivec materials.
- <sup>2</sup>.2 General Electric Operating Instructions "The Ferret Leak Detector" Type H-25, No. 198-4540K15-001F, Dated 1/81.
- 2.3 CNSI Drawing 1-436-111, Rev. D, sheets 1 and 2.
- 2.4 1-13C II Cask Handling Procedure, TR-OP-026.



ŧ

#### 3.0 <u>RESPONSIBILITIES</u>

3.1 Test Technicians - Performs leak testing in accordance with this procedure.

#### 4.0 DESCRIPTION

Leak testing is performed to verify the integrity and proper sealing of the cask. A R-12 Freon source is connected to the cask drain port, the cask is pressurized and after 10 minutes the potential areas of leakage are checked. If leakage in excess of the specified acceptance criteria is detected, the cask must be disassembled to repair the leaks.

#### 5.0 <u>REQUIREMENTS</u>

- 5.1 Prerequisites
  - 5.1.1 Interior and exterior cask surfaces shall be clean and free of contaminants which could affect leak test sensitivity.
- 5.2 Equipment
  - 5.2.1 Leak Detector General Electric Type H-25 "The Ferret Leak Detector" or equivalent equipment.
  - 5.2.2 Pressure Gauge A gauge capable of indicating required pressure with a calibrated accuracy of  $\pm .25^{\circ}$  of gauge reading.
  - 5.2.3 Temperature recording equipment capable of reading within  $\pm 2^{\circ} F$ .



5.2.4 Halogen Gas Source (dichlorodifluoromethane, Freon R-12) shall be the halogen gas.

5.2.5 Test Rig Adaptor

5.0 <u>ENVIRONMENTAL CONDITIONS</u>

Ambient conditions

#### 7.0 SPECIAL PRECAUTIONS

Proper radiological precautions should be used to prevent the spread of contamination which may be present.

- 8.0 PROCEDURE
  - 8.1 Leak Detector Preparation
    - Leak detector is to be calibrated and operated in accordance with Reference 2.2.
    - (2) Temperature measuring equipment shall be attached to cask with the temperature reading recorded.
    - (3) Calibration of the leak detector shall be established at .21 ounces/year for Freon R-12. (3.86  $\times 10^{-5}$  ATM - cc/sec)
  - 3.2 Discussion of allowable leak rates derived from containment Section 4.0 of 1-13C II Safety Analysis Report.









The following calculations are shown for the determination of cask charge pressure based on the ambient temperature at time of test.

NOTE: THE GRAPH, FIGURE 1, DEPICTS THE RESULTS OF THESE CALCULATIONS AND WILL BE USED FOR ALL NORMAL CONDITIONS AT TIME OF LEAK TEST.

$$P_1^3 - P_1^2 - P_1 - (4.1534 \times 10^{-7}) = \frac{1_h^T r_i}{L_s} = 0$$

Where:  $T_{\chi} = Cask Temperature, {}^{O}K$   $P_{t} = Required Charge Pressure, ATM$   $L_{s} = Permissable Standard Leak Rate (Dry air at 25 °C and$ ATM)1.34 x 10<sup>-5</sup> ATM - cc/sec

 $I_{h}$  = Detector Sensitivity = 0.21 oz/yr (3.36 x 10<sup>-5</sup> Atm - cc/sec)

8.3 Leak Testing

- (1) Check all surfaces with detector to determine cask is free of residual halides which could affect test.
- (2) Clean any surfaces found to have halide levels above background indications.
- (3) Examine flat gasket and O-ring for damage or deterioration and replace as applicable.
- (4) Install lid in cask in accordance with Reference 2.4.



(5) Record temperature measurement of cask.

- (5) Connect R-12 Freen source to the cask drain port using test rig adaptor
- (7) Pressurize cask to the proper level as determined by value from Figure 1 and record pressure
- (3) Allow 10 minutes dwell time after pressurization of cask.
- (9) Using leak detector check area around vent and test rig adaptor assembly for halides and record results.
- (10) Remove vent seal plug and check orifice for halides and record results.
- (11) If results of leak checks are acceptable, proceed to next step.
  If leakage in excess of acceptance level is detected, disassemble cask and inspect for damage to the seals, cask, and test rig.
  Return to 8.3 (1) and repeat test and record results.
- (12) Close off pressure line to cask drain port leaving cask ' pressurized and remove Freon R-12 source.





NOTE: RELEASE OF PRESSURICED R-12 IN CASK TO THE ATMOSPHERE WILL CONTAMINATE TEST AREA.

- (13) Reconnect Freon R-12 source at vent port.
- (14) Pressurize (area between gasket and 0-ring) to the established pressure from Figure 1. Record pressure.
- (15) Allow 10 minutes dwell time after pressurization of cask.
- (16) Using leak detector check area around contact surface between lid and cask, bolt connection areas, vent port and test rig adaptor assembly. Record Results
- (17) If results of leak check are acceptable, proceed to next step. If leakage rate in excess of acceptance criteria is detected, disassemble cask and inspect for damage to seals and casks surfaces. Repeat test as required. Record Results
- (13) Record Temperature
- (19) Verify the following leak checks have been completed:
  - (a) Vent and drain ports
  - (b) O-ring seal
  - (c) Flat gasket seal.



- (20) Turn off and remove leak testing equipment from test area. (Release of R-12 directly to test equipment may cause damage.)
- (21) Remove Freon R-12 source and open vent and drain ports to release pressure.
- (12) Remove lid and clean cask as necessary to remove residual halogen.
- (23) Reassemble and prepare cask for shipment in accordance with Reference 2.4.

#### 5.0 <u>RECORDS, REPORTS AND NOTIFICATIONS</u>

5.1 Reports

Each leak test performed shall be documented by a report which shall include as a minimum:

- (a) Type of test annual, gasket replacement,
- (b) Equipment and calibration data,
- (c) Temperature of cask surface prior to and after test
- (d) Charge pressure of Freon R-12,
- (e) Results of all leak checks performed, and
- (f) Date and signature of test operator.

5.2 Notifications

Any problems encountered during testing which will adversely affect test results shall be forwarded to US DOE representative for resolution immediately.



.

.

-~-

# 5.3 Records

.

.

.

.

.







	•			•	•	•	-																																						-
				•										•					•					///////////////////////////////////////									))												-
			•	••••••	•																· · · · · · · · · · · · · · · · · · ·																 •								
															•	•				•															· · · · · · · · · · · · · · · · · · ·		 •				•	•			
-			•				•••••••••••••••••••••••••••••••••••••••				•	•••••••••••••••••••••••••••••••••••••••	•			•																							-						
•			-	.   .   .	-	-		ŀ		·		•		•	•		.  .  .	-	ŀ							·		·	-	•	•	·	•				•	•		•					

(A)

·łł 12.40  $\frac{1}{1} = 0.21 \text{ OZ/YR}$   $\frac{1}{1} = 0.31 \text{ OZ/YR}$   $\frac{1}{1} = 0.39 \times 10^{-9} \text{ atm-cc/set}$ 12.20 12.00 (STANDARD CONDITIONS 1 11.80 i F Lttl 11.60 -----411 100' + 90 1111 1.11 1111 11 111111 1111 11 11 ł 1 80 1.0 110 • 70 1111 - 60 F1 E1 4() 11 { ° F CASE TEMPERATURE

14

1

.

- 1

t

(6130)

PRESSURE

CHARGE

REGUIRED

:

# APPENDIX 3B LEAK TEST PROCEDURE

#### 1.0 <u>\$7075</u>

1.1 <u>Purpose</u>

This procedure will be used to perform Halogen Gas Leak Testing of the 1-13C II Transport cask.

#### 1.2 Applicability

This procedure establishes the method for assembly verification, leak testing prior to each shipment of a loaded 1-13C II transport cask.

<u>MOTE:</u> THIS PROCEDURE IS NOT TO BE USED FOR FABRICATION <u>VERIFICATION THIRD USE. ANNUAL AND SEAL REPLACEMENT</u> LEAK TESTS.

#### 2.0 <u>REFERENCES</u>

- 2.1 ANSI N14.5 1977, American National Standard for leak test on packages for shipment of radioactivec materials.
- 2.2 General Electric Operating Instructions "The Ferret Leak Detector" Type H-25, No. 198-4540K15-001F, Dated 1/81.
- 2.3 CNSI Drawing 1-436-111, Rev. D, sheets 1 and 2.
- 2.4 1-13C II Cask Handling Procedure, TR-OP-026.



#### 3.0 <u>REQUIREMENTS</u>

- 3.1 Prerequisites
  - 3.1.1 Proper radiological precautions should be used during all phases of this leak test.
  - 3.1.2 The cask shall be examined visually for weld integrity and condition of gasket and leak surfaces.
  - 3.1.3 Cask surfaces shall be clean and free of contaminants which could affect leak test sensitivity.
- 3.2 Tools, Materials and Equipment

The following equipment is required for Halogen Gas Leak Testing:

- Leak Detector General Electric Type H-25 "The Ferret Leak Detector" or equivalent equipment.
- (2) Pressure Gauge A gauge capable of indicating required pressure with a calibrated accuracy of  $\pm$  .25% of gauge reading.
- (3) Temperature recording equipment capable of reading within  $\pm 2^{\circ}F$ .
- (4) Halogen Gas Source (dichlorodifluoromethane Freon R-12) shall be the halogen gas.
- (5) Test Rig Adapter


## 4.0 PROCEDURE

- Leak Detector Preparation 4.1
  - ·.) Leak detector is to be calibrated and operated in accordance with Reference 2.2.
  - (2)Temperature measuring equipment shall be attached to cask with the temperature reading recorded.
  - Calibration of the leak detector shall be established (3)at .50 oz/yr for Freon R-12.
- 4.2 Discussion of allowable leak rates derived from containment Section 4.0 of 1-13C II Safety Analysis Report.

The following calculations are shown for the determination of cask charge pressure based on the ambient temperature at time of test.

THE GRAPH, FIGURE II, DEPICTS THE RESULTS OF THESE NOTE: CALCULATIONS AND WILL BE USED FOR ALL NORMAL CONDITIONS AT THE TIME OF LEAK TEST.

$$P_1^3 - P_1^2 - P_1^{+1} - (4.1534 \times 10^{-7}) - \frac{I_k^T k}{L_s} = 0$$

Where:  $T_{v} = CLSk$  Temperature, <sup>o</sup>K P\_ = Required Charge Pressure, Atm L = Permissable Standard Leak Rate

 $5.0 \times 10^{-4}$  Atm - cc/sec







Revision 0

I<sub>n</sub> = Detector Sensitivity = 0.50 oz/yr

- 4.3 Leak Testing
  - (1) Verify cask has been prepared in accordance with Reference 2.4 and is ready for shipment.
    - (2) Check all surfaces with detector to determine cask is free of residual halides which could affect test.
    - (3) Clean any surfaces found to have halide levels above background indications.
    - (4) Record temperature measurement of cask.
    - (5) Remove vent plug and connect R-12 Freon source to the cask vent port.
    - (6) Pressurize cask to the proper level as determined by value from Figure II. Record Pressure
    - (7) Allow 10 minutes dwell time after pressurization of cask:
    - (8) Using leak detector, check area around contact surface between lid and cask, bolt connection areas and test rig adapter assembly. Record Results





Revision 0

- (?) If results of leak check are acceptable, proceed to next step. If leakage rate in excess of acceptance criteria is detected, disassemble cask and inspect for damage to seals and cask surfaces. Record Results
- (10) Record temperature
- (11) Turn off and remove leak testing equipment from test area. (Release of R-12 directly to test equipment may cause damage.)
- (12) Remove Freon R-12 source, release pressure, and
- (13) Use appropriate means to purge Freon gas from cask.
- (14) Complete cask assembly and prepare for shipment in accordance with Reference 2.4.

## 5.0 <u>RECORDS, REPORTS AND NOTIFICATIONS</u>

5.1 Reports

Each leak test performed shall be documented by a report which shall include as a minimum:

- (a) Type of test assembly verification
- (b) Equipment and calibration data,
- (c) Temperature of cask surface prior to and after test,
- (d) Charge prensure of Freon R-12.
- (e) Results of all leak check performed, and
- (f) Date and signature of test operator.





## 5.2 Notifications

Any problems encountered during testing which will adversely affect test results shall be forwarded to DOE representative for resolution immediately.

۰.

## 5.3 Records

A leak check data sheet shall be forwarded to Quality Assurance department.



8	
÷19	
-	

		05011030	ע ראסניב ססב	ssuar (neta)			
	3.						
					55		
4							
0							
				ŀ			
						T	
							_
5							
0							
-							
				•			
				-			
	-						
6			•				
0							
					•		
	-	•		- -			
'7' 							
		•					
,     					•		
SX		-					
					+		
					-		
י'   פנ							
91 91 F						- {	
) )							
     _ .							
		•					
			-				
iv   							
		(					
		1					
					.		
					+, 		
		S E O					
						<b>j</b>	
	4.			     			
6	1 5	D S		1/ 5. 1.) {S		]_	
	þ			/2       0   1   1			
	0 ]   			ן 1 א 1 אא 1			
				)2 .1 .0 .1 .0 .1 .0 .1 .0 .1 .0 .1 .0 .1 .0 .1 .0 .1 .0 .1 .0 .1 .0 .1 .0 .1 .0 .1 .0 .1 .0 .1 .0 .1 .0 .1		-	
		R		a   			
				       		T	
	ГА СН 			.  -0     			
	K D	A   					
		6.P					
	77  T 	(X.		с.     			
	D			       			

1)